EK-DCJ11-UG-PRE

DCJ11 Microprocessor User's Guide



PRELIMINARY



DCJ11 Microprocessor User's Guide

PRELIMINARY

Prepared by Educational Services of Digital Equipment Corporation

Preliminary, October 1983

Copyright © 1983 by Digital Equipment Corporation All Rights Reserved

The material in this manual is for informational purposes and is subject to change without notice.

Digital Equipment Corporation assumes no responsibility for any errors which may appear in this manual.

Printed in U.S.A.

The following are trademarks of Digital Equipment Corporation:

DEC	EduSystem	RSTS
DECnet	IAS	RSX
DECUS	MASSBUS	TOPS-10
DECsystem-10	MI! 'C-11	TOPS-20
DECSYSTEM-20	OMI BUS	UNIBUS
DECwriter	OS/8	VAX
DIBOL	PDP	VMS
dıgıtal	PDT	VT

CONTENTS

PREFACE

CHAPTER 1 ARCHITECTURE

1.1	INTRODUCTION	1-1
1.2	GENERAL-PURPOSE REGISTERS	1-2
1.3	PROCESSOR STATUS WORD	1-3
1.3.]	l Processor Modes	1-5
1.3.2	2 Priority Levels	1-5
1.3.3	3 The Trace/Trap Bit	1-6
1.3.4	4 Condition Codes	1-6
1.3.9	5 Processor Status (PS) Protection	1-6
1.4	INTERRUPTS AND TRAPS	1-11
1.5	HALTING DCJ11 OPERATION	1-14
1.6	PROGRAM INTERRUPT REQUEST REGISTER	1-15
1.7	CPU ERROR REGISTER	1-15
1.8	STACK PROTECTION	1-16
1.9	FLOATING-POINT PROCESSING	1-17
1.10	MEMORY SYSTEM REGISTERS	1-17
1.11	DIRECT-MEMORY ACCESS (DMA) MECHANISM	1-17

CHAPTER 2 PIN DESCRIPTION

2.1 INTR	ODUCTION
2.2 DATA	/ADDRESS LINES (DAL<21:00>) 2-2
2.2.1	Upper Data/Address Lines (DAL<21:16>) 2-2
2.2.2	Lower Data/Address Lines (DAL<15:00>) 2-2
2.3 SYST	EM CONTROL LINES
2.3.1	Bank Select (BS<1:0>) 2-2
2.3.2	Address Input/Output (AIO<3:0>)
2.3.3	Buffer Control (BUFCTL) 2-4
2.3.4	Continue (CONT) 2-4
2.3.5	Data Valid (DV) 2-4
2.4 TIMI	NG SIGNALS
2.4.1	Address Latch Enable (ALE) 2-5
2.4.2	Stretch Control (SCTL)
2.4.3	Strobe (STRB) 2-5
2.4.4	Clock 1 (CLK) 2-5
2.4.5	Clock 2 (CLK2) 2-5
2.5 STAF	T/STOP CONTROL 2-5
2.5.1	Initialize (INIT) 2-5
2.5.2	Halt (HALT) 2-6
2.6 STAT	CUS SIGNALS
2.6.1	Cache Miss (MISS) 2-6
2.6.2	Parity Error (PARITY) 2-6
2.6.3	Abort (ABORT) 2-6
2.6.4	Map Enable (MAP) 2-7
2.6.5	Predecode (PRDC) 2-7
2.7 INTE	RRUPT AND DMA CONTROL 2-7
2.7.1	<pre>Interrupt Request (IRQ<3:0>) 2-7</pre>
2.7.2	Direct-Memory Access Request (DMR) 2-8
2.7.3	Power Fail (PWRF) 2-8

2.7.4	Floating-Point	Exception	(FPE)		2
2.7.5	Event (EVENT).				2
2.8 TES	ST PINS				2
2.8.1	Test 1 (TEST1)				2
2.8.2	Test 2 (TEST2)				2
2.9 OSC	CILLATOR PINS				2
2.9.1	XTALI and XTAL	O Generatic	n		2
2.10 POW	VER PINS				2
2.10.1	Power (Vcc)				2
2.10.2	Ground (GND)				2
2.11 PIN	I DESCRIPTION SU	MMARY			2
				,	

CHAPTER 3 BUS CYCLES

3.1	INTRODUCTION	3-1
3.2	DURATION OF BUS CYCLES	3-2
3.3	BUS CYCLE PARTS	3-3
3.4	NON-I/O (NIO) CYCLE	3-3
3.5	BUS READ CYCLE	3-4
3.6	BUS WRITE CYCLE	3-6
3.7	GENERAL-PURPOSE (GP) READ CYCLE	3-8
3.8	GENERAL-PURPOSE (GP) WRITE CYCLE	3-9
3.9	INTERRUPT ACKNOWLEDGE CYCLE	3-10
3.10	DMA REQUESTS AND GRANTS	3-11

CHAPTER 4 MEMORY MANAGEMENT

	٦
4.1 INTRODUCTION	-1
4.2 ADDRESSING	-1
4.3 I SPACE AND D SPACE 4	-2
4.4 CONSTRUCTION OF A PHYSICAL ADDRESS 4	- 3
4.5 MANAGEMENT REGISTERS 4	- 5
4.5.1 Page Address Registers (PARs) 4	-6
4.5.2 Page Descriptor Registers (PDRs) 4	-6
4.5.2.1 Bypass Cache 4	-7
4.5.2.2 Page Length Field (PLF)	-7
4.5.2.3 Page Written	-7
4.5.2.4 Expansion Direction (ED)	-7
4.5.2.5 Access Control Field	-8
4 5.2.6 Reserved Bits	-8
4.6 INTERRIPT CONDITIONS UNDER MEMORY MANAGEMENT CONTROL	-8
4.9 INTERNOT CONDITIONS UNDER GEMORT MARAGEMENT CONTROLS. 4	<u></u>
4.7 RODI RECOVERT REGISTERS	_0
A 7 1 1 Error Flage	
4.7.1.1	_0
$4.7.1.1.1 \qquad \text{Abort} \qquad \text{Dece Length}$	
$4.7.1.1.2 \qquad \text{Abort} = \text{Page Gength} \dots \dots$	-9
4.7.1.1.3 Abort Read Only	-10
4.7.1.2 Reserved Bits 4	-10
4.7.1.3 Processor Mode 4	-10
4.7.1.4 Page Address Space 4	-10
4.7.1.5 Page Number 4	-10
4.7.1.6 Enable Relocation 4	-10
4.7.2 Memory Management Register #1 (MMR1) 4	-10
4.7.3 Memory Management Register #2 (MMR2)	-13
4.7.4 Memory Management Register #3 (MMR3) 4	-1]
4.7.4.1 Reserved Bits 4	-11
4.7.4.2 Enable I/O Map 4	-11

4.7.4.5 Kernel, Supervisor, and User Mode D Space Bits.	
	4-1
4.7.6 Clearing Status Registers Following Abort	4-1
4.7.7 Multiple Faults	4-1
4.8 MMU IMPLEMENTATION	4-1
4.8.1 Typical Memory Page	4-1
4.8.2 Non-Consecutive Memory Pages	4-1
4.8.4 Transparency	4-1
4.9 MEMORY MANAGEMENT UNIT REGISTER MAP	4-1
CHAPTER 5 SPECIAL FEATURES	
5.1 INTRODUCTION	5-1 5-1
5.2.1 Cache Control Register	5-1
5.2.1.1 Unconditional Cache Bypass (R/W)	5-2
5.2.1.2 Force Cache Miss (R/W)	5-3
5.2.1.3 Uninterpreted Bits	5-3
5.2.3 General Operation	5- 5-
5.2.4 Cache Memory In A Multiprocessor Environment	5-
5.2.5 Sample Implementation	5-
5.3 CONSOLE ODT	5-1
5.3.1 Terminal Interface	5-9
5.3.1.1 Receiver Control/Status Register (RCSR) 5.3.1.2 Receiver Buffer Register (RBUE)	5-1
5.3.1.3 Transmitter Control and Status Register (XSCR).	5-
5.3.1.4 Transmitter Buffer Register (XBUF)	5-
5.3.2 Console ODT Operation	5-
5.3.2.1 Console ODT Initialization	5-
5.3.3 Console ODT Command Set	5- 5-
5.3.3.1 (ASCII 057) Slash	5-
5.3.3.2 <cr> (ASCII 015) Carriage Return</cr>	5-
5.3.3.3 <lf> (ASCII 012) Line Feed</lf>	5-
5.3.3.4 $\$$ (ASCII 044) Or R (ASCII 122)	5-
5.3.3.6 G (ASCII 123) Processor Status word	5-
5.3.3.7 P (ASCII 120) Proceed	5-
5.3.3.8 Control-Shift-S (ASCII 023)	5-
5.3.4 Address Specification	5-
5.3.4.1 General Registers	5-
5.3.4.3 Floating-Point Accumulators)- 5-
5.3.5 Entering Octal Digits	- ś-
5.3.6 ODT Timeout	5-
5.3.7 Invalid Characters	5-
5.4 DCJ11 PIPELINE PROCESSING	5-
	_ _

6.2.1	Single-Operand Addressing	6-3 6-3
6.2.3	Direct Addressing	6-5
6.2.3.1	Register Mode	6-6
6.2.3.2	Autoincrement Mode	6-7
6.2.3.3	Autodecrement Mode	6-8
6.2.3.4	Index Mode	6-9
6.2.4	Deferred (Indirect) Addressing	6-11
6.2.5	Use of the PC As a General-Purpose Register	6-14
6.2.5.1	Immediate Mode	6-15
6.2.5.2	Absolute Addressing	6-16
6.2.5.3	Relative Addressing	6-17
6.2.5.4	Relative-Deferred Addressing	6-17
6.2.6	Use of the Stack Pointer As a General-Purpose	
		6-18
6.3 INST	TRUCTION SET	6-18
0.3.1	Instruction Formats	6-19
6.3.2	Byte Instructions	6-22
0.3.3	List of Instructions	6-25
0.3.4		6-20
0.3.4.1	General	6-20
6 3 4 3	Multiple-Drecision	6-30
6 3 / /	BE Word Operators	6-25
6 3 5	Double-Operand Instructions	6-37
6.3.5.1	General.	6 - 37
6.3.5.2		6-42
6.3.6	Program Control Instructions	6-45
6.3.6.1	Branches	6-45
6.3.6.2	Signed Conditional Branches	6-49
6.3.6.3	Unsigned Conditional Branches	6-51
6.3.6.4	Jump and Subroutine Instructions	6-52
6.3.6.5	Traps	6-56
6.3.6.6	Miscellaneous Program Control	6-60
6.3.6.7	Reserved Instruction Traps	6-62
6.3.6.8	Trace Trap	6-62
6.3.6.8.3	Special Cases Of The T-Bit	6-63
6.3.7	Miscellaneous Instructions	6-64
6.3.8	Condition Code Operators	6-66
CHAPTER (7 FLOATING-POINT ARITHMETIC	
7 1 7107	PODUCETON	۲ <i>۲</i>
7.1 INTE	AUDUCIION	7-1 7-1
	Non-Vanishing Floating-Doint Numberg	7-1
7 2 2	Floating-Point Zero	7-2
7.2.3	Undefined Variables	7-2
7.2.4	Floating-Point Data	7-2
7.3 FLO	ATING-POINT STATUS REGISTER	7-3
7.4 FLO	ATING EXCEPTION CODE AND ADDRESS REGISTERS.	7-7
7.5 FLO	ATING-POINT INSTRUCTION ADDRESSING	7-8
7.6 ACCI	JRACY	7-9
7.7 FLO	ATING-POINT INSTRUCTIONS	7-11
CHAPTER &	8 INTERFACING	

8.1	INTRODUCTION	8-3	1
-----	--------------	-----	---

8.2 GEN	NERAL-PURPOSE (GP) CODES
8.3 POI	WER-UP AND INITIALIZATION
8.3.1	Initialization Timing
8.3.2	Initialization Microroutine
8.3.3	Power-Up Configuration
8.3.4	Power-Up Circuit
8.4 OT	HER MICROROUTINES

APPENDIX A DC CHARACTERISTICS

APPENDIX B AC CHARACTERISTICS

APPENDIX C HARDWARE AND SOFTWARE DIFFERENCES

•

- APPENDIX D INSTRUCTION TIMING
- APPENDIX E GLOSSARY

INDEX

FIGURES

Figure

Page

1-1	DCJ11 Block Diagram	1-1
1-2	DCJ11 General-Purpose Registers	1-2
1-3	Processor Status Word	1-3
1-4	PIRO Register	1-15
1-5	CPU Error Register	1-15
2-1	DCJ11 Pin Assignments	2-1
2-2	Typical XTALI and XTALO Generation	2-9
3-1	Non-Stretched Non-I/O Cycle	3-3
3-2	Stretched Non-I/O Cycle	3-4
3-3	Non-Stretched Bus Read Cycle	3-5
3-4	Stretched Bus Read Cycle	3-5
3-5	Bus Write Cycle	3-7
3-6	General-Purpose (GP) Read Cycle	3-8
3-7	General-Purpose (GP) Write Cycle	3-9
3-8	Interrupt Acknowledge Cycle	3-10
4-1	Virtual Address Mapping Into Physical Address	4-2
4-2	Interpretation Of A Virtual Address	4-3
4-3	Displacement Field Of Virtual Address	4-4
4-4	Construction Of A Physical Address	4-4
4-5	Active Page Registers	4-6
4-6	Page Address Register	4-6
4-7	Page Descriptor Register (PDR)	4-7
4-8	Memory Management Register #0 (MMR0)	4-9
4-9	Memory Management Register #1 (MMR1)	4-1
4-10	Memory Management Register #3 (MMR3)	4-1
4-11	16-Bit Mapping	4-13
4-12	18-Bit Mapping	4-12
4-13	22-Bit Mapping	4-1
4-14	Typical Memory Page	4-1
4-15	Non-Consecutive Memory Pages	4-1
4-16	Typical Stack Memory Page	4-1

		c o
5-1	Cache Control Register	5-2
5-2	Hit/Miss Register	5-2
5-3	Physical Address Partitioning For Cache Memory	5-4
5-4	Cache Entry	5-5
5-5	Cache Entry With Parity	5-5
5-6	Sample Cache Control Register	5-6
5-7	Pageiver Control (Status Pagister (PCSP)	5-9
5-7	Receiver Control/Status Register (RCSR/	5-10
5-0	Receiver Buller Register (RBDr)	5-10
5-9	Transmitter Control/Status Register (XCSR)	5-10
5-10	Transmitter Buffer Register (XBUF)	5-11
5-11	Pipeline Filling Process	5-20
6-1	Single-Operand Addressing	6-3
6-2	Double-Operand Addressing	6-4
6-3	Mode 0 Register	6-5
6-4	Mode 2 Autoincrement	6-5
6-5	Mode 4 Autodecrement	6-5
6-6	Mode 6 Index	6-5
6-7	INC R3 Increment	6-6
6-9		6_7
6-0	COMP D4 Complement Bute	5-7
6-9	COMB R4 Complement Byte	n-/
6-10	CLR (R5)+ $CLear$	5-/
6-11	CLRB (R5)+ Clear Byte	6-8
6-12	ADD (R2) + R4 Add	6-8
6-13	INC - (RO) Increment	6-9
6-14	INCB - (R0) Increment Byte	6-9
6-15	ADD $-(R3)$, R0 Add	6-9
6-16	CLR 200(R4) Clear	6-10
6 - 17	COMB 200(R1) Complement Byte	6-10
6-18	ADD 30(R2) 20(R5) Add	6-11
6 10	Mode 1 Designer-Deferred	6-11
6 20	Mode I Register-Dereried	6 1 C
6-20	Mode 3 Autoincrement-Deferred	0-12
0-21	Mode 5 Autodecrement-Dererred	5-12
6-22	Mode 7 Index-Deferred	5-12
6-23	CLR @R5 Clear	6-23
6-24	INC @(R2)+ Increment	6-24
6-25	COM @-(R0) Complement	6-13
6-26	ADD @1000(R2), R1 Add	6-14
6-27	ADD #10,R0 Add	6-15
6-28	CLR @#1100 Clear	6-16
6-29	ADD @#2000 Add	6-16
6 - 30	INC A Increment.	6-17
6-31	CLP AN Clear	6-19
6-32	Single-Operand Group	6-10
6-33	Double-Operand Group 1	6-20
6 24	Double-Operand Group 2	5-20
0-34	Double-Operand Group Z	5-20
6 30	Program Control Group Branch	5-20
0-30	Program Control Group JSR	5-20
6-37	Program Control Group RTS	6-20
6-38	Program Control Group Traps	6-20
6-39	Program Control Group Subtract	6-21
6-40	Mark	6-21
6-41	Call To Supervisor Mode	6-21
6-42	Set Priority Level	6-21
6-43	Operate Group	6-21
6-44	Condition Group	6-21
6-45	Move To And From Previous Instruction/Data Space Crown	6_00
6-46	Byte Instructions	6-22
~ - 1 /		

7-1 7-2	Single-Precision Format	7-2
7-3	2's Complement Format	7-3
7-4	Floating-Point Status Register	7-4
7-5	Floating-Point Addressing Modes	7-11
8-1		8-2
8-2	Power-Up Configuration Register	8-3
8-4	Power-Up Circuit	8-8
8-5	Power-Down Sequence	8-9
8-6	Console ODT Start Sequence	8-10
A-1	Voltage Waveforms	A-4
B-1	Clock Timing	B-3
B-2	Three State Disable Test Circuit	B-4
B-3	TTL Output Test Circuit	B-4
B-4	MOS Output Test Circuit	B-4
B-5	Non-Stretched Bus Read Timing	B-4
B-6	Stretched Bus Read Timing	B-5
B-/	Bus Write Timing	B-5
B-0	General-Purpose Read Timing	B-0
B-9 B-10	Interrupt Agencyledge Timing	B-0 D-7
B = 10 R = 11	Interrupt Timing	5-/ B-7
~ ~ 1		57

TABLES

Table

l-1	Instructions Influenced By Processor Modes
1-2	Priority Levels
1-3	PS Protection For Explicit Accesses
1 - 4	PS Protection For Traps And Interrupts
1-5	PS Protection For RTL, RTT Instructions
1-6	PS Protection For MTPS Instruction
1-7	PS Initialization During Power-Up
1-8	Interrupts, Traps, and Aborts
2-1	BS Device Selection
2 - 2	ATO Decode
2 - 3	Interrupt Requests on IRO<3:0>
2 - 4	IRO<3:0> Interrupt Request Levels
3-1	ATO Codes for Bus Cycles
3 - 2	General-Purpose Read Codes
3-3	General-Purpose Write Codes
3-4	Interrupt Acknowledgement
4 – 1	I and D Space Referencing
4 - 2	Mode Bit Operations
5-1	Typical Hit/Miss Operations
5-2	Console ODT Commands
5-3	Pipeline Flow
7-1	FDS Register Bite
, <u>,</u>	CD Codes and Eurotions
C = 1	DCILL Programming Difference
- L	

Page

This user's guide is intended to familiarize the reader with the hardware and software characteristics of the DCJ11 microprocessor CPU chip. It is assumed that the reader has had some experience with microprocessor design. Readers should also have some familiarity with PDP-11 architecture.

The book is organized as follows:

Chapter 1 provides an architectural overview of the DCJ11.

Chapter 2 describes the function of each DCJ11 pin.

Chapter 3 describes the various types of DCJ11 bus cycles and provides an overview of the timing relationships among DCJ11 inputs and outputs during these cycles.

Chapter 4 describes the architecture and operation of the DCJll's integral memory management unit.

Chapter 5 provides information on three special features integral to the DCJ11: cache memory registers (this description also includes cache memory design considerations), console ODT (also called micro-ODT), and pipeline processing.

Chapter 6 describes the DCJ11 base instruction set.

Chapter 7 describes the integral floating-point unit and its instruction set.

Chapter 8 provides some introductory information on interfacing external logic to the DCJ11. Power-up and initialization circuits are provided.

Appendix A contains a summary of the DCJ11 DC characteristics.

Appendix B contains a summary of the DCJ11 AC characteristics.

Appendix C summarizes the hardware differences between: (1) the DCJ11 and the PDP-11/44 and (2) the DCJ11 and the PDP-11/70. Appendix C also contains a summary of the software differences between the DCJ11 and other processors in the PDP-11 family.

Appendix D describes how to determine the duration of a DCJ11 instruction. Timings for both the base instruction set and the floating-point instruction set are provided.

Appendix E contains a brief glossary of some DCJ11 terms.

1.1 INTRODUCTION

This chapter provides a brief introduction to the architecture of the DCJ11 microprocessor. The DCJ11 is organized as shown in Figure 1-1.



Figure 1-1 DCJ11 Block Diagram

As shown in Figure 1-1, the DCJ11 microprocessor consists of a data chip`and a control chip.

The data chip performs all arithmetic and logic functions, handles all data and address transfers, and generates most of the signals used for system timing. In addition to the primary execution data path, the data chip contains memory management logic, an I/O state sequencer, and floating-point and cache control registers. The control chip directs the operation of the data chip with microinstructions. The major components of the control chip are the microprogram control store and the microprogram sequencing logic.

A detailed description of the data chip and control chip and the interface between them is beyond the scope of this book. We will consider the data chip and control chip as one functional unit and will describe only those portions of this unit that are architecturally significant to the design engineer.

The remainder of this chapter briefly describes each of the major components of the DCJ11 architecture. The chapter covers six major topics:

- o General-purpose registers
- o Processor status word
- o Traps and interrupts
- o Floating point processing
- o Memory system registers
- o DMA mechanism

1.2 GENERAL-PURPOSE REGISTERS

As shown in Figure 1-2, the DCJ11 has a dual set of six registers R0 through R5 and R0' through R5', three stack pointers (R6) corresponding to the three processor modes (see Paragraph 1.3.1), and a program counter (R7). R0 through R5 is also referred to as register set 0 and R0' through R5' is also called register set 1.

These registers are called general-purpose because they can be used in a variety of ways. General-purpose registers serve as accumulators, index registers, autoincrement registers, autodecrement registers, or as stack pointers for temporary storage of data. Arithmetic operations can be performed between one general-purpose register and another or between a general-purpose register and memory or an I/O device register.

RO	R0'	KSP
R1	R1'	SSP
R2	R2'	USP
R3	R3'	· · · ·
R4	R4'	PC
R5	R5'	PSW

Figure 1-2 DCJ11 General-Purpose Registers

At any given time, either register set R0 through R5 is used or register set R0[°] through R5[°] is used. The two sets can not be used simultaneously. These general-purpose registers are organized as two sets to increase the speed of context switching and some types of real-time data handling.

1-2

Register R6 is used as the hardware stack pointer (SP), which indicates the last entry in the appropriate stack (the stacks are common temporary areas with LIFO - last in, first out characteristics). There are three stack pointers: a kernel stack pointer (KSP), a supervisor stack pointer (SSP), and a user stack pointer (USP). Each stack pointer is associated with a different processor mode (see Paragraph 1.3.1). When an interrupt or trap occurs, the current CPU state (PC and PS) is automatically pushed on the stack indicated by the interrupt or trap vector (see Paragraph 1.4 for more information on interrupts and traps). The stack-based architecture also facilitates reentrant programming.

Register R7 is used as the program counter (PC). The PC contains the address of the next instruction to be executed; thereby controlling the order of execution of instructions. The PC is a general-purpose register in the sense that it is directly accessible by all single- and double-operand instructions. Much of the power of the DCJ11 instruction set is achieved by utilizing the PC in conjunction with various addressing modes. The PC is not normally used as an accumulator for arithmetic operations.

1.3 PROCESSOR STATUS WORD

As shown in Figure 1-3, the processor status word (PS) contains the condition codes describing the arithmetic or logical results of the last instruction, a trace bit that forces a trap at the end of an instruction (used for program debugging), the current processor priority, and the current and previous processor modes. The PS is located at physical address 1777776.



Figure 1-3 Processor Status Word

BIT	NAME	FUNCTION
15:14	Current Mode	Current processor mode:

- 15:14 Current Mode (RW, protected)
- Bits Mode 15 14 0 0 Kernel 0 1 Supervisor 1 0 Illegal
 - l l · User

- 13:12 Previous Mode (RW, protected)
- 11 Register Set (RW, protected)
- 10:9 Unused (Read only)
- 8 Reserved (RW)
- 7:5 Priority (RW, protected)

Previous processor mode; same encoding as for bits <15:14>.

General register set select: 0 = register set 0 (R0--R5).1 = register set 1 (R0'--R5').

The bits are unused and are always read as zeroes.

This bit is reserved for future DIGITAL use.

Processor interrupt priority level:

Priority Level

Also called the T-bit. When set, the processor traps to location 14

instruction. This bit cannot be

7	6	5	
1	1	1	7
1	1	0	6
1	0	1	5
1	0	0	4
0	1	1	3
0	1	0	2
0	0	1	1
0	0	0	0

Bits

Trace Trap (RW, protected)

set directly by writing data to the PS. This bit is typically set by the RTI/RTT instruction. Trace trap is disabled when this bit is zero. Condition Codes

Processor condition codes:

at the end of the current

- N: Set if the result of the previous operation was negative.
- Z: Set if the result of the previous operation was zero.
- V: Set if the previous operation resulted in an ar_thmetic overflow.
- C: Set if the previous operation resulted in a carry of its most significant bit.

(RW)

4

1.3.1 Processor Modes - Three processor modes (user, supervisor, and kernel) permit a fully protected environment for a multiprogramming system by providing the programmer with three distinct sets of processor stacks and memory management registers for memory mapping. In addition, certain PDP-11 instructions are privileged in that their operation is inhibited in supervisor and user modes. For example, in supervisor or user mode, the processor will ignore the RESET and SPL (Set Priority Level) instructions and the HALT instruction will cause a trap through the vector at virtual address 4 in kernel data space. In kernel mode, the processor will execute all instructions. A summary of the effects of processor modes on various instruction types is provided in Table 1-1.

Table 1-1 Instructions Influenced by Processor Modes

Instruction or Instruction Type	Operation in Kernel Mode	Operation in Supervisor/User Mode
HALT	Depends on halt option selected (see Paragraph 1.5)	Traps through a vector at location 4 in kernel data space.
WAIT, RESET, SPL	Executes as specified	Executes as a NOP.
RTI, RTT, MPTS	Can alter PS<7:5>	Can not alter PS<7:5>
Stack Reference	Checked for stack overflow.	Not checked for stack overflow.

1.3.2 Priority Levels - 'The priority level (mask bits) is contained in bits <7:5> of the PS and is used by software to determine which interrupts will be processed, as indicated in Table 1-2.

Table 1-2 Priority Levels

Octal of PS<	Value 7:5>	Int Act	ter i knov	upt vled	: Le lged	evel 1	L
	7	Nor	ne				-
	6	7					
	5	7,	6				
	4	7,	6,	5			
	3	7,	6,	5,	4		
	2	7,	6,	5,	4,	3	
	1	7,	6,	5,	4,	3,	2
	0	7,	6,	5,	4,	З,	2,

1.3.3 The Trace/Trap Bit - The trace/trap bit (bit 4) is used for program debugging, enabling single-step execution of instructions for step-by-step monitoring.

1

1.3.4 Condition Codes - The four condition codes N, Z, V, and C contain information about the result of the last CPU operation. These bits are set as described in Paragraph 1.3.

1.3.5 Processor Status (PS) Protection - Tables 1-3, 1-4, 1-5, 1-6, and 1-7 summarize how the PS is protected under a variety of conditions. The PS is initialized at power-up (the value to which it is initialized depends on power-up options) and is cleared at console start. The RESET instruction does not affect the PS.

Table 1-3 PS Protection For Explicit Accesses

PS Bit(s)	User	Super	Kernel
Condition	loaded	loaded	loaded
Codes	from	from	from
PS <3:0>	source	source	source
Trap Bit	un-	un-	un-
PS <4>	changed	changed	changed
Processor	loaded	loaded	loaded
Priority	from	from	from
PS <7:5>	source	source	source
Register	loaded	loaded	loaded
Select	from	from	from
PS <11>	source	source	source
Previous	loaded	loaded	loaded
Mode	from	from	from
PS <13:12>	source	source	source
Current	loaded	loaded	loaded
Mode	from	from	from
PS <15:14>	source	source	source

1-7

Table 1-4 PS Protection For Traps and Interrupts

PS Bit(s)	TRAPS	& INTERN	RUPTS
	User	Super	Kernel
Condition	loaded	loaded	loaded
Codes	from	from	from
PS <3:0>	vector	vector	vector
Trap Bit PS <4>	loaded from vector	loaded from vector	loaded from vector
Processor	loaded	loaded	loaded
Priority	from	from	from
PS <7:5>	vector	vector	vector
Register	loaded	loaded	loaded
Select	from	from	from
PS <11>	vector	vector	vector
Previous Mode PS <13:12>	copied from PS <15:14>	copied from PS <15:14>	copied from PS <15:14>
Current	loaded	loaded	loaded
Mode	from	from	from
PS <15:14>	vector	vector	vector

Table 1-5 PS Protection For RTI, RTT Instructions

1				L
PS Bit(s)	User	RTI, RTT Super	Kernel	
Condition Codes PS <3:0>	loaded from stack	loaded from stack	loaded from stack	
Trap Bit PS <4>	loaded from stack	loaded from stack	loaded from stack	
Processor Priority PS <7:5>	un- changed	un- changed	loaded from stack	
Register Select PS <11>	ORed from stack*	ORed from stack*	loaded from stack	
Previous Mode PS <13:12>	ORed from stack*	ORed from stack*	loaded from stack	
Current Mode PS <15:14>	ORed from stack*	ORed from stack*	loaded from stack	
	1		+	i i

* "ORed from stack" means that when the old PS is popped from the stack (restored), it cannot clear PS<15:11> in the current PS if these bits have been set.

1-9

Table 1-6 PS Protection for MTPS Instruction

4	1	1	1
Í		MTPS	
PS Bit(s)	User	Super	Kernel
Condition Codes PS <3:0>	loaded from source	loaded from source	loaded from source
Trap Bit PS <4>	un- changed	un- changed	un- changed
Processor Priority PS <7:5>	un- changed	un- changed	loaded from source
Register Select PS <11>	un- changed	un- changed	un- changed
Previous Mode PS <13:12>	un- changed	un- changed	un- changed
Current Mode PS <15:14>	un- changed	un- changed	un- changed

Table 1-7 PS Initialization During Power-Up

PS Bit(s)	POWER-UP
Condition Codes PS <3:0>	cleared
Trap Bit PS <4>	cleared
Processor Priority PS <7:5>	depends on power- up option
Register Select PS <11>	cleared
Previous Mode PS <13:12>	cleared
Current Mode PS <15:14>	cleared i.e., kernel mode

1.4 INTERRUPTS AND TRAPS

This paragraph provides a brief overview of DCJ11 interrupts and traps and describes user-visible registers related to interrupts and traps. Abort conditions are also covered. For detailed timing and bus information, see Chapter 3 - Bus Cycles.

Interrupts and traps are requests that cause the DCJ11 to temporarily suspend the execution of the current program and provide service for the device or condition that caused the interrupt or trap. Interrupts differ from traps in that interrupts are initiated by some external event, while traps are caused by conditions internal to the DCJ11.

The DCJ11 operates at any of 8 levels of priority. In general, an interrupt or trap affects the DCJ11 if its priority is greater than the DCJ11's priority as indicated by PS<7:5>. The exception to this is a non-maskable interrupt or trap, which occurs independently of the processor priority. Note that non-maskable

1-11

interrupts and traps have a priority structure amongst themselves.

When an interrupt or trap occurs, the current PS and PC are preserved in order to allow a return to the interrupted program. The new contents of the PC and the PS are fetched from two consecutive memory words called a vector. The first word of the vector contains the interrupt or trap service routine starting address (the new PC), and the second word contains the new PS. Vectors are either predefined by the DCJ11 or are user defined. User defined vectors are vectors associated with interrupts occuring on IRQ<3:0>. The predefined vectors are shown in Table 1-8.

Specifically, for an interrupt or trap, the following sequence of events occurs:

PS>	templ	;save PS, PC in temporary
PC>	temp2	;scratchpad locations
0>	PS<15:14>	;force kernel mode
M[V]>	PC	;fetch PC from vector, data space
M[V+2]>	PS	;fetch PS from vector, data space
templ<15:14>>	PS<13:12>	;set previous mode
SP-2>	SP	; pushed stack selected by new PS
temp1>	M[SP]	; push old PS on stack, data space
SP-2>	SP	-
temp2>	M[SP]	;push old PC on stack, data space ;then execute interrupt service
		;routine

After the interrupt or trap service routine has been completed, an RTI (Return From Interrupt) or RTT (Return From Trap) instruction is typically executed. The top two words of the stack are automatically popped off the stack and placed in the PC and PS, respectively, thereby restoring the state of the interrupted program.

The DCJ11 also responds to a variety of conditions which can abort the current operation. An abort is similar to an interrupt or trap in that a vector is used to point to a service routine. Aborts differ from traps and interrupts in that the DCJ11 services an abort immediately rather than waiting until the end of the current macroinstruction. Aborts generated by the DCJ11 itself include memory management and address errors. Aborts which must be generated by external logic include bus timeouts, non-existent memory accesses, and parity aborts. The signal ABORT is asserted to indicate the presence of an abort condition.

DCJ11 interrupts, traps, and aborts (with their associated priorities) are summarized in Table 1-8. For interrupts and aborts, the name of the signal which initiates the interrupt or abort (if any) appears in the last column. For completeness, Table 1-8 also lists several instructions that result in traps. These instructions are mutually exclusive and have no priority structure.

Table 1-8 Interrupts, Traps, and Aborts

•

Description	Interrupt, Trap, or Abort	Vector Address	Priority Level	Signal
Red stack violation (CPU error register, bit 2)	Abort	4	NM	
Address error (CPU error register, bit 6)	Abort	4	NM	
<pre>Memory management violation (MMR0, bits <15:13>)</pre>	Abort	250	NM	
Timeout/non-existent memory (CPU error register, bits <5:4>)	Abort	4	NM	ABORT
Parity error	Interrupt or Abort	114	NM	PARITY, ABORT
Trace (T bit) set (PSW, bit 4)	Trap	14	NM	
Yellow stack violation (CPU error register, bit 3)	Trap	4	NM	
Power fail (PWRF)	Interrupt	24	NM	PWRF
Floating point exception (FPA present)	Interrupt	244	NM	FPE
Floating point exception (no FPA)	Trap	244	NM	
PIR 7 (PIRQ, bit 15)	Trap	24 0	. 7	
Interrupt level 7	Interrupt	UD	7	IRQ7
EVENT	Interrupt	100	6	EVENT
PIR 6 (PIRQ, bit 14)	Trap	240	6	
Interrupt level 6	Interrupt	UD	6	IRQ6
PIR 5 (PIRQ, bit 13)	Trap	240	5	

Interrupt level 5	Interrupt	UD	5	IRQ5
PIR 4 (PIRQ, bit 12)	Trap	240	4	
Interrupt level 4	Interrupt	UD	4	IRQ4
PIR 3 (PIRQ, bit 11)	Trap	240	3	
PIR 2 (PIRQ, bit 10)	Trap	240	2	
PIR 1 (PIRQ, bit 9)	Trap	240	1	
TRAP Instruction	Trap	34		
EMT Instruction	Trap	30	.	 '
IOT Instruction	Trap	20		
Illegal Instruction	Trap	10		·

NM = Non-maskable UD = User-defined

-- = None

1.5 HALTING DCJ11 OPERATION

A halt operation differs from a interrupt, trap, or abort in that there is no vector associated with it. It is similar, however, in the sense that it interrupts the usual operation of the DCJ11. The two main means of halting the operation of the DCJ11 are to: (1) assert the HALT line or (2) execute a HALT instruction.

The HALT line has a lower priority than any interrupt, trap, or abort. However, it has the highest priority during vector reads. This is to allow the user to break out of potential infinite loops. An infinite loop could occur for example if a vector is not properly mapped during a memory management operation.

Execution of the HALT instruction performs different operations depending upon the CPU operating mode and the halt option currently selected. See Chapter 8 - Interfacing for more details on halt options. In kernel mode, a halt option of 1 causes a trap through location 4 and sets bit 7 of the CPU error register when HALT is executed. If the halt option is 0 in kernel mode, execution of the HALT instruction causes the DCJ11 into console ODT. Execution of the HALT instruction in user or supervisor mode causes a trap through location 4 and sets bit 7 of the CPU error register.

1.6 PROGRAM INTERRUPT REQUEST REGISTER

The program interrupt request register (PIRQ) provides seven levels of software interrupt (i.e., trap) capability. An interrupt request is queued by setting one of bits <15:9>, which to interrupt priority levels 7 correspond through 1 (respectively). Bits <7:5> and <3:1> are set by the DCJ11 to the encoded value of the highest pending request. When the program interrupt request is granted, the processor traps through a vector virtual location 240. It is the responsibility of the at interrupt service routine to clear the appropriate bit in the PIRQ before exiting. The format of the PIRQ is as shown in figure 1-4.



Bits <15:9> can be read or written. Bits <7:5> and <3:1> are read-only. The remaining bits are always read as zeros. PIRQ is cleared by a console start, by a RESET instruction, and at power-up time. The PIRQ resides at physical address 1777772.

1.7 CPU ERROR REGISTER

The CPU error register assists the operating system by identifying the source of a trap through location 4. The CPU error register is located at physical address 1777766. The format of the CPU error register is as shown in Figure 1-5.



Figure 1-5 CPU Error Register

Description Name Bit _ _ _ <15:8> Unused These bits are unused and are always read as zeros. 7 Set when execution of a HALT instruction Illegal HALT is attempted in user or supervisor mode, (Read only) or in kernel mode when the HALT option is enabled (refer to the power-up options in Paragraph 8.3.3). Address Error Set when a word access is made to an odd 6 byte address, or when an instruction (Read only) fetch from an internal register is attempted. Set when reference is made to a 5 Non-Existent non-existent memory address. Memory (Read only) I/O Bus Set when reference is made to a 4 non-existent I/O page address. Timeout (Read only) 3 Yellow Stack Set when a yellow zone stack Trap overflow trap occurs. (Read only) 2 Red Stack Trap Set when a red stack trap occurs. (Read only) <1:0> Unused These bits are unused and are always read as zeros. The CPU error register is cleared by any write reference to itself, by a power-up, or by a console start. The RESET

1.8 STACK PROTECTION

instruction has no effect on this register.

The DCJ11 provides hardware protection for the kernel stack. The supervisor and user stacks are not protected by hardware but may be checked by memory management and appropriate software.

Stack protection in kernel mode is provided by defining yellow and red stack traps. Kernel stack references are checked against a fixed limit of 400 (octal). If the virtual address of a kernel stack reference is less than 400 (octal), a yellow stack trap occurs at the end of the current instruction. A stack trap can only occur on a kernel stack reference, which is defined as: any trap or interrupt push on the kernel stack, a JSR instruction in kernel mode, or a reference in kernel mode using addressing Mode 4

or 5 with R6 as the selected register.

The DCJ11 also checks for kernel stack aborts during interrupt, trap, or abort sequences. If an abort is caused by a kernel stack push during an interrupt, a trap, or an abort sequence, the DCJ11 initiates a red stack trap by creating an emergency stack at vector locations 0 and 2, vectoring through location 4, and setting bit 2 of the CPU error register.

1.9 FLOATING-POINT PROCESSING

The DCJ11 contains an integral floating-point processor which can perform single- and double-precision floating-point operations. User-accessible architecture associated with floating-point processing includes: six 64-bit floating-point accumulators (ACO--AC5), a floating-point status register (FPS), a floating-point exception address (FEA) register, and a floating-point exception code (FEC) register. Chapter 7 describes these in detail and provides information on programming with floating-point instructions.

1.10 MEMORY SYSTEM REGISTERS

Memory system registers are used for: (1) cache memory implementation and (2) memory management.

The memory system registers associated with cache memory are the cache control register (CCR) and the hit/miss register (HMR). These registers are described in detail in Chapter 5 - Special Features.

The memory system registers associated with memory management include page address registers (PARs), page descriptor registers (PDRs), and memory management registers 0, 1, 2, and 3 (MMR0, MMR1, MMR2, MMR3). These are described in detail in Chapter 4 -Memory Management.

1.11 DIRECT-MEMORY ACCESS (DMA) MECHANISM

An external device typically performs a DMA transfer by taking control of a buffered version of the DCJ11's data/address bus (DAL<21:00>). A device requests control of the DAL lines by asserting the DMR input to the DCJ11. This causes the DCJ11 to place DAL<15:00> in a high impedance state (DAL<21:16> is placed in a high impedance state via external buffers) and extend the current microcycle. It is the responsibility of external logic to end the microcycle by asserting the DCJ11's CONT input.

The DCJ11 acknowledges a DMA request by asserting its MAP output

1-17

at the appropriate time. See Chapter 3 - Bus Cycles for the specific timing involved. This also causes the current microcycle to extend until CONT is asserted.

A DMA request may be acknowledged and granted for all types of microcycles except bus writes and GP writes. The lack of a DMA grant, however, does not necessarily prevent external logic from performing a DMA transfer during these cycles. A buffered version of the DAL for example could be used for a DMA transfer when SCTL is asserted (the DAL itself would not be used since it carries the write data during this portion of the cycle).

NOTE

It is possible to acknowledge a DMA request between the read and write portions of a bus locked Read-Modify-Write cycle (see Paragraph 3.2). If this is not desirable, external logic should be designed to disable DMA requests at this time.

2.1 INTRODUCTION

This chapter describes the functions performed by each DCJ11 pin. The pins, and thus the chapter, are divided into nine groups:

- o Data/address lines (DAL<21:00>)
- o System control lines (BS<1:0>, AIO<3:0>, BUFCTL, CONT, DV)
 o Timing signals (ALE, SCTL, STRB, CLK, CLK2)
- o Start/stop control (INIT, HALT)
- Status signals (MISS, PARITY, ABORT, MAP, PRDC) 0
- Interrupt and DMA control (IRQ<3:0>, DMR, PWRF, FPE, EVENT) Test pins (TEST1, TEST2) 0
- 0
- o Oscillator pins (XTLI, XTLO)
- o Power pins (Vcc, GND)

Figure 2-1 illustrates the pin assignments of the DCJ11 and indicates whether a signal associated with a pin is an input, an output, or both (bidirectional).

TEST1 AIO 0 AIO 1 AIO 2 AIO 3 PWRF FPE EVENT	1 2 3 4 5 6 6 7 8	60 59 58 57 56 55 54 53	* * * * * * * * *	DAL 6 DAL 7 DAL 8 DAL 0 DAL 9 DAL 10 DAL 11 DAL 12 DAL 13
VCC BS 0 BS 1 MAP ABORT DAL 21 DAL 20 DAL 19 DAL 19 DAL 18 DAL 17 DAL 16 DMR MISS PRDC NOT USED	16 17 18 19 20 21 22 23 24 25 26 27 28 29 30	311 45 44 43 42 41 40 39 38 37 36 35 34 33 32 31 32 31		GND DAL 4 DAL 5 DV BUFCTL ALE STRB SCTL XTAL0 XTAL0 XTAL1 CLK CLK2 INIT CONT TEST2

Figure 2-1 DCJ11 Pin Assignments

MR 8889

2.2 DATA/ADDRESS LINES (DAL<21:00>)

There are 22 pins associated with data and address information. These are usually referred to as the data/address (or DAL) lines. The DAL lines are functionally divided into two groups: the upper data/address lines (DAL<21:16>) which are output only and the lower data/address lines (DAL<15:00>) which are bidirectional.

2.2.1 Upper Data/Address Lines (DAL<21:16>) - These six time-multiplexed output lines constitute the most significant 6 bits of a 22-bit physical address. DAL<21:16> carries valid information at the beginning of every bus cycle. Internal status is asserted on these lines during the second part of every bus cycle for manufacturing test purposes only.

2.2.2 Lower Data/Address Lines (DAL<15:00>) - These time-multiplexed I/O lines constitute the 16-bit data and address bus. During the first part of a cycle that involves an I/O transfer, the DAL lines carry a physical address, an interrupt acknowledge priority level, or a general-purpose (GP) code, depending upon the type of cycle being performed (see Chapter 3 -Bus Cycles for more information on cycle types). During a Bus Read or Bus Write cycle, DAL<15:00> carries the lower 16 bits of a physical address. During an Interrupt Acknowledge cycle, DAL<3:0> carries the priority of the acknowledged level. During a General-Purpose Read or General-Purpose Write cycle, DAL<7:0> carries the GP code.

During the second part of a cycle that involves an I/O transfer, the DAL lines carry 8 or 16 bits of data. During read cycles, external logic places data onto the DAL. If the DCJ11 only requires a byte of information, it reads a full word but ignores either the upper of lower byte. For write cycles, the DAL carries 8 or 16 bits of data, depending upon whether the cycle involves the writing of a byte or a word.

2.3 SYSTEM CONTROL LINES,

There are nine pins associated with system control: BS<1:0>, AIO<3:0>, BUFCTL, CONT, and DV.

2.3.1 Bank Select (BS<1:0>) - These time-multiplexed output signals transmit bank select and cache access information. At the beginning of a Bus Read or Bus Write cycle, the BS signals define the type of device being accessed by the physical address on the DAL as shown in Table 2-1.

Table 2-1 BS Device Selection

BS1	BS0	DESCRIPTION
1	1	Internal register - A memory-addressable register that resides within the DCJ11. Included are the processor status word, all MMU registers, the PIRQ register, the CPU error register and the cache hit/miss register. Excluded are the general-purpose registers, which are not memory addressable.
1	0	External I/O device - Any device or register external to the DCJ11 that is referenced by a bus address in the upper 8K bytes of the physical address range (17760000 to 17777777). Excluded are system registers (BS code 01) and internal registers (BS code 11).
0	1	System register -

System register -A memory-addressable register in the address range 17777740 to 17777750. Always included as a system register is the DCJ11's internal cache control register (CCR).

NOTE

The CCR is the only system register implemented in the DCJ11. Accesses to the CCR generate the same BS code as for the other system registers mentioned above. This facilitates the creation of "shadow" read-only copies of the CCR on cache based systems.

0

0

Memory -

A reference to any location in physical address space in the range 00000000 to 17757777.

During the second part of an I/O cycle, BS1 is asserted when the cache memory (if present) is to be bypassed. In the second part of the cycle, BS0 is asserted whenever a cache memory force miss is required.

2.3.2 Address Input/Output (AIO<3:0>) - The AIO outputs identify the type of cycle currently being executed. External logic typically latches and decodes these signals. Table 2-2 specifies the AIO code associated with each cycle type. See Chapter 3 - Bus Cycles for detailed information on the various cycle types.

Table 2-2 AIO Decode

AIO3	A102	AI01	A100	CYCLE TYPE
1	1	1	1	NIO (internal operation only, no I/O)
1	1	1	0	GP (General-Purpose) read
1	1	0	1	Interrupt acknowledge, vector read
1	1	0	0	Instruction-stream request read
1	0	1	1	Read/Modify/Write - no bus lock
1	0	1	0	Read/Modify/Write - bus lock
1	0	0	1	Data-stream read
1	0	0	0	Instruction-stream demand read
0	1	0	1	GP word write
0	0	1	1	Bus byte write
0	0	0	1	Bus word write

2.3.3 Buffer Control (BUFCTL) - The BUFCTL output defines whether the DCJ11 is driving or receiving data on the DAL. BUFCTL is typically used by external logic to control the direction of data passing through buffers that send data to the DCJ11. When asserted, BUFCTL indicates that the DCJ11 is not driving data on the DAL. This occurs: (1) during the portion of a read cycle when data is being driven on the DAL, and (2) during the stretched portion of any nonwrite cycle. BUFCTL is deasserted when the DCJ11 is driving data or an address on the DAL.

2.3.4 Continue (\overrightarrow{CONT}) - The \overrightarrow{CONT} input is asserted by external logic to terminate a stretched cycle after it has finished using the DAL. CONT is so named because it enables the DCJ11 to continue on to the next cycle.

2.3.5 Data Valid (DV) - The DV input is typically asserted by external logic to latch data into the DCJ11 from the DAL. When asserted, DV causes the DCJ11 to latch data when BUFCTL and SCTL are asserted, that is, during stretched non-write cycles. External logic must ensure that DV is not asserted during DMA transactions, since this would cause the latching of unpredictable data.

2.4 TIMING SIGNALS

There are five pins associated with timing and synchronization: ALE, SCTL, STRB, CLK, and CLK2.

2.4.1 Address Latch Enable $(\overline{ALE}) - \overline{ALE}$ when asserted indicates that DAL<21:00>, AIO<3:0>, BS<1:0>, and MAP all contain valid data. The leading edge of ALE is typically used by external logic to latch addresses, AIO codes, bank select (BS) codes, and the map enable (MAP) control signal.

2.4.2 Stretch Control (\overline{SCTL}) - The \overline{SCTL} output, when asserted, identifies the stretched portion of a cycle. During write cycles, the leading or trailing edge of \overline{SCTL} can be used for latching data. During read cycles, the trailing edge of \overline{SCTL} can be used for latching data. \overline{SCTL} can also be used to determine when externally generated aborts may occur.

2.4.3 Strobe (STRB) - The assertion of the STRB output occurs one clock period after the assertion of ALE. The deassertion of STRB identifies the end of one microcycle and the beginning of another. STRB is a general-purpose strobe signal and is typically used for system bus control.

2.4.4 Clock 1 (CLK) - CLK is usually a clock output for diagnostic use only. When used as an output, CLK reflects the state of the DCJ11's internal clock. The frequency of CLK equals the frequency of the external crystal oscillator circuit connected to the XTALI and XTALO pins. If TEST2 is asserted, the DCJ11's internal clock is disabled and CLK is placed in the high-impedance state. In this case, CLK can serve as a MOS input ($V_{TL} = .3Vcc$, $V_{TH} = .7Vcc$, $t_{LH} = t_{HL} = 7$ ns) driven by an external clock.

2.4.5 Clock 2 (CLK2) - The CLK2 output has the same frequency as CLK. Like CLK, CLK2 reflects the state of the DCJ11's internal clock and is disabled by the assertion of TEST2. Unlike CLK, CLK2 is typically used as a system clock or master clock for external logic. CLK and CLK2 have minimal skew when loaded equally.

2.5 START/STOP CONTROL

There are two pins associated with starting and stopping the operation of the DCJ11: INIT and HALT.

2.5.1 Initialize (INIT) - The INIT input, when asserted, initializes (resets) the DCJ11 by forcing it through a power-up procedure. The power-up sequence is described in detail in Paragraph 8.3.2.
2.5.2 Halt (HALT) - The HALT input, when asserted, forces the DCJ11 into console mode (i.e., initiates console ODT). HALT is the lowest priority nonmaskable interrupt except during vector read cycles. During vector read cycles, HALT becomes the highest priority non-maskable interrupt. This allows escape from potential infinite looping which could result from programming errors. Since it is non-maskable, HALT is unaffected by the CPU priority specified by PS<7:5>. See Chapter 1 - Architecture for a list of the non-maskable interrupts and their relative priorities. See Chapter 5 - Special Features for a description of console ODT.

2.6 STATUS SIGNALS

There are five pins associated with indicating DCJ11 status: MISS, PARITY, ABORT, MAP, and PRDC.

2.6.1 Cache Miss (MISS) - The MISS input is generated by external logic in DCJ11 based systems incorporating cache memory. The assertion of MISS typically indicates that the current memory reference resulted in a cache memory miss. If MISS is asserted during the first part of a bus read cycle, the cycle is stretched.

2.6.2 Parity Error (PARITY) - The assertion of the PARITY input indicates the occurrence of a memory parity error. PARITY is used to generate parity aborts and parity interrupts. If PARITY is asserted and ABORT is also asserted, then a parity error abort is generated. The DCJ11 immediately traps through a vector located at virtual address 114 without completing the current instruction. If PARITY is asserted but ABORT is not asserted, then a parity error interrupt is generated. At the end of the current instruction, the interrupt is serviced through the vector located at virtual address 114. Note that PARITY is sampled only during the stretched portion of a cycle.

2.6.3 Abort (ABORT) - ABORT can serve as an input or an output of the DCJ11. ABORT is typically configured in an open-collector driver circuit such that aborts generated by either external logic or the DCJ11 can cause ABORT to be asserted (i.e., a wired OR arrangement). Note that the DCJ11 pulls ABORT high internally.

The DCJ11 asserts ABORT during the first part of an I/O cycle if a memory management error or address error occurs. For a memory management error, the DCJ11 traps through a vector located at virtual address 250 in kernel data space. For an address error, the DCJ11 traps through a vector located at virtual address 4 in kernel data space. The DCJ11 sets the appropriate bit in the CPU error register.

ABORT can also be asserted by external logic in the event of such conditions as a bus timeout, non-existent memory reference, parity

error, etc. External logic must ensure that: (1) the cycle is stretched and that ABORT is asserted during the stretched portion (i.e., when SCTL is asserted) and (2) ABORT is not asserted during a non-I/O cycle. If PARITY is not asserted, the assertion of ABORT by external logic causes a trap through a vector located at virtual address 4 in kernel data space. The CPU error register specifies the cause of the abort. If PARITY and ABORT are asserted, the DCJ11 immediately performs a trap through a vector located at virtual address 114 in virtual address space.

2.6.4 Map Enable $(\overline{MAP}) - \overline{MAP}$ is a time-multiplexed output. The assertion of MAP during the first part of a cycle indicates that the I/O map has been enabled (the I/O map is enabled by setting bit 5 of MMR3 to 1). The assertion of MAP during the second part of a cycle acknowledges the assertion of the DMR input.

NOTE The I/O map, if needed, is implemented in circuitry external to the DCJ11.

2.6.5 Predecode (PRDC) - The PRDC output, when asserted, indicates that the contents of the prefetch buffer (PB) are being decoded as the next macroinstruction. This implies that the contents of the PB are valid. The PB is part of the DCJ11 prefetch pipeline, the operation of which is explained in Chapter 5 - Special Features.

2.7 INTERRUPT AND DMA CONTROL

There are eight pins associated with the control of program interrupts and DMA transfers: IRQ<3:0>, DMR, PWRF, FPE, and EVENT.

2.7.1 Interrupt Request (IRQ<3:0>) - IRQ<3:0> are four input lines that correspond to four different levels of external interrupt requests. Interrupt requests at any of these four levels can be masked by PS<7:5>. In order to be serviced, the requesting device must have an interrupt priority higher than the priority indicated by PS<7:5>. Interrupt requests on IRQ<3:0> are blocked or allowed as summarized in Table 2-3:

Table 2-3 Interrrupt Requests on IRQ<3:0>

	Priority				
PS<7:5>	Level	IRQ3	IRQ2	IRQ1	IRQ0
111	7	Blocked	Blocked	Blocked	Blocked
110	6	Allowed	Blocked	Blocked	Blocked
101	5	Allowed	Allowed	Blocked	Blocked
100	4	Allowed	Allowed	Allowed	Blocked
0xx	3-0	Allowed	Allowed	Allowed	Allowed

x = Irrelevant

CDII

From Table 2-3, it is seen that each IRQ line is associated with a different interrupt level, as summarized in Table 2-4.

Table 2-4 IRQ<3:0> Interrupt Request Levels

	Interrupt
IRQ Line	Request Level
IRQ3	7
IRQ2	6
IRQ1	5
IRQO	4

2.7.2 Direct Memory Access Request (\overline{DMR}) - The \overline{DMR} input to the DCJ11 when asserted typically means that an external device wants to perform a DMA transaction. \overline{DMR} is sampled by the DCJ11 at the start of all cycles. If the cycle does not involve a write operation, the DCJ11 responds to the assertion of \overline{DMR} by: (1) stretching the cycle, (2) placing DAL<15:00> in the high-impedance state, and (3) acknowledging the DMA request by asserting \overline{MAP} during the second part of the cycle. If the cycle involves a write operation, the cycle is stretched but DAL<15:00> is not placed in the high-impedance state and \overline{MAP} is not asserted.

2.7.3 Power Fail (PWRF) - PWRF is a high-priority nonmaskable interrupt input that, when asserted, forces a trap through a vector located at virtual address 24 in kernel data space. External logic typically asserts PWRF to indicate the occurrence of an AC power failure. The trap vector points to an appropriate user-defined power fail service routine.

2.7.4 Floating-Point Exception (FPE) - FPE is a high-priority nonmaskable interrupt input that, when asserted, forces a trap through a vector located at virtual address 244 in kernel data space. FPE would be asserted by an external FPA coprocessor to indicate the occurrence of a floating-point exception. The trap vector would point to an appropriate user-defined floating-point exception service routine. **2.7.5 Event (EVENT)** - The EVENT input is a maskable priority level 6 interrupt (i.e., it is acknowledged if PS<7:5> is less than 6). When EVENT is asserted (and not masked), the DCJ11 performs a trap through a vector located at virtual address 100 in kernel data space. EVENT is typically used by external logic as a line time clock (LTC) interrupt input.

2.8 TEST PINS

There are two pins associated with testing, TESTI and TEST2. These signals disable DCJ11 functions and are are used in connection with board-level testing.

2.8.1 Test 1 (TEST1) - The TEST1 input (when asserted by external logic) disables all DCJ11 outputs by placing them in the high-impedance state. This permits external logic to operate on the data and control lines connected to the DCJ11 without interference from the DCJ11.

2.8.2 Test 2 (TEST2) - The TEST2 input, when asserted, disables the DCJ11's internal clock. The CLK and CLK2 pins are placed in the high-impedance state. Board level in-circuit testing logic can be designed such that when TEST2 is asserted, an external clock drives the DCJ11 clock circuitry through the CLK pin.

2.9 OSCILLATOR PINS (XTALI, XTALO)

The XTALI and XTALO pins are used to connect an external crystal circuit to the DCJ11. The recommended crystal circuit is shown in Figure 2-2.



Figure 2-2 Typical XTALI and XTALO Generation

2.10 POWER PINS

There are four pins associated with power: two for +5VDC (Vcc) and two for ground (GND).

2.10.1 Power (Vcc) - There are two pins, both called Vcc, which are used to input +5VDC to the DCJ11. +5VDC is supplied by external circuitry and is typically maintained to within \pm 5%.

2.10.2 Ground (GND) - The two GND pins provide a ground reference for the DCJ11. Typically, these pins are connected to the ground reference of external logic.

.

2.11 PIN DESCRIPTION SUMMARY

PIN NO.	PIN NAME	DEFINITION	INPUT OR OUTPUT	FUNCTION
1	TESTI	Test l	Input	Disables all DCJ11 outputs.
2-5	AIO<3:0>	Address Input/Output	Output	Indicate the type of cycle currently being executed (e.g., bus read, GP write, IACK, etc.)
6	PWRF	Power Fail	Input	A high-priority non- maskable interrupt that forces a trap through vector location 24. Indicates an AC power failure.
7	FPE	Floating-Point Exception	Input	A high-priority non- maskable interrupt that forces a trap through vector location 244. Typically generated by a floating-point coprocessor to indicate an exception condition.
8	EVENT	Event	Input	A maskable interrupt that forces a trap through vector location 100. Typically used as a line time clock.

9	HALT	Halt	Input	A low-priority non- maskable interrupt that forces the DCJll into console ODT.
10-13	IRQ<3:0>	Interrupt Request	Input	Four maskable interrupt request lines.
14	PARITY	Parity Error	Input	Indicates a memory parity error.
15	GND	Ground	Input	Ground reference.
16	Vcc	Power	Input	+5 VDC power input.
17-18	BS<1:0>	Bank Select	Output	Multiplexed. Either define the type of physical address on the DAL or indicate if a cache memory bypass or force miss should occur.
19	MAP	Map Enable	Output	Multiplexed, indicates that either the I/O map is enabled or a DMA request has been granted.
20	ABORT	Abort	1/0	Indicates the occurrence of an abort condition, i.e.; a memory management or address error, bus timeout, non-existent memory, or parity error.
21-26	DAL<21:16>	Data/Address Lines	Output	Most significant six bits of the time multiplexed data and address bus.
27	DMR	Direct Memory Access Request	Input	Forces the current cycle to be extended and causes MAP to be asserted during the second part of the cycle.

ŧ

ŕ

28	MISS	Cache Miss	Input	Indicates whether the current memory reference resulted in a cache hit or miss.
29	PRDC	Predecode	Output	Indicates when the contents of the prefetch buffer are being decoded as the next macroinstruction.
30	Not Used			, ,
31	TEST2	Test 2	Input	Disables the clock outputs. Permits external logic to drive the DCJll's internal clock circuitry through the CLK pin.
32	CONT	Continue	Input	Terminates a stretched cycle.
33	INIT	Initialize	Input	Initializes or resets the system by forcing it through a power-up procedure.
34	CLK2	Clock 2	Output	Clock output with the same frequency as CLK. Typically used as a system clock.
35	CLK	Clock l	Output	Clock output for diagnostic use only.
36	XTALI	Crystal Input	Input	Oscillator input line.
37	XTALO	Crystal Output	Output	Oscillator output line.
38	SCTL	Stretch Control	Output	Indicates that a cycle is being stretched. The edges can be used to strobe data.
39	STRB	Strobe	Output	General-purpose strobe.
40	ALE	Address Latch Enable	Output	Typically used to latch addresses, AIO codes, and the map enable and BS control signals.

41	BUFCTL	Buffer Control	Output	Indicates the direction of data on the DAL. Asserted when the DCJ11 is not driving the DAL.
42	DV	Data Valid	Input	Causes the DCJll to to latch data from the DAL.
43-44, 47-6 0	DAL<15:00>	Data/Address Lines	1/0	Lower 16 bits of the time multiplexed data and address bus.
45	GND	Ground	Input	Ground reference.
46	Vcc	Power	Input	+5 VDC power input.

i

1

۰. ۱ , •

3.1 INTRODUCTION

This chapter describes the various types of DCJ11 bus cycles. A bus cycle is a sequence of events which defines the activity on the DCJ11's I/O bus. Bus cycles are also sometimes referred to as "microcycles", since each bus cycle is associated with the execution of one microinstruction. The execution of a DCJ11 macroinstruction such as ADD, JMP, etc., can involve the execution of several bus cycles. The type of bus cycle that the DCJ11 performs depends upon the type of bus activity (if any) required to complete the execution of a microinstruction.

Sometimes the DCJ11 performs an internal operation which requires no bus activity. If this is the case, the DCJ11 executes a non-I/O (NIO) cycle. An NIO bus cycle (described in detail in Paragraph 3.4) is the only type of bus cycle that does not involve the transfer of information over the DCJ11's I/O bus.

DCJ11 bus cycles fall into six broad categories:

- 1. Non-I/O
- 2. Bus Read
- 3. Bus Write
- 4. General-Purpose Read
- 5. General-Purpose Write

6. Interrupt Acknowledge

The deassertion of the signal STRB marks the beginning (and the end) of a bus cycle. ALE (asserted shortly after STRB is deasserted) can be used by external logic to latch AIO<3:0>. The information on AIO<3:0> specifies the type of bus cycle being performed according to Table 3-1:

Table 3-1 AIO Codes for Bus Cycles

AIO<3:0>	Description	Bus Cycle Type
1111	Non-I/O operation	Non-I/O
1110	GP read	General-Purpose Read
1101	Interrupt acknowledge/ vector read	Interrupt Acknowledge
1100	Instruction stream request read	Bus Read
1011	Read-Modify-Write, no bus lock	Bus Read*
1010	Read-Modify-Write, bus lock	Bus Read*
1001	Data stream read	Bus Read
1000	Instruction stream demand read	Bus Read
0101	GP word write	General-Purpose Write
0011	Bus byte write	Bus Write
0001	Bus word write	Bus Write

* Note that the AIO codes for read-modify-write cycles are identified as Bus Read cycles. This refers to the first part of the cycle (i.e., the "read" part). The second part of the cycle (i.e., the "write" part) will be a Bus Write cycle with a different AIO code.

3.2 DURATION OF BUS CYCLES

The length of a bus cycle is usually expressed as a number of periods of the DCJ11's master clock (CLK). All bus cycles last for a minimum of four clock periods. However, cycles may be extended or "stretched" beyond this minimum by an internal event or by external logic. When a cycle is stretched, it is always stretched for a minimum of four additional clock periods. A cycle can continue to be stretched in increments of two periods and can remain stretched indefinitely. Stretched cycles are ended by the assertion of the signal CONT. CONT is sampled by the DCJ11 on the first falling edge of T4 and on every other succeeding falling edge of T4.

A bus cycle will be stretched unless either of the following two groups of conditions exists:

- A Bus Read cycle is executed and BS<1:0> = 00 throughout the cycle (i.e., the cycle involves a memory read and does not involve a cache bypass or force miss) and DMR and MISS are not asserted during the cycle (no DMA grant or cache miss). Furthermore, ABORT must not be asserted if the cycle involves an instruction stream demand read.
- 2. A Non-I/O cycle is executed and DMR is not asserted during the cycle.

Timing diagrams for both stretched and non-stretched cycles are provided in the paragraphs that follow.

3.3 Bus Cycle Parts

Reference is sometimes made to the "first" (or "early") part and the "second" (or "later") part of a bus cycle. The first part of a bus cycle is defined as the duration of the first two clock periods, shown as T0 and T1 in the bus cycle timing diagrams. The second part of a bus cycle is defined as the duration of the remaining clock periods in the cycle. A non-stretched cycle has only two clock periods in its second part. These are shown as T2 and T3 in the bus cycle timing diagrams. A stretched cycle has at least six clock periods in its second part. These are shown as T2 through T7 in the bus cycle timing diagrams. Note that if a cycle is stretched for more than six clock periods in its second part, T4 is repeated in pairs.

3.4 NON-I/O (NIO) CYCLE

When the DCJ11 executes a microinstruction which involves no interaction with external logic (i.e., requires no I/O bus activity), it performs a Non-I/O (or NIO) cycle. Non-stretched and stretched Non-I/O cycles are illustrated in Figures 3-1 and 3-2, respectively.



Figure 3-1 Non-Stretched Non-I/O Cycle



Figure 3-2 Stretched Non-I/O Cycle

The deassertion of $\overline{\text{STRB}}$ marks the beginning of the cycle, which is followed shortly afterwards by the assertion of ALE. ALE typically latches the AIO code which identifies the cycle as non-I/O. The DAL, BS<1:0>, MAP, and ABORT outputs are undefined and should be ignored by external logic. External logic must not assert ABORT during an NIO cycle. If a direct memory access request (DMR) is granted, the cycle is stretched and SCTL and BUFCTL are asserted.

As shown in Figure 3-1, a non-stretched NIO cycle is four clock periods in duration. If a DMA request is received during the first part of the cycle the cycle is stretched to eight or more clock periods (note the assertion of DMR during the first part of the cycle in Figure 3-2). Otherwise, the cycle does not stretch. If the NIO cycle is stretched, BUFCTL and SCTL are asserted during the stretched part of the cycle. The time-multiplexed signal MAP asserted during the second part of the stretched cycle indicates the granting of the DMA request. The cycle continues to be stretched in increments of two clock periods (T4) until CONT is asserted.

3.5 BUS READ CYCLE

The different types of bus read cycles which the DCJ11 can perform include instruction-stream request or demand reads, data-stream reads, and the read portion of a read/modify/write cycle. The AIO code defines which of these is selected. The types of devices from which information can be read include memory, I/O devices, and explicitly addressable registers. During the first part of the cycle, BS<1:0> defines which of these is selected. All read cycles involve the reading of a full word. If the DCJ11 needs only a byte, it reads a word and ignores the unused byte. Note the distinction between request reads and demand reads. A request read occurs when the DCJ11 is prefetching information. If an abort occurs at this time, it does not affect macroinstruction flow (i.e., aborts are ignored). All other types of reads are demand reads, during which aborts are recognized and serviced via the service vectors shown in Table 1-8.

Non-stretched and stretched Bus Read cycles are illustrated in Figure 3-3 and 3-4, respectively.



Figure 3-4 Stretched Bus Read Cycle

3-5

ALE can be used to latch the AIO code, the physical address on the data/address lines (DAL), the Bank Select (BS) information, and I/O Map Enable (MAP) information.

A Bus Read cycle will stretch if any of the following conditions exist:

- BS<1:0> does not equal 00 during the first part of the cycle (anything other than a memory reference)
- BS<1:0> does not equal 00 during the second part of the cycle (a cache memory force miss or a cache bypass)
- o MAP is asserted during the second part of the cycle (a DMA grant)
- MISS is asserted during the second part of the cycle (a cache miss)
- ABORT is asserted by the DCJ11 during an instruction stream demand read, data stream read, or read-modify-write cycle

Otherwise, a Bus Read cycle will execute in four clock periods.

For non-stretched Bus Read cycles, the read data is synchronously latched into the DCJ11 only on the rising edge of T3, as shown in Figure 3-3.

For stretched Bus Read cycles, data is latched into the DCJ11 both at the rising edge of T3 and when DV is asserted during the stretched portion of the cycle (see Figure 3-4). Thus if read data is valid at the rising edge of T3, it is latched at that time and DV is not required. If the read data is not valid at the rising edge of T3, DV is required to latch the valid data. Note that DV should be inhibited if the stretched Bus Read is due only to a DMA grant.

A stretched cycle lasts at least eight clock periods. A cycle is stretched in increments of two clock periods (T4) and is ended by the assertion of CONT.

If an internally generated abort condition such as an MMU error or address error exists, the DCJ11 asserts ABORT during the first part of the cycle. If this type of abort occurs, the DAL, BS, and MAP information should be ignored for the remainder of the cycle. If an abort is externally generated (such as bus timeout, non-existent memory reference, etc.), it must occur during the stretched portion of the cycle.

3.6 BUS WRITE CYCLE

There are two different types of bus write cycles: Bus Word Write cycles and Bus Byte Write cycles. The AIO code defines which of these is selected. The types of devices to which information can be written include memory, I/O devices, and bus addressable registers. During the first part of the cycle, BS<1:0> defines

which of these is selected.

Bus Write cycle timing is illustrated in Figure 3-5. Note that Bus Write cycles are always stretched cycles.



Figure 3-5 Bus Write Cycle

ALE typically latches the AIO code, the physical memory address on the DAL, the BS information, and the I/O map enable signal (MAP).

SCTL is asserted during the stretched portion of the cycle. The write data is valid when SCTL is asserted and the leading and trailing edges of SCTL can be used by external logic to latch this data. BUFCTL is not asserted during Bus Write cycles.

If an MMU error or address error abort occurs, the DCJ11 asserts ABORT during the first part of the cycle. Externally generated aborts must cause ABORT to be asserted during the stretched portion of the cycle.

> NOTE If an abort occurs during the first part of the cycle, the DAL, BS, and MAP information should be ignored for the remainder of the cycle.

During Bus Byte Write cycles, all 16 bits of DAL<15:0> are driven. If the address is even, the correct data is on the low byte. If the address is odd, the correct data is on the high byte. The data on the unused byte is unspecified.

Since a Bus Write cycle is always stretched, CONT must be asserted to end the cycle.

3.7 GENERAL-PURPOSE (GP) READ CYCLE

General-purpose read cycles allow the DCJ11 to read data from non-PDP-11 addressable external logic. A general-purpose read cycle involves the driving of an address on DAL<7:0> (called the general-purpose or GP code) which external logic must decode and respond to. General-purpose read cycles involve the reading of a full word. If the DCJ11 requires only a byte, it reads a word and ignores the unneeded byte. Timing for General-Purpose Read cycles is shown in Figure 3-6.



Figure 3-6 General-Purpose Read Cycle

ALE is typically used to latch the AIO code and the general-purpose code on the DAL. A GP Read is always stretched and thus always lasts a minimum of eight clock periods. The GP (which specifies the source of the read data) is driven onto code DAL<7:0> during the first part of the cycle. At this time, DAL<21:8> should be ignored. The general-purpose read codes are summarized in Table 3-2.

Table 3-2 General-Purpose Read Codes

- Code Function
- 000 Reads the power-up mode, HALT option, FPA option, POK, and boot address. See Chapter 8 - Interfacing for further details.

001 Reads FPA data (if FPA exists).

- 002 Reads the power-up mode, HALT option, FPA option, POK, and boot address, and clears FPA's FPS.
- 003 Acknowledges FPE and reads FEC (floating exception code) register (if FPA exists).

Note that GP Read data is latched into the DCJ11 both at the

rising edge of T3 and when DV is asserted during the stretched portion of the cycle (see Figure 3-6). Thus if the data is valid at the rising edge of T3, it is latched at that time and DV is not required. If the data is not valid at the rising edge of T3, DV is required to latch the valid data. Since a GP Read cycle is stretched, it must be ended by the assertion of CONT.

> NOTE General-Purpose Read cycles can not be aborted by the DCJ11 and should not be aborted by external logic.

3.8 GENERAL-PURPOSE (GP) WRITE CYCLE

General-Purpose Write cycles allow the DCJ11 to write data to non-PDP-11 external logic. A General-Purpose Write cycle involves the driving of an address on DAL<7:0> (called the general-purpose or GP code) which external logic must decode and respond to. GP write cycles involve the writing of either a word or a byte. Timing for General-Purpose Write cycles is shown in Figure 3-7.



ALE is typically used to latch the AIO code and the general-purpose code on the DAL. A GP Write is always stretched and thus always lasts a minimum of eight clock periods. The GP code (which specifies the destination of the write data) is driven onto DAL<7:0> during the first part of the cycle. At this time, DAL<21:8> should be ignored. Table 3-3 provides a summary of the GP Write codes. See Chapter 8 - Interfacing for further details.

3-9

Table 3-3 General-Purpose Write Codes

Code	Function
003	Writes FPA 16-bit data
014	Asserts bus reset signal
034	Indicates exit from console ODT
040	Reserved for future use
100	Acknowledges assertion of EVENT
140	Acknowledges Power Fail
214	Negates bus reset signal
220	Microdiagnostic test 1 passed
224	Microdiagnostic test 2 passed
230	Microdiagnostic test 3 passed
234	Indicates entry into console ODT

SCTL is asserted during the stretched portion of the GP Write cycle. The write data is valid (and can be latched) on the rising or falling edges of SCTL. The write data is driven onto DAL<15:00>. Since a GP Write cycle is always stretched, it must be ended by the assertion of CONT.

> NOTE General-Purpose Write cycles can not be aborted by the DCJ11 and should not be aborted by external logic.

3.9 INTERRUPT ACKNOWLEDGE BUS CYCLE

An Interrupt Acknowledge cycle (also called an Interrupt Vector Read cycle) is performed to service an interrupt request from IRQ<3:0>. Interrupt Acknowledge timing is illustrated in Figure Note that the interrupt request on IRQ<3:0> must be 3-8. deasserted by the end of the cycle.



Interrupt Acknowledge Cycle Figure 3-8

ALE is typically used by external logic to latch the AIO code and the acknowledged interrupt level. The interrupt level acknowledged is driven onto DAL<3:0> at the beginning of the cycle as shown in the table below.

Table 3-4 Interrupt Acknowledgement

DAL<3:0>		IRQ	level	acknowledged
0001	•			4
0010				5
0100				6
1000				7

At this time DAL<21:4>=0.

As shown in Figure 3-8, the interrupt vector address is placed on the DAL by the interrupting device during the second part of the cycle. An Interrupt Acknowledge cycle is always stretched and consists of at least eight clock periods. It is stretched in increments of two clock periods (T4) until the CONT input is asserted, at which time the cycle is ended.

Note that the interrupt vector is latched into the DCJll both at the rising edge of T3 and when DV is asserted during the stretched portion of the cycle. Thus if the interrupt vector is valid at the rising edge of T3, it is latched at that time and DV is not required. If the interrupt vector is not valid at the rising edge of T3, DV is required to latch it.

An Interrupt Acknowledge cycle can be aborted during the stretched portion of the cycle if ABORT is asserted by external logic. The DCJ11 does not assert ABORT during the first part of an Interrupt Acknowledge cycle. If an abort occurs, the DCJ11 ignores the interrupt request and continues execution.

3.10 DMA REQUESTS AND GRANTS

If external logic needs to use the DAL to transfer data, it must: (1) cause the DCJ11 to put the DAL in the high-impedance state, and (2) stretch the cycle currently in progress while external logic makes use of the DAL. This is accomplished by asserting the DMR input during the first part of a cycle. In response, the DMA request will be acknowledged and granted for all cycle types except Bus Write and GP Write cycles. During Write cycles (which are always stretched), the DAL carries write data during the second part of a cycle, during which time the DAL is not placed in the high-impedance state. External logic could be designed such that DMA transfers could occur during Write cycles as long as the DMA transfer did not use the DAL coming directly from the DCJll (a buffered version of the DAL could be used instead). In other words, external logic is not prevented from performing a DMA operation simply because a DMA grant does not occur.

A DMA request is acknowledged by asserting $\overline{\text{MAP}}$ during the second part of a cycle. A cycle involving a DMA transfer is stretched and thus lasts a minimum of eight clock periods. It will continue to be stretched in increments of two clock periods until the CONT input is asserted. Note that the deassertion of DMR does not end the cycle.

4.1 INTRODUCTION

The DCJll contains a memory management unit (MMU) which provides the user with the hardware necessary to effect complete memory management and protection. The MMU is designed to provide access to all of physical memory and is an important part of multi-user, multiprogramming systems where memory protection and relocation facilities are necessary.

The MMU is used to assign segments of memory called pages to a user program and prevent that user from making unauthorized accesses to pages outside his assigned area. A user is thus prevented from accidental or willful destruction of any other user program or the system executive program.

The MMU is usually used in conjunction with a supervisory program which determines how the MMU is to operate. In multiprogramming environments this supervisory program controls the execution of the various user programs, manages the allocation of memory and peripheral device resources, and safeguards the integrity of the system as a whole by careful control of each user program.

The basic characteristics of the DCJ11 memory management unit are: o 16 kernel mode memory pages

- o 16 supervisor mode memory pages
- o 16 user mode memory pages
- o 8 pages in each mode for instructions
- o 8 pages in each mode for data
- o Page lengths from 64 to 8192 bytes
- o Each page provided with full protection and relocation
- o Transparent operation
- o Memory access to 4 million bytes

The remainder of this chapter explains these characteristics in detail.

4.2 ADDRESSING

When the MMU is active, a 16-bit address referenced in a program is interpreted as a virtual address (VA) containing information to be used in constructing a new 22-bit physical address (PA). The information contained in the virtual address is combined with relocation information contained in a register called the page address register (PAR) to yield the 22-bit physical address. Using the MMU, memory can be dynamically allocated in pages composed of from 1 to 128 contiguous blocks of 64 bytes each. Figure 4-1 illustrates the relocation of virtual addresses to physical addresses via page address registers.



PAR = PAGE ADDRESS REGISTER

MR-11462

Figure 4-1 Virtual Address Mapping Into Physical Address

The starting physical address for each page is an integral multiple of 64 bytes, and each page has a maximum size of 8192 bytes. Pages may be located anywhere within the 22-bit physical address space.

Only one set of eight page address registers are illustrated in Figure 4-1. Actually, six such sets of page address registers are used by the MMU. The determination of which set of page registers is enabled at any given time depends on the current CPU mode of operation (i.e., kernel, supervisor, or user mode) and whether the MMU is mapping instructions (into I space) or data (into D space). Refer to Paragraph 4.5 for further details.

4.3 I SPACE AND D SPACE

When the MMU is active, all addresses are mapped into either instruction (I) space or data (D) space. I space is used for all instruction fetches, index words, absolute addresses and immediate operands. D space is used for all other references. I space and D space each have 8 PARs in each mode of CPU operation (kernel, supervisor, and user). Using memory management register #3 (MMR3), D space can be disabled such that all references (instruction and data) are mapped through I space.

Table 4-1 defines how memory references are mapped into the I and D spaces. Note that the determination of whether a memory reference gets mapped into I space or D space depends on: the type of instruction, the addressing mode, and the register selected.

Table 4-1 I and D Space Referencing (first/second/third memory references)

Address Mode and Reg Select	Normal Instruction	MTPI, MFPI	MTPD,MFPD, MFPI
	(not MTP1, MFP1 MTPD, or MFPD)	(PS<15:12> not 1111)	(PS<15:12> = 1111)
00 - 07	na	na	na
10 - 16	D	I	D
17	I	I	D
20 - 26	D	I	D
27	I	I	D
30 - 36	D/D	D/I	D/D
37	I/D	I/I	I/D
40 - 46	D	I	D
47	I	I	D
50 - 56	D/D	D/I	D/D
57	I/D	I/I	I/D
60 - 67	I/D	I/I	I/D
70 - 77	I/D/D	I/D/I	I/D/D

4.4 CONSTRUCTION OF A PHYSICAL ADDRESS

The basic information needed for the construction of a physical address comes from the virtual address (illustrated in Figure 4-2) and the appropriate PAR set.



Figure 4-2 Interpretation of a Virtual Address

The virtual address consists of:

- 1. The active page field (APF). This 3-bit field determines which of eight page address registers (PARO through PAR7) will be used to form the physical address.
- 2. The displacement field (DF). This 13-bit field contains an address relative to the beginning of a page. This permits page lengths up to 8K bytes. The DF is further subdivided into two fields as shown in Figure 4-3.





The displacement field (DF) consists of:

- 1. The block number (BN). This 7-bit field is interpreted as the block number within the current page.
- 2. The displacement in block (DIB). This 6-bit field contains the displacement of the address within the block specified by the block number.

The remainder of the information needed to construct the physical address comes from the 16-bit page address field (PAF) (i.e. the contents of the page address register (PAR)) that specifies the starting address of a particular memory page. The PAF is actually a block number in physical memory, e.g., PAF = 3 indicates a starting address of 192 (3 x 64 bytes per block) decimal or 300 octal in physical memory.



The formation of the physical address is illustrated in Figure 4-4.

Figure 4-4 Construction of a Physical Address

EQUALS

PHYSICAL ADDRESS

The logical sequence involved in constructing a physical address is as follows:

06 05

00

- Select a set of page address registers depending on the CPU mode (kernel, supervisor, or user) and the type of memory reference (I or D space).
- 2. Use the active page field (APF) from the virtual address to select one of eight page address registers (PAR0 through PAR7).

- 3. The page address field (PAF) of the selected page address register (PAR) contains the starting address of the currently active page as a block number in physical memory.
- 4. The block number (BN) from the virtual address is added to the page address field to yield the number of the block in physical memory which will contain the physical address being constructed.
- 5. The displacement in block (DIB) from the displacement field of the virtual address is appended to the physical block number to yield a true 22-bit DCJ11 physical address.

4.5 MANAGEMENT REGISTERS

ş

The DCJ11 MMU implements three sets of 32 16-bit registers as shown in Figure 4-5. One set of registers is used in kernel mode, another in supervisor mode, and the other in user mode. The choice of which set to be used is determined by the current CPU mode contained in the processor status register (PS). Each set consists of two groups of 16 registers. One group is used for references to instruction (I) space and one to data (D) space. The I space group is used for all instruction fetches, index words, absolute addresses, and immediate operands. The D space group is used for all other references, providing D space has not been disabled by memory management register #3. Each group contains 8 pairs of 16-bit registers. Half of the registers in each group are page address registers, which operate as explained previously. The other registers are page descriptor registers (PDRs). PARs and PDRs are always selected in pairs. A PAR/PDR pair contains all the information needed to describe and locate a currently active memory page.

Each of the memory management registers described above are located in the uppermost 8K bytes of the physical address space (see Paragraph 4.9).



Figure 4-5 Active Page Registers

4.5.1 Page Address Registers (PARs) - As shown in Figure 4-6, each page address register contains a 16-bit page address field (PAF) which specifies the starting address of a page as a block number in physical memory.



The page address register which contains the page address field may be thought of as a relocation register containing a relocation constant, or as a base register containing a base address.

4.5.2 Page Descriptor Registers (PDRs) - Page descriptor registers (PDRs) contain information on page expansion direction, page length, and access control. Refer to Figure 4-7.



Figure 4-7 Page Descriptor Register (PDR)

4.5.2.1 Bypass Cache - Bit 15 implements a conditional cache bypass mechanism. If set, references to the selected virtual page can bypass cache memory if a cache is present in the system.

4.5.2.2 Page Length Field (PLF) - This 7-bit field occupying bits <14:8> of the PDR specifies the block number, which defines the boundary of that page. The block number of the virtual address is compared against the page length field to detect length errors. An error occurs when expanding upwards if the block number is greater than the page length field and when expanding downwards if the block number is less than the page length field.

4.5.2.3 Page Written - Bit 6 (the W bit) indicates whether or not this page has been modified (i.e., written into) since either the PAR or PDR was loaded (W = 1 means the page has been modified). The W bit is useful in applications which involve disk swapping and memory overlays. It is used to determine which pages have been modified and hence must be saved in their new form and which pages have not been modified and can simply be overlaid.

Note that the W bit is reset to 0 whenever either PAR or PDR is modified (written into).

4.5.2.4 Expansion Direction (ED) - Bit 3 specifies in which direction the page expands. If ED = 0 the page expands upwards from block number 0 to include blocks with higher addresses; if ED = 1 the page expands downwards from block number 127 to include blocks with lower addresses. Upward expansion is usually used for program space while downward expansion is usually used for stack space.

4-7

4.5.2.5 Access Control Field - This 2-bit field, occupying bits <2:1> of the page descriptor register contains the access rights of a particular page. The access codes or "keys" specify the manner in which a page may be accessed and whether or not a given access should result in an abort of the current operation. A memory reference which causes an abort must not be completed by the system interface. Aborts are used to catch "missing page faults", prevent illegal accesses, etc.

In the context of access control the term "write" is used to indicate the action of any instruction which modifies the contents of any addressable byte. "Write" is synonymous with what is sometimes called a "store" or "modify" in many computer systems.

The modes of access are as follows:

00	non-resident	abort all accesses
01	read-only	abort on write attempt
10	unused	abort all accesses
11	read/write	access

4.5.2.6 Reserved Bits - Bits 7, 5, 4, and 0 are spare and are always read as 0. These bits are reserved for possible future expansion.

4.6 INTERRUPT CONDITIONS UNDER MEMORY MANAGEMENT CONTROL

With the MMU enabled, all trap, abort, and interrupt vectors are considered to be in kernel mode virtual address space. When a trap, abort, or interrupt occurs, control is transferred according to a new program counter (PC) and processor status word (PS) contained in a two-word vector that is relocated through the kernel page address register set. The old PC and PS is pushed onto the R6 stack specified by bits <15:14> of the new PS (00 = kernel, 01 = supervisor, 11 = user). Bits <15:14> also determine the new PAR set. In this manner it is possible for a kernel mode program to have complete control over service assignments for all interrupt conditions since the interrupt vector is located in kernel space. The kernel program may assign the service of a trap, abort, or interrupt condition to a supervisor or user mode program by simply setting bits <15:14> of the new PS.

4.7 FAULT RECOVERY REGISTERS

Aborts generated by the MMU are vectored through kernel virtual location 250. Memory management registers #0, #1, #2, and #3 are used to determine why the abort occurred, and allow for easy program restarting. Note that an abort to a location which is itself an invalid address will cause another abort. Thus the kernel program must insure that kernel virtual address 250 is mapped to a valid address, otherwise a loop will occur which will **4.7.1 Memory Management Register #0 (MMR0) - MMR0** contains error flags, the page number whose reference caused the abort, and various other status flags. The register is organized as shown in Figure 4-8.



Figure 4-8 Memory Management Register #0 (MMR0)

4.7.1.1 Error Flags - Bits <15:13> are error flags. They may be considered to be in a "priority queue" in that flags to the right are less significant and should be ignored if a higher bit is set. That is, a non-resident fault service routine would ignore length and access control faults. A page length fault service routine would ignore access control faults.

Bits <15:13> when set (error conditions) cause the MMU to freeze the contents of MMR0 bits <6:1>, MMR1, and MMR2. This is to facilitate error recovery.

Bits <15:13> may be written under program control. No abort will occur, but the contents of the memory management registers will be frozen as in an abort.

Bits <15:13> are cleared at power-up, by a console start, or by a RESET instruction.

4.7.1.1.1 Abort -- Non-Resident - Bit 15 is set by attempting to access a page with an access control field key equal to 0 or 2. It is also set by attempting to use memory relocation with a processor mode of 2 (i.e., the illegal processor mode).

4.7.1.1.2 Abort -- Page Length - Bit 14 is set by attempting to access a location in a page with a block number (virtual address bits <12:6>) that is outside the area authorized by the page length field of the PDR for that page. Bits 14 and 15 may be set simultaneously by the same access attempt. Bit 14 may also be set by attempting to use memory relocation with a processor mode of 2. **4.7.1.1.3 Abort -- Read Only -** Bit 13 is set by attempting to write in a "read-only" page. Read-only pages have access keys of 01.

4.7.1.2 Reserved Bits - Bits <12:7> are spare and are always read as 0. These bits are reserved for possible future expansion.

4.7.1.3 Processor Mode - Bits <6:5> indicate the CPU mode (kernel, supervisor, or user) associated with the page causing an abort (kernel = 00, supervisor = 01, user = 11, illegal mode = 10). If an illegal mode is specified, bit 15 is set.

4.7.1.4 Page Address Space - Bit 4 indicates the type of mapping (I or D) the MMU attempted when an abort occurred (0 = I space, 1 = D space). It is used in conjunction with bits <3:1>, page number.

4.7.1.5 Page Number - Bits <3:1> contain the page number of a reference causing an MMU abort. Note that pages, like blocks, are numbered from 0 upwards.

4.7.1.6 Enable Relocation - When bit 0 is set to a 1, the MMU is enabled and performs address relocation. When bit 0 is cleared, the MMU is inoperative and addresses are not relocated or protected. Bit 0 is cleared at power-up, by a console start, or by a RESET instruction.

4.7.2 Memory Management Register \$1 (MMR1) - MMR1 (see Figure (4 - 9)records any autoincrement/autodecrement of the general-purpose registers, including references through the PC. This information is necessary to recover from an error resulting in an abort. MMRI is cleared at the beginning of each instruction fetch. Whenever a general-purpose register is autoincremented or autodecremented, the register number and the amount (in 2's complement notation) by which the register was modified is written into MMR1. The low order byte of MMR1 is written first. It is not possible for a DCJ11 instruction to autoincrement/decrement more than two general-purpose registers per instruction before an "abort-causing" reference.

It is up to the software to determine which set of registers (kernel/supervisor/user -- general set 0/general set 1) was modified, by determining the CPU and register modes as contained in the PS at the time of the abort.



Figure 4-9 Memory Management Register #1 (MMR1)

4.7.3 Memory Management Register #2 (MMR2) - MMR2 is loaded with the current 16-bit virtual address at the beginning of each instruction fetch. MMR2 is read-only; it can not be written. MMR2 is the virtual program counter.

4.7.4 Memory Management Register #3 (MMR3) - As shown in Figure 4-10, MMR3 enables or disables the use of D space PARs and PDRs and 22-bit mapping and controls data on the time-multiplexed output MAP (pin 19 of the DCJ11).



Figure 4-10 Memory Management Register #3 (MMR3)

4.7.4.1 Reserved Bits - Bits <15:6> are spare and are always read as 0. These bits are reserved for possible future expansion.

4.7.4.2 Bnable I/O Map - Bit 5 is set to assert the MAP output of the DCJ11. If bit 5 = 1 MAP is asserted. If bit 5 = 0 MAP is unasserted. On initialization, MMR3 is cleared.

4.7.4.3 Enable 22-Bit Mapping - If bit 4 = 0 and the MMU is enabled (bit 0 of MMR0 = 1), the DCJ11 uses 18-bit mapping. If bit 4 = 1 and the MMU is enabled, the DCJ11 uses 22-bit mapping. If the MMU is disabled, bit 4 is ignored and 16-bit mapping is used. Figures 4-11, 4-12, and 4-13 illustrates the three mapping alternatives available.









4-12



Figure 4-13 22-Bit Mapping

4.7.4.4 Enable Call To Supervisor Mode (CSM) Instruction - Bit 3 is used to enable a CSM instruction. If bit 3 is set to a 1, a CSM instruction will execute. If bit 3 = 0, a CSM instruction will cause a trap through vector location 10.

4.7.4.5 Kernel, Supervisor, And User Mode D Space Bits - Bits 2, 1, and 0 are the kernel, supervisor, and user mode D space bits, respectively. These bits determine whether D space mapping is enabled or disabled for each CPU mode. When D space is disabled, all memory references use the I space registers; when D space is enabled, both the I space and the D space registers are used. When a mode bit is set, D space is enabled; when a mode bit is clear, D space is disabled (see Table 4-2).

Table 4-2 Mode Bit Operations

BIT	STATE	OPERATION
2	0 1	Disable kernel D space Enable kernel D space
1	0 1	Disable supervisor D space Enable supervisor D space
0	0 1	Disable user D space Enable user D space

4.7.5 Instruction Back-Up/Restart Recovery - The process of "backing-up" and restarting a partially completed instruction involves:

- 1. Performing the appropriate memory management tasks to alleviate the cause of the abort (e.g., loading a missing page).
- 2. Restoring the general-purpose registers indicated in MMR1 to their original contents at the start of the instruction by subtracting the "modify value" specified in MMR1.
- 3. Restoring the PC to the "abort time" PC by loading R/ with the contents of MMR2, which contains the value of the virtual PC at the time the instruction generating the abort was fetched.

Note that this back-up/restart procedure assumes that the general-purpose register used in the aborted program segment will not be used by the abort recovery routine. This is automatically the case if the recovery program uses a different general register set.

4.7.6 Clearing Status Registers Following Abort - At the end of an abort service routine, bits <15:13> of MMR0 must be set to 0 to resume error checking. On the next memory reference following the clearing of these bits, the various memory management registers will resume monitoring the status of the addressing operations. MMR2 will be loaded with the next instruction address, MMR1 will store register change information, and MMR0 will log MMU status information.

4.7.7 Multiple Faults - Once an abort has occurred, any subsequent errors that occur will not affect the state of the memory management status registers. The information saved in MMRO, MMR1, MMR2, and MMR3 will always refer to the first abort that was detected.

4.8 MMU IMPLEMENTATION

The MMU is a very general purpose memory management tool. It can be used in a manner as simple or as intricate as desired. It can be anything from a simple memory expansion device to a very complete memory management facility.

In most normal applications, it is assumed that control over memory page assignments and their protection resides in a supervisory type program which operates at the nucleus of a CPU's executive (i.e. in kernel mode). It is further assumed that this kernel mode program would set access keys in such a way as to protect itself from willful or accidental destruction by supervisor mode or user mode programs. Facilities are also provided so that the nucleus can dynamically assign memory pages of varying sizes in response to system needs.

4.8.1 Typical Memory Page - When the MMU is enabled, the kernel mode program, a supervisor mode program, and a user mode program each have eight active pages (described by the appropriate PARs and PDRs) for data, and eight for instructions. Each page is made up of from 1 to 128 blocks and is pointed to by the page address field of the corresponding PAR as illustrated in Figure 4-14.



Figure 4-14 Typical Memory Page

The memory segment illustrated in Figure 4-14 has the following attributes:

1. Page length: 40 blocks.

2. Virtual address range: 140000 - 144777.

3. Physical address range: 312000 - 316777.

4. Nothing has been modified (i.e., written) in this page.

5. Read-only protection.
- 6. Upward expansion.
- 7. Cache (if present in the system) is not bypassed.

These attributes were determined according to the following scheme:

- 1. PAR6 and PDR6 were selected by the active page field of the virtual address. (Bits <15:13> of the virtual address = 110)
- 2. The initial address of the page was determined from the page address field of PAR6. (312000 (octal) = 3120 (octal) blocks x 64 (octal) bytes). Note that the PAR which contains the PAF constitutes what is often referred to as a base register containing a base address or a relocation register containing a relocation constant.
- 3. The page length (47 (octal) + 1 = 40 (decimal) blocks) was determined from the page length field contained in PDR6. Any attempts to reference beyond the 40 blocks in this page will cause a page length error which will result in an abort, vectored through kernel virtual address 250.
- 4. The physical addresses were constructed according to the scheme illustrated in Figure 4-4.
- 5. The W bit (W = 0) indicates that no locations in this page have been modified (i.e., written). If an attempt is made to modify any location in this particular page, an access control violation abort will occur. If this page were involved in a disk swapping or memory overlay scheme, the W bit would be used to determine whether it had been modified and thus required saving before overlay.
- 6. This page is read-only protected, i.e. no locations in this page may be modified. The mode of protection is specified by the access control field of PDR6.
- 7. The direction of expansion is upward (ED = 0). If more blocks are required in this segment, they will be added by assigning blocks with higher relative addresses.
- 8. The Bypass Cache bit (bit 15) = 0 which means that cache memory is not bypassed during this relocation operation.

Note that the various attributes which describe this page can all be determined under software control. The parameters describing the page are all loaded into the appropriate PAR and PDR under program control. In a normal application it is assumed that the particular page which itself contains these registers would be assigned to the control of a supervisory type program operating in kernel mode. **4.8.2 Mon-Consecutive Memory Pages -** It should be noted that although the correspondance between virtual addresses and PAR/PDR pairs is such that higher VAs have higher PAR/PDRs, this does not mean that higher virtual addresses necessarily correspond to higher physical addresses. It is quite simple to set up the PAFs of the PARs so that higher virtual address blocks may be located in lower physical address blocks as illustrated in Figure 4-15.



Figure 4-15 Non-Consecutive Memory Pages

Note that although a single memory page must consist of a block of contiguous locations, memory pages do not have to be located in consecutive physical address locations. Also note that the assignment of memory pages is not limited to consecutive non-overlapping physical address locations.

4.8.3 Stack Memory Pages - When constructing DCJ11 programs, it is often desirable to isolate all program variables from program instructions by placing them on a register-indexed stack. These variables can then be pushed or popped from the stack as needed. DCJ11 stacks expand linearly downward to lower addresses when data is pushed onto them. Thus, when a memory page which contains a stack needs more room, it must expand downward. Blocks with lower addresses relative to the current page must be added. This mode of operation is specified by setting the expansion direction (ED) bit of the appropriate PDR to a 1. Figure 4-16 illustrates a typical stack memory page.

4-17



PAR 6 PAF PDR 6 BC PLF W ED ACF

MR 11458

Figure 4-16 Typical Stack Memory Page

This page will have the following parameters:

- o PAR6: PAF = 3120
- o PDR6: PLF = 175 (octal) or 125 (decimal) (128 3).
- o ED = 1
- o W = 0 or 1
- o ACF = n (to be determined by the programmer as the need dictates)

Note: the W bit is set by internal chip hardware.

In this case the stack begins 128 blocks above the relative origin of this memory page and extends downward for a length of three blocks. A page length error abort vectored through kernel virtual address 250 will be generated by the MMU when an attempt is made to reference any location below the assigned area, i.e. when the block number from the virtual address is less than the page length field of the appropriate PDR.

4.8.4 Transparency - In a multiprogramming application memory pages can be allocated such that a particular program seems to have a complete 64K memory configuration. Using relocation, a kernel mode supervisory type program can easily perform all memory management tasks in a manner entirely transparent to a supervisor mode or user mode program. In effect, a DCJ11 system can be configured to provide maximum throughput and response to a variety of users each of which seems to have a powerful system all to himself.

4.9 MEMORY MANAGEMENT UNIT -- REGISTER MAP

REGISTER

į

ADDRESS

Memory Man Memory Man Memory Man Memory Man	agement Register agement Register agement Register agement Register	<pre>#0 (MMR0) #1 (MMR1) #2 (MMR2) #3 (MMR3)</pre>	17777572 17777574 17777576 17772516
User I Spa	ce PDR0		17777600
User I Spa	ce PDR7		17777616
User D Space	ce PDR0		17777620
User D Spac	ce PDR7		17777636
User I Spac	ce PARO		17777640
User I Spac	ce PAR7		17777656
User D Spac	ce PARO		17777660
User D Spac	ce PAR7		17777676
Supervisor	I Space PDR0		17772200
Supervisor	I Space PDR7		17772216
Supervisor	D Space PDR0		17772220
Supervisor	D Space PDR7		17772236
Supervisor .	I Space PAR0		17772240
Supervisor	I Space PAR7	•	17772256

Supervisor D Space PAR0	17772260
•	•
Supervisor D Space PAR7	17772276
Kernel I Space PDR0	17772300
•	•
Kernel I Space PDR7	17772316
Kernel D Space PDR0	17772320
	•
Kernel D Space PDR7	17772336
Kernel I Space PAR0	17772340
•	•
Kernel I Space PAR7	17772356
Kernel D Space PAR0	17772360
•	•
Kernel D Space PAR7	17772376

5.1 INTRODUCTION

This chapter discusses three special features incorporated into the DCJ11: cache memory status and control registers, console ODT, and pipeline processing hardware.

5.2 CACHE MEMORY STATUS AND CONTROL REGISTERS

The DCJ11 contains hardware that allows the user to incorporate cache memory into his system. This hardware consists of the cache control register and the hit/miss register. This hardware allows for a broad spectrum of cache implementations and the user has considerable flexibility in designing a cache memory scheme to fit his application. The paragraphs that follow not only describe the cache memory status and control registers in detail but also present some general considerations involved in designing cache memory into a DCJ11 based system. A sample cache memory implementation is also presented to illustrate a typical application of the cache memory status and control registers.

5.2.1 Cache Control Register - The cache control register (CCR) contains information which is used to control the operation of cache memory. It is accessed by referencing location 1777746. Only bits 9 and <3:2> of the CCR are interpreted by the DCJ11. Bits <10:0> are read/write bits. Bits <15:11,8> are always read as zeroes.

In order for the uninterpreted read/write bits (bits 10, <8:4>, and <1:0> to be used by external logic, the user must include a "shadow register" (write only) in his DCJ11 design. The shadow register simply retains a hardware accessible copy of the CCR information. Although the DCJ11 allows the reading and writing of CCR<10:0> and the writing of CCR<15:11>, changing bits <15:11>, 8, <7:4>, and <1:0> will have no hardware effect on the DCJ11.

CCR bits <15:11> are uninterpreted and always read as zeroes by the DCJ11 (see sample implementation in Paragraph 5.2.5). The user typically designs an external register for these bits if they must be interpreted. The format of the CCR is shown in Figure 5-1.



*Written as a logic 1 at power-up or when console ODT is started

Figure 5-1 Cache Control Register

5.2.1.1 Unconditional Cache Bypass (R/W) - When bit 9 is set to 1, all memory references access main memory, and all cache hits are invalidated.

5.2.1.2 Force Cache Miss (R/W) - When either of bits <3:2> is set to 1, all references are forced to main memory and all cache activity is suspended. This in effect disables the cache system.

5.2.1.3 Uninterpreted Bits - Bits <15:10>, <8:4>, and <1:0> are uninterpreted by the DCJ11. Bits 10, <8:4>, and <1:0> are read/write bits and bits <15:11> are always read as zeroes.

5.2.2 Hit/Miss Register - The Hit/Miss Register (HMR) indicates whether the six most recent CPU memory references resulted in cache hits or cache misses. It is accessed by referencing 1777752. location Refer to Figure 5-2. Bits <15:6> are always read as zeroes. Bits <5:0> are read-only bits. Bits enter from bit 0) and are shifted leftward. the right (at A logical one indicates a cache hit, and a zero indicates a cache miss. This register is used to help diagnose the cache system.



Figure 5-2 Hit/Miss Register

5.2.3 General Operation - Cache memory is typically a high-speed memory that buffers data between the CPU and main memory. When a memory access occurs, the system looks for data in the fast cache memory first. If found (a hit), the data is read or written to or from the cache and execution proceeds at the fastest rate. If not found (a miss), the data must be read from or written to main memory.

In a write-through cache system a CPU request to write data into memory causes data to be written to both the cache and to main memory. This is to insure that both stores are always updated immediately. PDP-11 systems with cache normally use the write-through technique.

Typical hit/miss operations in a write-through cache system are summarized in Table 5-1.

	What Happens	In		
	MAIN MEMORY			
READ hit miss	no change updated	no change no change		
WRITE hit miss	updated no change	updated updated		

Table 5-1: Typical Hit/Miss Operations

In a typical program, WRITES occur only 10-15% of the time and READS occur 85-90% of the time. Thus, READ misses cause the cache to be updated.

The I/O page of physical memory (the top 8K bytes) is not typically cached. This is because the I/O page contains device status registers which, when read, must always convey the latest information.

When a DMA device writes to a cached location, the overwritten cache entry is typically invalidated. The cache system monitors DMA transactions to determine if this action is needed.

There are several design parameters that must be considered when constructing a cache memory, cache size and block size to name but two. A detailed discussion of cache design is beyond the scope of this document, but an introduction to the subject is found in Section VI of the KB11-C Processor Manual (EK-KB11C-TM). An 8 KB direct mapped cache is presented as an implementation example in Paragraph 5.2.5. 5,2.4 Cache Memory In A Multiprocessor Environment - In а multiprocessor system where each processor has its own cache memory, care must be taken to avoid caching data that was A simple invalidated by another processor ("stale" data). software method can prevent this situation. Any shared address must bypass the cache, i.e., the reference must go to main memory, and if the address was previously cached, the entry must be invalidated. The DCJ11 provides three bypass mechanisms: unconditional bypass in which every reference is bypassed; an а conditional bypass in which bypassing is on a page-by-page basis; and finally, a selective bypass in which the bypassing is done during operand references. The unconditional bypass is selected by setting bit 9 of the Cache Control Register (see Paragraph 5.2.1). The conditional bypass is selected when bit 15 of the currently selected Page Descriptor Register PDR is set (see Paragraph 4.5.2). The selective bypass occurs during the operand references of the instructions used in multiprocessing functions (TSTSET, WRTLCK and ASRB).

5.2.5 Sample Implementation - The following is a description of the operation of an 8 Kb direct mapped cache with a block size of two bytes as implemented on a DCJ11 based system. This is only one of many possible implementations.

A direct mapped cache is organized such that each physical memory address is associated with a particular "block" of memory in the cache. In this case we have an 8 KB cache with a block size of two bytes. This means there are 4K blocks in the cache. Each word in physical memory is associated with one of these 4K blocks.

Consider each physical address as being made up of three parts (see Figure 5-3). The first part is bit zero. Bit zero specifies which of the two bytes in a two-byte block is to be accessed. The next part, bits <12:1>, is called the cache index and specifies which of the 4K blocks in the cache is to be accessed. The third part, bits <21:13>, is called the cache tag. One cache tag per block is stored in the cache to uniquely identify physical memory locations.



MR 11437

Figure 5-3 Physical Address Partitioning for Cache Memory

For example, if the DCJ11 accesses location 10002477, cache control logic (designed by the user) looks at the cache tag associated with the information currently in cache block number 1237 (bits <12:1>). If this cache tag is 400 (bits <21:13>), the cache control logic sends both bytes in that block to the DCJ11. Since bit 0 is a 1, the DCJ11 automatically selects the high byte (the low byte is ignored). If the stored cache tag is not 400, the control logic fetches two bytes from memory (10002476 and 10002477), sends 10002477 to the DCJ11, loads the two bytes into cache block 1237, and changes the cache tag of that block to 400.

Any location whose cache index is 1237 will be loaded into block 1237 of cache memory. This is the only place the cache control logic has to look if the DCJ11 accesses the data from a location whose cache index is 1237.

Figure 5-4 illustrates a format for each cache block. The 9-bit cache tag is stored in bits <24:16> and the two bytes of data which comprise the block are stored in bits <15:0>. Bit 25 is a Valid Bit which indicates whether or not this cache block contains valid data. Data would be invalid for example immediately after power-up, and the cache control logic would clear the valid bit in this case.



Figure 5-4 Cache Entry

Notice that only the cache tag of a location need be stored in a cache entry because only the cache tag is required to uniquely identify a location. The cache index need not be compared because anything stored in block 1237 (for example) is known to have bits <12:1> of its address set to 1237.

If desired, cache entries can also include parity information as shown in Figure 5-5.



Figure 5-5 Cache Entry With Parity

The Parity 0 Bit stores parity information for byte 0, the Parity 1 Bit stores parity information for byte 1, and the Parity 2 Bit stores parity information for the cache tag/valid bit combination.

The Cache Control Register for this example is configured as shown





BIT

<15:11> Not Used (read as zeroes)

NAME

10 Write Wrong Tag Parity (read/write)

- 9 Bypass Cache (read/write)
- 8 Flush Cache (read as zero)
- 7 Not Used (read/write)

6 Write Wrong Data Parity (read/write)

FUNCTION

These bits are not used in this example. The DCJ11 will ignore any data written to these bits and will always read these bits as zeroes.

This bit, when set, causes the cache tag parity bit (Parity 2) to be written with wrong parity when a cache entry is updated (i.e. upon CPU read misses and write hits). This causes a cache tag parity error on the next access to a location referenced by the entry.

This bit, when set, forces all CPU memory references to go directly to main memory. Read or write hits will result in invalidation of accessed locations in the cache.

Setting this bit causes the entire contents of the cache to be declared invalid. Writing a "0" into this bit will have no effect.

This bit is not used in this example.

This bit, when set, causes the parity bits of the two data bytes (Parity 0 and Parity 1) to be written with wrong parity when

updated (i.e. upon CPU read misses and write hits). This causes a cache parity error to occur on the next access to a location referenced by the entry.

- <5:4> Not Used (read/write)
- <3:2> Force Miss (read/write)

These bits are not used in this example.

These bits, when either is set, force all DCJ11 memory references to go directly to main memory. Unlike cache bypasses, force misses have no effect on cache entries. Enabling force miss effectively removes cache memory from the system.

These bits are not used in this

- 1 Not Used (read/write)
- 0 Disable Cache Traps
 (read/write)

This bit, when set, disables cache parity interrupts. When this bit is cleared, an interrupt occurs when a parity error is encountered.

All words read from the cache are checked for parity. A parity error in the accessed word causes the following CPU responses:

- CCR<0> Action
 - 0 Interrupt through vector 114 and force miss.

example.

1 Force miss only.

The CCR is cleared on power-up or by a console start. It is unaffected by a RESET instruction.

The cache response matrix for this example would be:

	C	PU		DMA			
	Hit	Miss		Hit	Miss		
Read	Read cached data	Read cached Read memory data & allocate cache		Read memory	Read memory		
Write	Write thru cache to memory	Write memory		Invalidate cache & writé mem	Write memory		
Read bypass	Invalidate cache & read mem	Read memory		na	na		
Write bypass	Invalidate cache & write mem	Write memory		na	na		
Read forced miss	Read memory	Read memory		na	na		
Write forced miss	Write memory	Write memory		na	na		

na = not applicable

5.3 CONSOLE ODT

The console octal debugging technique or console ODT allows the DCJ11 to respond to commands and information entered via a user-designed console terminal interface. The interface bus uses addresses 17777560 through 17777566 to communicate with console ODT. These addresses are generated in the DCJ11 and cannot be changed. Console ODT is a very useful aid in running and debugging programs. Communication between the user and DCJ11 is via a stream of ASCII characters which are interpreted by the DCJ11 as console commands. These commands are a subset of the commands used in DIGITAL's ODT-11 software for minicomputers.

5.3.1 Terminal Interface - The minimum optional hardware requirements for an interface permitting communication with console ODT are outlined in the paragraphs that follow (these requirements are met by the DLART DL-compatible asynchronous receiver/transceiver peripheral chip - DIGITAL Part No. DC319-AA).

5.3.1.1 Receiver Control/Status Register (RCSR) - The RCSR (Figure 5-7) must exist at address 1777560 for character input to console ODT. Console ODT does not execute output bus cycles to this address; therefore the RCSR only needs to respond to input bus cycles. System software may affect certain bits, such as Interrupt Enable (bit 6), but console ODT ignores this.



Figure 5-7 Receiver Control/Status Register (RCSR) - Address 17777560

Bit Description

- <15:8> Unused. These bits may be in any state since console ODT does not use them.
- <7> Done flag. After a character is received and exists in the receiver buffer register (RBUF), the Done flag must be set to a 1. When the character is read from RBUF Done flag must be cleared by hardware.
- <6:0> Unused. These bits may be in any state since console ODT does not use them.

5-9

5.3.1.2 Receiver Buffer Register (RBUF) - The RBUF (Figure 5-8) must exist at address 1777562 for character input to console ODT. This register only needs to respond to input bus cycles since console ODT does not execute output bus cycles to this address. System software operates similarly, but DIGITAL diagnostics may cause output cycles and thus may not operate properly.



- Figure 5-8 Receiver Buffer Register (RBUF) Address 17777562 Bit Description
- <15:8> Unused. These bits can be in any state since console ODT does not use them.
- <7:0> ASCII character. These eight bits are read by the processor and interpreted as a console ODT command. When bit 7 of RCSR is a 1, the processor reads data from the RBUF. After the input cycle, the hardware must clear bit 7 of RCSR to 0.

5.3.1.3 Transmitter Control And Status Register (XCSR) - The XCSR (Figure 5-9) must exist at address 17777564 for character output from console ODT. ODT does not execute output bus cycles to this address; therefore, the XCSR only needs to respond to input bus cycles. System software may cause output cycles to affect certain bits, such as Interrupt Enable, but console ODT ignores this.



Figure 5-9 Transmitter Control/Status Register(XCSR) - Address 1777564

Bit Description

- <15:8> Unused. These bits may be in any state since console ODT does not use them.
 - <7> Done flag. In the idle state, this bit is a l indicating that the XBUF is ready to receive a character. After an output cycle to the transmitter buffer register (XBUF) by the processor, this bit must be cleared to 0 by the hardware. When the XBUF is ready to receive another character, the hardware sets this bit to 1.
- <6:0> Unused. These bits may be in any state since console ODT does not use them. Note that these bits may be meaningful to other DIGITAL interfaces.

5.3.1.4 Transmitter Buffer Register (XBUF) - The XBUF (Figure 5-10) must exist at address 1777566 for character output from console ODT. This register only needs to respond to output bus cycles since console ODT does not execute input bus cycles to this address. System software operates similarly but DIGITAL diagnostics may cause an input cycle and thus may not operate properly.



Figure 5-10 Transmitter Buffer Register (XBUF) - Address 17777566 Bit Description

<15:8> Unused. These bits may be in any state since console ODT does not use them.

<7:0> ASCII character. These eight bits are written by the processor with the ASCII character output by ODT. When bit 7 of XCSR is a 1, the processor may perform an output cycle to XBUF.

5.3.2 Console ODT Operation - Console ODT operates the console terminal interface in half-duplex mode. Communication between console ODT and the interface is accomplished via programmed I/O techniques rather than interrupts. When console ODT is outputting characters using the transmit side of the interface, the receive side of the interface is not monitored for incoming characters. Any characters coming in at this time are lost. Console ODT does not check for error bits in the interface. If another processor is at the other end of the interface, that processor must operate within format of half-duplex transmission. No input the characters should be sent until console ODT has finished outputting.

5.3.2.1 Console ODT Initialization - Console ODT operation is initiated by any of the following:

- 1. Execution of a HALT instruction in kernel mode (if kernel HALT is enabled).
- 2. Assertion of the HALT signal on the system bus. The signal must be asserted long enough so that it is seen by the processor at the end of the current macroinstruction
- 3. At power-up, if the appropriate power-up option is selected.

Console ODT Input Sequence

The Console ODT entry sequence is as follows:

- Output <CR><LF> to XBUF. 1.
- Output the contents of PC in six digits to XBUF. 2.
- 3. Read and ignore character in RBUF. (May be a program character.)
- 4. Output <CR><LF> to XBUF.
- 5.
- Output the prompt character, 0, to XBUF. Enter a wait loop for input. The Done flag, bit 7 in RCSR, is 6. tested. If it is 0, the test continues.
- 7. If RCSR bit 7 is a 1, then the low byte of RBUF is read.

5.3.2.2 Console ODT Output Sequence -

Console ODT does the following when it has a character ready for output:

- 1. Test XCSR bit 7 (Done flag) and if a 0, continue testing.
- 2. If XCSR bit 7 is a 1, write character to low byte of XBUF (high byte should be ignored by interface).

5.3.3 Console ODT Command Set - The console ODT command set is a subset of ODT-11 and uses the same command characters. Only specific characters are recognized as valid inputs; other inputs invoke a "?" response. The commands are summarized in Table 5-2.

The word "location," as used in the paragraphs that follow refers to a memory location, an I/O device register, an internal processor register, or the processor status word (PS).

Table	5-2	Console	ODT	Commands
-------	-----	---------	-----	----------

Command	Symbol	Function				
Slash	n/	Opens the specified location (n) and outputs its contents. n is an octal number.				
Carriage Return	<cr></cr>	Closes an open location.				
Line Feed	<lf></lf>	Closes an open location and then opens the next contiguous location.				
Internal Register Designator	\$n or Rn	Opens a specific processor register (n). n is an integer from 0 to 7 or the character S.				
Processor Status Word Designator	S	Opens the PS - must follow an \$ or R command.				
Go	G	Starts program execution.				
Proceed	Р	Resumes execution of a program.				

Binary Dump

Control-Shift-S

Manufacturing use only.

The parity bit (bit 7) on all input characters is ignored (i.e., not stripped) by console ODT. If an input character is echoed, the state of the parity bit is copied to the output buffer (XBUF). Output characters internally generated (e.g., <CR>) by ODT have the parity bit equal to 0. All commands are echoed except for ASCII codes in the range 0-17 (octal). Where applicable, the upper- and lowercases of command characters are recognized.

NOTE

In the examples that follow, the response from the processor is underlined, while the user's entry is not. When the user inputs an address or data, leading zeroes are not required. The DCJ11, however, outputs 8 digit octal addresses and 6 digit octal data words.

5.3.3.1 / (ASCII 057) Slash - This command is used to open a memory location, I/O device register, internal processor register, or processor status word and must be preceded by other characters which specify a location. In response to /, console ODT prints the contents of the location (i.e., six characters) and then a space (ASCII 40). After printing is complete, console ODT waits for either new data for that location or a valid close command.

Example:

@001000/<u>012525<SPACE></u>

where:

6	=	console ODT prompt character.
001000	=	octal location desired by the user (leading 0s are not required).
1	=	command to open and print contents of location.
012525	2	contents of octal location 1000.
SPACE>	z	space character generated by console ODT.

5.3.3.2 <CR> (ASCII 015) Carriage Return - This command is used to close an open location. If a location's contents are to be changed, the user should precede the <CR> with the new data. If no change is desired, <CR> closes the location without altering its contents.

Example: <u>@R1/004321<SPACE></u> <CR> <u><CR> <LF></u> @

Processor register Rl was opened and no change was desired so the user issued<CR>. In response to the <CR>, console ODT printed <CR><LF>@.

Example: <u>@R1/004321<SPACE></u> 1234 <CR> <u><CR> <LF></u> @

In this case the user desired to change R1, so new data, 1234, was entered before issuing the <CR>. Console ODT deposited the new data in the open location and then printed <CR><LF>@.

Console ODT does not directly echo the <CR> entered by the user but instead prints a <CR>, followed by an <LF>, and @.

5.3.3.3 <LF> (ASCII 012) Line Feed - This command is used to close an open location and then open the next contiguous location. Memory locations and processor registers are incremented by 2 and 1 respectively. If the PS is open when a <LF> is issued, it is closed and a <CR><LF>@ is printed; no new location is opened. If the open location's contents are to be changed, the new data should precede the <LF>. If no data is entered, the location is closed without being altered.

Example: <u>@R2/123456<SPACE></u> <LF> <u><CR><LF></u> R3/054321<SPACE>

In this case, the user entered <LF> with no data preceding it. In

response, console ODT closed R2 and then opened R3. When a user has the last register, R7, open, and issues <LF>, console ODT opens the beginning register, R0.

Example: <u>@R7/000000<SPACE></u> <LF> <u><CR><LF></u> R0/<u>123456<SPACE></u>

Unlike with most other commands, console ODT does not echo the <LF>. Instead it prints <CR>, then <LF>, so that terminal printers operate properly. In order to make this easier to decode, console ODT does not echo ASCII characters in the range 0 - 17 (octal).

5.3.3.4 \$ (ASCII 044) Or R (ASCII 122) Internal Register Designator - Either character when followed by a register number, 0 to 7, or PS designator, S, will open that specific processor register.

The \$ character is recognized to be compatible with ODT-11. The R character was introduced because it can be conveniently typed with one key stroke and because it is an easily remembered symbol for a register.

Example: @\$0/000123<SPACE>

or

@R7/000123<SPACE> <LF> R0/054321<SPACE>

If more than one character is typed after the R or \$, console ODT uses the last character typed as the register designator.

5.3.3.5 S (ASCII 123) Processor Status Word - This designator is for opening the PS (processor status word) and may be employed only after the user has entered an R or \$ register designator.

Example: <u>@RS/100377<SPACE></u> 0 <CR> <u><CR><LF></u>

NOTE

The trace bit (bit <4>) of the PS cannot be modified by the user. This is done so that PDP-11 program debugging utilities (e.g., ODT-11), which use the T bit for single-stepping, are not accidentally harmed by the user.

If the user issues a <LF> while the PS is open, the PS is closed and ODT prints <CR><LF>@. No new location is opened in this case. **5.3.3.6** G (ASCII 107) Go - This command is used to start program execution at a location entered immediately before the G. This function is equivalent to the LOAD ADDRESS and START switch sequence on other PDP-11 consoles.

Example: <u>@200G<NULL><NULL></u>

The console ODT sequence for a G, after echoing the command character, is as follows.

- 1. Print two nulls (ASCII 0). This is intended to prevent the G character from getting flushed during the bus initialization sequence that follows, assuming a double-buffered UART chip is used in the console terminal interface.
- 2. Load R7 (PC) with the entered data. If no data is entered, 0 is used. (In the above example, R7 is set to 200, and that is where program execution begins.)
- 3. The PS, MMR0<15:13,0>, MMR3, PIRQ, CPU Error Register, Memory System Error Register, Cache Control Register, and Floating Point Status Register are cleared to zero.
- 4. The cache, if present, is flushed (if so implemented).
- 5. The system bus is initialized by the processor.
- 6. The service state is entered by the DCJ11. Any outstanding service requests are processed. If the bus HALT signal is asserted, the processor reenters the console ODT state. This feature is used to initialize a system without starting a program (R7 is altered).

5.3.3.7 P (ASCII 120) Proceed - This command is used to resume execution of a program and corresponds to the CONTINUE switch on other PDP-11 consoles. No programmer-visible machine state is altered using this command.

Example: <u>@</u>P

Program execution resumes at the address pointed to by R7. After the P is echoed, the DCJ11 immediately fetches the next instruction. After the instruction is executed, outstanding interrupts, if any, are serviced. If the HALT bus signal is asserted, it is recognized at the end of the instruction, and the DCJ11 enters the console ODT state. Upon entry, the content of the PC (R7) is printed. In this fashion, the user can single-instruction step through a program and obtain a PC "trace" on the terminal. **5.3.3.8 Control-Shift-S (ASCII 023) Binary Dump** - This command is used for manufacturing test purposes and is not a normal user command. It is described here to explain the processor's response if accidentally invoked. It is intended to more efficiently display a portion of memory compared to using the "/" and <LF> commands. The protocol is as follows.

- 1. After a prompt character, console ODT receives a control-shift-S command and echoes it.
- 2. The host system at the other end of the serial line must send two 8 bit bytes which console ODT interprets as a starting address. These two bytes are not echoed.

The first byte specifies starting address <15:08> and the second byte specifies starting address <07:00>. Address bits <21:16> are always forced to be 0; the dump command is restricted to the first 32K words of address space.

3. After the second address byte has been received, console ODT outputs ten bytes to the serial line starting at the address previously specified. When the output is finished, console ODT prints <CR><LF>0.

If a user accidentally enters this command, it is recommended in order to exit from the command that two @ characters (ASCII 100) be entered as a starting address. After the binary dump, an @ prompt character is printed.

5.3.4 Address Specification - All I/O addresses (17760000 to 1777777) must be entered by the user with all 22 bits specified. For example, if a user desires to open the RCSR of the console serial interface he must enter 17777560, not 177560, or 777560.

5.3.4.1 General Registers - Whenever R0-R5 are referenced in console ODT, they access the general register set currently specified by PS bit 11 (PS<11>). If a program operating in general register set zero (PS<11> = 0) is halted and a general register is opened, register set zero is accessed. Similarily, if a program is operating in register set one, references to R0-R5 access register set one.

If a specific register set is desired, PS<11> must be set by the user to the appropriate value, and then the R0 through R5 commands can be used. If an operating program has been halted, the original value of PS<11> must be restored in order to continue execution.

Example: PS = 000000

<u>@R4/052525<SPACE></u> <CR> <u><CR><LF></u>

R4 in register set zero has been opened.

@RS/00000<SPACE> 4000 <CR> <CR><LF> @R4/17777<SPACE> <CR> <CR><LF> @RS/004000<SPACE> 0 <CR> <CR><LF> @P

In this case, R4 in register set one was desired. The PS was opened, and PS<11> was set to 1 (register set one). Then R4 was examined and closed. The original value of PS<11> was restored, and the program was continued using the P command.

5.3.4.2 Stack Pointers - Whenever R6 is referenced in console ODT, it accesses the stack pointer specified by the PS current mode bits (PS<15:14>). If a program operating in kernel mode (PS<15:14> = 00) is halted and R6 is opened, the kernel stack pointer is accessed. Similarly, if a program is operating in supervisor or user mode, R6 accesses the supervisor or user stack pointers.

If a specific stack pointer is desired, PS<15:14> must be set by the user to the appropriate value and then the R6 command can be used. If an operating program has been halted, the original value of PS<15:14> must be restored in order to continue execution.

Example: PS = 140000

@R6/123456<SPACE> <CR> <<u>CR><LF></u>

The user mode stack pointer has been opened.

@RS/140000<SPACE> 0 <CR> <CR><LF> @R6/123456<SPACE> <CR> <CR><LF> @RS/000000<SPACE> 140000<CR> <CR><LF> @P

In this case, the kernel mode stack pointer was desired. The PS was opened, and PS<15:14> were set to 00 (kernel mode). Then R6 was examined and closed. The original value of PS<15:14> was restored, and then the program was continued using the P command.

5.3.4.3 Floating Point Accumulators - The floating point accumulators cannot be accessed from console ODT. Only floating point instructions can access these registers.

5.3.5 Entering Octal Digits - When the user is specifying an address, console ODT will use the last eight octal digits if more than eight have been entered. When the user is specifying data, console ODT will use the last six octal digits if more than six have been entered. The user need not enter leading 0s for either address or data; console ODT forces 0s as the default. If an odd address is entered, console ODT responds to the error by printing ?<CR><LF>@.

5.3.6 ODT Timeout - If the user specifies a nonexistent address or causes a parity error, console ODT responds to the error by printing ?<CR><LF>@.

5.3.7 Invalid Characters - Console ODT will recognize upper- or lowercase characters as commands. Any character that console ODT does not recognize during a particular sequence is echoed (except for ASCII characters in the range 0 - 17 (octal)), and console ODT prints ?<CR><LF>0.

5.4 DCJ11 PIPELINE PROCESSING

The DCJ11 gets much of its performance from its prefetch and predecode mechanisms. The primary benefit of prefetch and predecode is that memory references are overlapped with internal operations, and the need for explicit instruction fetch and decode cycles is minimized. The prefetch and predecode operations are performed automatically by the DCJ11 chip and cannot be altered by the user.

A primary function of the prefetch mechanism is to fill four registers with information and replenish the registers as required. These four registers, the virtual program counter (VPC), the physical program counter (PPC), the prefetch buffer (PB), and the instruction register (IR) are collectively referred to as the prefetch pipeline. The contents of registers in the beginning of the pipeline are used to determine the contents of registers further down the pipeline. When the pipeline is filled, the prefetch mechanism is said to be in steady state. Four microcycles are required to fill an empty pipeline. Figure 5-11 illustrates the process of filling the pipeline.

Microcycle 1	Microcycle 2	Microcycle 3	Microcycle 4			
VPC < PC	PPC < MMU(VPC) VPC < VPC + 2	PB < M[PPC] PPC < MMU(VPC) VPC < VPC + 2	IR < PB PB < M[PPC] PPC < MMU(VPC) VPC < VPC + 2 PC < PC + 2			
			MMRZ < PC			

Figure 5-11 Pipeline Filling Process

In microcycle 1, the VPC is is simply set to the same value as the PC. In microcycle 2, the VPC is sent through the MMU and the resulting physical address is loaded into the PPC. The VPC is then incremented by 2. At this point we have a valid VPC and PPC and the pipeline is said to be synchronized. Sometimes while executing a macroinstruction, the pipeline is synchronized but not filled. In that case, only microcycles 3 and 4 need be performed for the next macroinstruction.

In microcycle 3, the word in memory addressed by the PPC is fetched into the PB. The PPC is updated with the relocated (mapped) VPC and the VPC is incremented again. In microcycle 4, the PB is sent to the IR and is decoded as the next macroinstruction (note that the DCJ11 asserts PDRC at this time). The new contents of the PB are fetched from the memory location referenced by the PPC. The PPC is again updated with the relocated (mapped) VPC and the VPC is updated (incremented) once again. Also during microcycle 4, the original PC is loaded into MMR2 (if MMR0<15:13> = 000) and is incremented by 2.

In steady state (i.e., when microcycle 4 is complete), the IR contains the macroinstruction being executed, the PB contains the data from the memory location pointed to by the PC, the PPC contains the physical address of the next word to be prefetched,

and the VPC contains the incremented value of the PC.

Once in steady state, a stream of macroinstructions that operate only on registers may be executed at the rate of one per microcycle (i.e., microcycle 4). While one instruction is being executed, the next one is being decoded, and the following one is being prefetched into the PB. As illustrated in Figure 5-11 during microcycle 4: the contents of the prefetch buffer are loaded into the IR, the word addressed by the PPC is loaded into the PB, the VPC is relocated and loaded into the PPC, and the VPC is incremented by 2. This maintains the steady state, allowing the next macroinstruction to be executed in the next microcycle. Note also that the DCJ11 bus is kept busy 100% of the time.

The instructions that operate on immediate data and a register also make maximum use of the prefetch mechanism. At steady state, a stream of these macroinstructions execute in two microcycles (microcycles 3 and 4). During microcycle 3, the data in the PB is moved to a scratch register. During microcycle 4, the operation is performed. In both cycles, the steady state of the prefetch mechanism is maintained by prefetching the next instruction stream word. The DCJ11 bus is again kept busy 100% of the time.

The prefetch pipeline is refilled after a power-up sequence or if a prefetch fault occurs. Prefetch faults occur when the PS, CCR, PC, or any of the memory management registers are written. A prefetch fault invalidates only the PB. This means that the pipeline remains synchronized and can be refilled in two microcycles.

5.4.1 Pipeline Flow Example - Consider the following example program:

Virtual Address	Symbolic Representation	Octal Code		
1000	MOV R2,R3	010203		
1002	BIS #1,R3	052703		
		000001		
1004	ADD R1,R3	060105		
1006	CLR R0 、	005000		
1012	ADD R3,R0	060300		

The flow of information through the pipeline occurs as shown in Table 5-3.

× -						
Pipel: Regis	ine ter		Microcycle			
	n	<u>n+1</u>	n+2	n+3	n+4	n+5
PC	1002	1004	1006	1010	1012	1014
IR	MOV (010203)	BIS (052703)	BIS (052703)	ADD (060105)	CLR (005000)	ADD (060300)
PB	BIS (052703)	000001	ADD (060105)	CLR (005000)	ADD (060300)	*
PPC	MMU(1004)	MMU(1006)	MMU(1010)	MMU(1012)	MMU(1014)	MMU(1016)
VPC	1006	1010	1012	1014	1016	1020

Table 5-3 Pipeline Flow

* Instruction at location 1014

Note that the example starts at microcycle n, by which time the prefetch pipeline has been filled (i.e., the pipeline is in steady state). All the instructions in the example execute in one microcycle except the BIS instruction, which executes in two microcycles.

6.1 INTRODUCTION

The first part of this chapter is divided into six major sections:

- Single-Operand Addressing -- One part of the instruction word specifies the registers; the other part provides information for locating the operand.
- Double-Operand Addressing -- One part of the instruction word specifies the registers; the remaining parts provide information for locating two operands.
- O Direct Addressing -- The operand is the content of the selected register.
- Deferred (Indirect) Addressing -- The contents of the selected register is the address of the operand.
- Use of the PC as a General-Purpose Register -- The PC is different from other general-purpose registers in one important respect. Whenever the processor retrieves an instruction, it automatically advances the PC by 2. By combining this automatic advancement of the PC with four of the basic addressing modes, we produce the four special PC modes -- immediate, absolute, relative, and relative-deferred.
- Use of the Stack Pointer as a General-Purpose Register --General-purpose registers can be used for stack operations.

The second part of this chapter describes each of the instructions in the DCJ11 instruction set.

6.2 ADDRESSING MODES

Data stored in memory must be accessed and manipulated. Data handling is specified by a DCJll instruction (MOV, ADD, etc.), which usually specifies the:

- o Function to be performed (operation code).
- General-purpose register to be used when locating the source operand, and/or destination operand (where required).
- Addressing mode, which specifies how the selected registers are to be used.

A large portion of the data handled by a computer is structured

6 - 1

(in character strings, arrays, lists, etc.). The DCJ11 addressing modes provide for efficient and flexible handling of structured data.

A general-purpose register may be used with an instruction in any of the following ways.

- 1. As an accumulator -- The data to be manipulated resides in the register.
- 2. As a pointer -- The contents of the register is the address of an operand, rather than the operand itself.
- 3. As a pointer that automatically steps through memory locations -- Automatically stepping forward through consecutive locations is known as autoincrement addressing; automatically stepping backwards is known as autodecrement addressing. These modes are particularly useful for processing tabular or array data.
- 4. As an index register -- In this instance, the contents of the register and the word following the instruction are summed to produce the address of the operand. This allows easy access to variable entries in a list.

An important DCJ11 feature, which should be considered with the addressing modes, is the register arrangement.

- o Two sets of six general-purpose registers (R0--R5 and R0[^]--R5[^])
- A hardware stack pointer (SP) register (R6) for each processor mode (kernel, supervisor, user)
- o A program counter (PC) register (R7)

Registers R0--R5 and R0'--R5' are not dedicated to any specific function; their use is determined by the instruction that is decoded.

- o They can be used for operand storage. For example, the contents of two registers can be added and stored in another register.
- o They can contain the address of an operand or serve as pointers to the address of an operand.
- o They can be used for the autoincrement or autodecrement features.
- o They can be used as index registers for convenient data and program access.

The DCJ11 also has instruction addressing mode combinations that facilitate temporary data storage structures. These can be used for convenient handling of data that must be accessed frequently. This is known as stack manipulation. The register that keeps track of stack manipulation is known as the stack pointer (SP). Any register can be used as a stack pointer under program control; however, certain instructions associated with subroutine linkage and interrupt service automatically use register R6 as a "hardware stack pointer." For this reason, R6 is frequently referred to as the SP.

- o The stack pointer (SP) keeps track of the latest entry on the stack.
- o The stack pointer moves down as items are added to the stack and moves up as items are removed. Therefore, the stack pointer always points to the top of the stack.
- o The hardware stack is used during trap or interrupt handling to store information, allowing an orderly return to the interrupted program.

Register R7 is used by the processor as its program counter (PC). It is recommended that R7 not be used as a stack pointer or accumulator. Whenever an instruction is fetched from memory, the program counter is automatically incremented by two to point to the next instruction word.

6.2.1 Single-Operand Addressing - The instruction format for all single-operand instructions (such as CLR, INC, TST) is shown in Figure 6-1.



Figure 6-1 Single-Operand Addressing

Bits <15:6> specify the operation code that defines the type of instruction to be executed.

MR 5458

Bits <5:0> form a 6-bit field called the destination address field. The destination address field consists of two subfields:

- o Bits <5:3> specify the destination mode. Bit 3 is set to indicate deferred (indirect) addressing.
- o Bits <2:0> specify which of the 8 general-purpose registers is to be referenced by this instruction word.

6.2.2 Double-Operand Addressing - Operations that imply two operands (such as ADD, SUB, MOV, and CMP) are handled by instructions that specify two addresses. The first operand is called the source operand; the second is called the destination operand. Bit assignments in the source and destination address fields may specify different modes and different registers. The instruction format for the double operand instruction is shown in



Figure 6-2 Double-Operand Addressing

The source address field is used to select the source operand (the first operand). The destination is used similarly, and locates the second operand and the result. For example, the instruction ADD A, B adds the contents (source operand) of location A to the contents (destination operand) of location B. After execution, B will contain the result of the addition and the contents of A will be unchanged.

Examples in this paragraph and the rest of the chapter use the following sample DCJ11 instructions. (A complete listing of the DCJ11 instructions appears in Paragraph 6.3.)

Mnemonic	Description	Octal Code
CLR	Clear. (Zero the specified destination.)	0050DD
CLRB	Clear byte. (Zero the byte in the specifie destination.)	đ 1050DD
INC	Increment. (Add one to contents of the destination.)	0052DD
INCB	Increment byte. (Add one to the contents o the destination byte.)	f 1052DD
СОМ	Complement. (Replace the contents of the destination by its logical complement; each 0 bit is set and each one bit is cleared.)	0051DD
СОМВ	Complement byte. (Replace the contents of the destination byte by its logical complement; each 0 bit is set and each 1 bit is cleared.)	1051DD
ADD	Add. (Add the source operand to the destination operand and store the result at the destination address.)	06SSDD
DD = desti SS = sourc () = conte	ination field (six bits) ce field (six bits) ents of	

6.2.3 Direct Addressing - The following summarizes the four basic modes used with direct addressing. Direct Modes (Figures 6-3 to 6-6) Assembler Mode Name Syntax Function 0 Register Register contains operand. Rn OPERAND INSTRUCTION MB 6460 Figure 6-3 Mode 0 Register Assembler Mode Name Syntax Function 2 Autoincrement (Rn)+ Register is used as a pointer to sequential data and then incremented. INSTRUCTION OPERAND ADDRESS +2 FOR WORD, +1 FOR BYTE MB-5461 Figure 6-4 Mode 2 Autoincrement Assembler Mode Name Syntax Function Autodecrement - (Rn) 4 Register is decremented and then used as a pointer. INSTRUCTION -2 FOR WORD ADDRESS OPERAND -1 FOR BYTE MR-5462 Figure 6-5 Mode 4 Autodecrement Assembler Mode Name Syntax Function 6 Index Value X is added to (Rn) to X(Rn) produce address of operand. Neither X nor (Rn) is modified. INSTRUCTION ADDRESS OPERAND х MR-5463

Figure 6-6 Mode 6 Index

6-5

6.2.3.1 Register Mode - With register mode any of the general registers may be used as simple accumulators, with the operand contained in the selected register. Since they are hardware registers (within the processor), the general registers operate at high speeds and provide speed advantages when used for operating on frequently accessed variables. The assembler interprets and assembles instructions of the form OPR Rn as register mode operations. Rn represents a general register name or number and OPR is used to represent a general instruction mnemonic. Assembler syntax requires that a general register be defined as follows.

R0 = %0 (% sign indicates register definition)
R1 = %1
R2 = %2, etc.

Registers are typically referred to by name as R0, R1, R2, R3, R4, R5, R6, and R7. However, R6 and R7 are also referred to as SP and PC, respectively.

OPR Rn

Register Mode Examples (Figures 6-7 to 6-9)

1. Symbolic Octal Code Instruction Name

INC R3

005203 Increment

Operation: Add one to the contents of general-purpose register R3.



2. Symbolic				Octal Code			Instruction Name				
А	DD R2	, R4			0602	04		A	đđ		
Operat	ion:	Add	the	content	s of	R2	to	the	contents	of	R4.

6-6



Figure 6-8 ADD R2,R4 Add

3. Symbolic

Octal Code Instruction Name

Instruction Name

Instruction Name

Clear byte

COMB R4 105104 Complement byte Operation: 1's complement bits <7:0> (byte) in R4. (When general registers are used, byte instructions operate only on bits <7:0>; i.e., byte 0 of the register.)



Figure 6-9 COMB R4 Complement Byte

6.2.3.2 Autoincrement Mode [OPR (Rn)+] - This mode (mode 2) provides for automatic stepping of a pointer through sequential elements of a table of operands. It assumes the contents of the selected general-purpose register to be the address of operand. Contents of registers are stepped (by one for the byte instructions, by two for word instructions, always by two for R6 location. and R7) to address the next seguential The autoincrement mode is especially useful for array processing and stack processing. It will access an element of a table and then step the pointer to address the next operand in the table. Although most useful for table handling, this mode is completely general and may be used for a variety of purposes.

OPR(Rn) +

Autoincrement Mode Examples (Figures 6-10 to 6-12)

1. Symbolic

CLR (R5)+ 005025 Clear

Operation: Use contents of R5 as the address of the operand. Clear selected operand and then increment the contents of R5 by two.

Octal Code



105025

2. Symbolic

CLRB (R5)+

6-7

Octal Code

Operation: Use contents of R5 as the address of the operand. Clear selected byte operand and then increment the contents of R5 by one.



Figure 6-11 CLRB (R5)+ Clear Byte

3. Symbolic Octal Code Instruction Name

ADD (R2)+,R4

062204 Add

Operation: The contents of R2 are used as the address of the operand, which is added to the contents of R4. R2 is then incremented by two.



Figure 6-12 ADD (R2)+,R4 Add

6.2.3.3 Autodecrement Mode [OPR-(Rn)] - This mode (mode 4) is useful for processing data in a list in reverse direction. The contents of the selected general-purpose register are decremented (by one for byte instructions, by two for word instructions) and then used as the address of the operand. The choice of postincrement, predecrement features for the DCJ11 were not decisions, intended to facilitate arbitrary but were hardware/software stack operations.

OPR-(Rn)

Autodecrement Mode Examples (Figures 6-13 to 6-15)

1.	Symbolic	Octal Code	Instruction Name
	INC - (R0)	005240	Increment

Operation: The contents of R0 are decremented by two and used as the address of the operand. The operand is incremented by one.



Figure 6-13 INC - (R0) Increment

2. Symbolic Octal Code Instruction Name

INCB - (R0) 105240 Increment byte

Operation: The contents of R0 are decremented by one and then used as the address of the operand. The operand byte is increased by one. BEFORE AFTER



ADD -(R3), RO

3.

064300 Add

Operation: The contents of R3 are decremented by two and then used as a pointer to an operand (source), which is added to the contents of R0 (destination operand).



6.2.3.4 Index Mode [OPR X(Rn)] - In this mode (mode 6) the contents of the selected general-purpose register, and an index word following the instruction word, are summed to form the address of the operand. The contents of the selected register may be used as a base for calculating a series of addresses, thus allowing random access to elements of data structures. The selected register can then be modified by program to access data in the table. Index addressing instructions are of the form OPR
X(Rn), where X is the indexed word located in the memory location following the instruction word and Rn is the selected general-purpose register.

OPR X(Rn)

Index Mode Examples (Figures 6-16 to 6-18)

1.	Symbolic	Octal Code	Instruction Name
	CLR 200(R4)	005064 000200	Clear

Operation: The address of the operand is determined by adding 200 to the contents of R4. The operand location is then cleared.



Figure 6-16 CLR 200(R4) Clear

2. Symbolic

COMB 200 (R1)

105161 000200

Octal Code

Complement byte

Instruction Name

Operation: The contents of a location, which are determined by adding 200 to the contents of Rl, are 1's complemented (i.e., logically complemented).



Figure 6-17 COMB 200(R1) Complement Byte

3. Symbolic Octal Code Instruction Name ADD 30(R2),20(R5) 066265 Add 000030 000020

Operation: The contents of a location, which are determined by

adding 30 to the contents of R2, are added to the contents of a location that is determined by adding 20 to the contents of R5. The result is stored at the destination address, that is, 20(R5).



Figure 6-18 ADD 30(R2),20(R5) Add

6.2.4 Deferred (Indirect) Addressing - The four basic modes may also be used with deferred addressing. Whereas in register mode the operand is the contents of the selected register, in register-deferred mode the contents of the selected register is the address of the operand.

In the three other deferred modes, the contents of the register select the address of the operand rather than the operand itself. These modes are therefore used when a table consists of addresses rather than operands. The assembler syntax for indicating deferred addressing is @ [or () when this is not ambiguous]. The following summarizes the deferred versions of the basic modes.

Deferred Modes (Figures 6-19 to 6-22)

		Assembler	
Mođe	Name	Syntax	Function

1 Registerdeferred



Figure 6-19 Mode 1 Register-Deferred

Mode	Name	Assembler Syntax	Function		
3	Autoincrement Deferred	- @(Rn)+	Register	is	f

Register is first used as a



The following examples illustrate the deferred modes.

Register-Deferred Mode Example (Figure 6-23)

Symbolic	Octal Code	Instruction Name
CLR @R5	005015	Clear

Operation: The contents of location specified in R5 are cleared.



Figure 6-23 CLR @R5 Clear

Autoincrement-Deferred Mode Example (Mode 3) (Figure 6-24)

Symbolic Octal Code Instruction Name

INC @(R2)+ 005232 Increment

Operation: The contents of R2 are used as the address of the address of the operand. The operand is increased by one; the contents of R2 are incremented by two.



Figure 6-24 INC @(R2)+ Increment

Autodecrement-Deferred Mode Example (Mode 5) (Figure 6-25)

Symbolic Octal Code

COM @-(R0) 005150

i

Operation: The contents of R0 are decremented by two and then used as the address of the address of the operand. The operand is 1's complemented (i.e., logically complemented).



Figure 6-25 COM @-(R0) Complement

Index-Deferred Mode Example (Mode 7) (Figure 6-26)

Symbolic	Octal Code	Instruction Name
ADD @1000(R2),R1	067201 001000	Add

Operation: 1000 and the contents of R2 are summed to produce the address of the address of the source operand, the contents of which are added to the contents of R1; the result is stored in R1.



Figure 6-26 ADD @1000(R2),R1 Add

6.2.5 Use Of The PC As A General-Purpose Register - Although register 7 is a general-purpose register, it doubles in function as the program counter for the DCJ11. Whenever the processor uses the program counter to acquire a word from memory, the program counter is automatically incremented by two to contain the address of the next word of the instruction being executed or the address of the next instruction to be executed. (When the program uses the PC to locate byte data, the PC is still incremented by two.)

The PC responds to all the standard DCJ11 addressing modes. However, with four of these modes the PC can provide advantages for handling position-independent code and unstructured data. When utilizing the PC, these modes are termed immediate, absolute (or immediate-deferred), relative, and relative-deferred. The modes are summarized below.

Mode	Name	Assembler Syntax	Function
2	Immediate	# n	Operand follows instruction.
3	Absolute	6‡A	Absolute address of operand follows instruction.

Relative address (index value) follows the instruction.

7 Relativedeferred 6A

Index value (stored in the word after the instruction) is the relative address for the address of the operand.

When a standard program is available for different users, it is often helpful to be able to load it into different areas of memory and run it in those The DCJ11 can accomplish the areas. relocation of a program very efficiently through the use of position-independent code (PIC), which is written by using the PC addressing modes. If an instruction and its operands are moved in such a way that the relative distance between them is not altered, the same offset relative to the PC can be used in all positions in memory. Thus, PIC usually references locations relative to the current location.

The PC also greatly facilitates the handling of unstructured data. This is particularly true of the immediate and relative modes.

6.2.5.1 Immediate Mode [OPR N,DD] - Immediate mode (mode is 2) equivalent in use to the autoincrement mode with the PC. It provides time improvements for accessing constant operands bv including the constant in the memory location immediately following the instruction word.

OPR #n,DD

Immediate Mode Example (Figure 6-27)

Symbolic	Octal Code	Instruction Name
ADD #10,R0	062700 000010	Add

Operation: The value 10 is located in the second word of the instruction and is added to the contents of R0. Just before this instruction is fetched and executed, the PC points to the first word of the instruction. The processor fetches the first word and increments the PC by two. The source operand mode is 27 (autoincrement the PC). Thus, the PC is used as a pointer to fetch the operand (the second word of the instruction) before it is incremented by two to point to the next instruction.



Figure 6-27 ADD #10,R0 Add

6.2.5.2 Absolute Addressing [OPR @ A] - This mode (mode 3) is the equivalent of immediate-deferred or autoincrement-deferred using the PC. The contents of the location following the instruction address of the operand. Immediate data is are taken as the interpreted as an absolute address (i.e., an address that remains constant no matter where in memory the assembled instruction is executed).

OPR Q#A

Absolute Mode Examples (Figures 6-28 and 6-29)

1.	Symbolic	Octal Code	Instruction Name				
	CLR @#1100	005037 001100	Clear				

Operation: Clear the contents of location 1100.



Figure 6-28 CLR @#1100 Clear

2. Symbolic

Octal Code Instruction Name

ADD @#2000,R3

063703 002000

Add

Operation:

on: Add contents of location 2000 to R3.



Figure 6-29 ADD @#2000 Add

6.2.5.3 Relative Addressing [OPR A Or OPR X(PC)] - This mode (mode 6) is assembled as index mode using R7. The base of the address calculation, which is stored in the second or third word of the instruction, is not the address of the operand, but the number which, when added to the (PC), becomes the address of the operand. This mode is useful for writing position-independent code since the location referenced is always fixed relative to the PC. When instructions are to be relocated, the operand is moved by the same amount.

OPR A or OPR X(PC) (X is the location of A relative to the instruction)

Relative Addressing Example (Figure 6-30)

Symbolic	Octal Code	Instruction Name
INC A	005267	Increment
	000054	

Operation: To increment location A, contents of memory location immediately following instruction word are added to (PC) to produce address A. Contents of A are increased by one.



Figure 6-30 INC A Increment

6.2.5.4 Relative-Deferred Addressing [OPR @A Or OPR @X(PC)] - This mode (mode 7) is similar to relative mode, except that the second word of the instruction, when added to the PC, contains the address of the address of the operand, rather than the address of the operand.

OPR @A or OPR @X(PC) (X is the location containing the address of A, relative to the instruction)

Relative-Deferred Mode Example (Figure 6-31)

Symbolic	Octal Code	Instruction Name
CLR @A	005077 000020	Clear

Operation: Add second word of instruction to updated PC to produce address of address of operand. Clear operand.



Figure 6-31 CLR @A Clear

6.2.6 Use Of The Stack Pointer As A General-Purpose Register -The processor stack pointer (SP, register 6) is in most cases the general register used for the stack operations related to program nesting. Autodecrement with register 6 "pushes" data onto the stack and autoincrement with register 6 "pops" data off the stack. Since the SP is used by the processor for interrupt handling, it has a special attribute: autoincrements and autodecrements are always done in steps of two. Byte operations using the SP in this way leave odd addresses unmodified.

6.3 INSTRUCTION SET

The rest of this chapter describes the DCJll's instruction set. Each instruction's explanation includes the instruction's mnemonic, octal code, binary code, a diagram showing the format of the instruction, a symbolic notation describing its execution and effect on the condition codes, a description, special comments, and examples.

Each instruction's explanation is headed by its mnemonic. When the word instruction has a byte equivalent, the byte mnemonic also appears.

The diagram that accompanies each instruction shows the octal op code, binary op code, and bit assignments. [Note that in byte instructions the most significant bit (bit 15) is always a one.]

Symbols:

() = contents of SS or src = source address DD or dst = destination address loc = location <--- = becomes ↑ = "is popped from stack" ↓ = "is pushed onto stack" ↓ = boolean AND ∨ = boolean OR ∨ = boolean OR ∨ = exclusive OR ~ = boolean not REG or R = register B = Byte ■ = 0 for word, 1 for byte , = concatenated

6.3.1 Instruction Formats - The following formats include all instructions used in the DCJ11. Refer to individual instructions for more detailed information.

1.	Single-Operand Group:	CLR, CLRB, COM, COMB, INC, INCB,
	(Figure 6-32)	DEC, DECB, NEG, NEGB, ADC, ADCB,
		SBC, SBCB, TST, TSTB, ROR, RORB,
		ROL, ROLB, ASR, ASRB, ASL, ASLB,
		JMP, SWAB, MFPS, MTPS, SXT,
		TSTSET, WRTLCK, XOR

_	15	_							06	05					00
Г			-	1	1	- T	1	- T			Т.	1			
		•		OP C	ODE								DD(SS)	
L		I		_	1					1					

MR 5191

Figure 6-32 Single-Operand Group

2. Double-Operand Group:

a.	Group 1:	BIT,	BITB	, BIC,	, BICB	, BIS,	BISB
	(Figure 6-33)	ADD,	SUB,	MOV,	MOVB,	CMP,	CMPB





4. Operate Group: HALT, WAIT, RTI, RESET, RTT, NOP, MFPT (Figure 6-43)

1	15				 				 			_	0 0
			1		 		- T	- T	 	1	1		
						OP	CODE						
	-		1	-	 				 				

Figure 6-43 Operate Group

Figure 6-44 Condition Group

5. Condition Code Operators (all condition code instructions) (Figure 6-44)



MR-5198

MR 5199

6. Move To/From Previous Instruction/Data Space Group: MTPD, MTPI, MFPD, MFPI

(Figure 6-45)





6.3.2 Byte Instructions - The DCJ11 includes a full complement of instructions that manipulate byte operands. Since all DCJ11 addressing is byte-oriented, byte manipulation addressing is straightforward. Byte instructions with autoincrement or autodecrement direct addressing cause the specified register to be modified by one to point to the next byte of data. Byte operations in register mode access the low-order byte of the specified register. These provisions enable the DCJ11 to perform as either a word or byte processor. The numbering scheme for word and byte addresses in memory is shown in Figure 6-46.



MR 5201

Figure 6-46 Byte Instructions

The most significant bit (bit 15) of the instruction word is set to indicate a byte instruction.

Example:

Symbolic	Octal Code	Instruction Name
CLR	0050DD	Clear word
CLRB	1050DD	Clear byte

6.3.3 List Of Instructions - The following is a list of the DCJ11 instruction set.

SINGLE-OPERAND

General

Mnemonic	Instruction	Op Code
CLR(B)	Clear destination	050DD
COM (B)	Complement destination	051DD
INC(B)	Increment destination	052DD
DEC (B)	Decrement destination	053DD
NEG (B)	Negate destination	054DD
TST (B)	Test destination	057DD
WRTLCK	Read/lock destination, write/unlock R0 into	
	destination	0073DD
TSTSET	Test destination, set low bit	0072DD

Shift and Rotate

ł	Mnemonic	Instruction	Op Code
	ASR(B) ASL(B) ROR(B) ROL(B) SWAB	Arithmetic shift right Arithmetic shift left Rotate right Rotate left Swap bytes	 062DD 063DD 060DD 061DD 0003DD

Multiple-Precision

Mnemonic	Instruction	Op Code
ADC (B)	Add carry	055DD
SBC (B)	Subtract carry	056DD
SXT	Sign extend	0067DD

PS Word Operators

Mnemonic	Instruction	Op Code
MFPS	Move byte from PS	1067DD
MTPS	Move byte to PS	1064SS

DOUBLE-OPERAND

General

Mnemonic	Instruction	Op Code
MOV(B)	Move source to destination	1SSDD
CMP(B)	Compare source to destination	2SSDD
ADD	Add source to destination	06SSDD

SUB ASH ASHC MUL DIV	Subtract Arithmeti Arithmeti Multiply Divide	source from destination c shift c shift combined	n 16SSDD 072RSS 073RSS 070RSS 071RSS
Logical			
Mnemonic	Instructi	on	Op Code
BIT(B) BIC(B) BIS(B) XOR	Bit test Bit clear Bit set Exclusive	OR	 ■ 3SSDD ■ 4SSDD ■ 5SSDD 074RDD
PROGRAM CON	ITROL		Op Code
Mnemonic	Instructio	on	Base Code
Branch			
BR BNE BEQ BPL BMI BVC BVS BCC BCS	Branch (u Branch if Branch if Branch if Branch if Branch if Branch if Branch if Branch if	nconditional) not equal (to zero) equal (to zero) plus minus overflow is clear overflow is set carry is clear carry is set	000400 001000 001400 100000 100400 102000 102400 103000 103400
Signed Cond	litional B	ranch	Op Code
Mnemonic	Instructio	n	or Base Code
BGE	Branch if	greater than or equal	002000
BLT BGT BLE	Branch if Branch if Branch if (to zero)	less than (zero) greater than (zero) less than or equal	002400 003000 003400
Unsigned Cc	onditional	Branch	Op Code or
Mnemonic	Instructio	on	Base Code
BHI BLOS BHIS BLO	Branch if Branch if Branch if Branch if	higher lower or same higher or same lower	101000 101400 103000 103400

.

Jump and Su	ubroutine	
		Op Code
		or
Mnemonic	Instruction	Base Code
JMP	Jump	0001DD
JSR	Jump to subroutine	004RDD
RTS	Return from subroutine	00020R
SOB	Subtract one and branch (if $\neq 0$)	077R00
Trap and Ir	nterrupt	
		Op Code
		or
Mnemonic	Instruction	Base Code
EMT	Emulator trap 104000 -	104377
TRAP	Trap 104400 -	104///
BPT	Breakpoint trap	000003
IOT	Input/output_trap	000004
RTI	Return from interrupt	000002
RTT	Return from interrupt	000006
	•	
Miscellaned	ous Program Control	
		Op Code
		or
Mnemonic	Instruction	Base Code
CSM	Call to supervisor mode	0070DD
MARK	Mark	0064NN
SPL	Set Priority Level	00023N
MISCELLANEC	DUS	0.0.1
		Up Code
Maamania	Technuchien	Or De de de de
Mnemonic	Instruction	Base Code
HALT	Halt	000000
WATT	Wait for interrunt	000000
DECEM	Peret external bus	000001
NEDEL	Reset external bus	000005
MEPT	Move processor type	000007
MTPD	Move to previous data space	106655
MTPI	Move to previous instruction	
	space	0066SS
MFPD	Move from previous data space	006555
MFPI	Move from previous instruction	
	space	106588
CONDITION		
CONDITION (JODE OFERAIORS	On Codo
Mnomonia	Instruction	DI Code
mnemon1C	Instruction	pase code
CLC	Clear C	000241
CLV	Clear V	000242
CL2	Closr 7	000242
		000244
СПИ	CIEdI N	000250

•

Clear all CC bits	000257
Set C	000261
Set V	000262
Set Z	000264
Set N	000270
Set all CC bits	000277
No operation	000240
	Clear all CC bits Set C Set V Set Z Set N Set all CC bits No operation

6.3.4 Single-Operand Instructions - The DCJll instructions that involve only one operand are described in the paragraphs that follow.

6.3.4.1 General -

CLR CLRB



Condition Codes: N: set if most significant bit of result is set;

				C: S	lea: et lea: et	red if n red	oth cesu	erw: lt :	ise is (); c]	leare	ed of	ther	wise
Descri	ptior	1:	i e c F	Word: Replaces the contents of the destination address by their logical complement. (Each bit equal to 0 is set and each bit equal to 1 is cleared.) Byte: Same.										
Exampl	e:		(COM	R0									
			1	Befo	re					Aft	er [,]			
				(R0)	= (0133	333			(R()) =	1644	444	
				NZV 011	C 0					N2 10	2VC			
INC INCB	·		•											
	INCREME	NT DST												■052DD
	15				· · · · ·	·····			06	05				00
	0/1	0 (0 0	1	0	1	0	1	0			DD		
Operat	ion:			(dst) <		(dst) +	1	-				MR 11506

Condition Codes: N: set if result is < 0; cleared otherwise Z: set if result is 0; cleared otherwise V: set if (dst) held 077777; cleared otherwise C: not affected

Description: Word: Add 1 to the contents of the destination. Byte: Same.

Example: INC R2

 Before
 After

 (R2) = 000333
 (R2) = 000334

 NZVC
 NZVC

 0000
 0000

DEC DECB



Condition Codes: N: set if result is < 0; cleared otherwise

			Z: V: C:	se se nc	et : et : ot a	if r if (affe	dst cte	lt :) wa đ	is (as 1	; cle 00000	eared); cl	l oti eare	nerwise ed other	wise
Descrip	tion:		Wo de By	rd: sti te:	: Su inat : Sa	ubtr tion ame.	act	1 1	Eron	the	cont	ents	s of the	•
Example	:		DE	CI	R5									
			Be	for	re					Afte	er			
			(R	5)	= (0000	01			(R5)) = 0	0000	00	
			NZ 10	VC 00						NZVC 0100				
NEG NEGB													•	
N	EGATE DST												■054DD	
r.	<u>15</u>			, 1		·	,		06	05			· · · · · · · · · · · · · · · · · · ·	
L						·						JU J		
Operatio	on:		(đ	st)	<-		(đ:	st)					Mit 11503	
Conditio	on Code	25:	N: Z: V: C:	se se cl	et i et i et i .ear	lf r lf r lf r red	esu esu esu if	lt i lt i lt i resu	s < s 0 s 1	0; c ; cle 00000 is 0;	lear ared); cl set	ed c oth eare oth	otherwis herwise ed other herwise	e wise
Descript	tion:		Wo ad is no po By	rd: dre re tat sit	Re ss pla ion ive Sa	epla by aced th co ame.	ces its by e mo unt	the 2's its ost erpa	e co s co self neg art.	ntent mplem . (Ir ative)	s of nent. n 2's e num	the Not con ber	e destin te that nplement has no	ation 100000
Example	:		NE	GF	٥٥									
			Ве	for	e					Afte	er			
			(R	0)	= (0000	10			(R0)	= 1	7777	70	
			NZ 00	VC 00						NZVC 1001				
TST TSTB														
т	EST DST												■057DD	
Г	15			ı		, ,	·	·	06	05				
L				·	Ļ	· ·							MR 11501	

Operation:

•

_

(dst) <-- (dst)

Condition Codes:	N: set if result is < Z: set if result is 0; V: cleared C: cleared	0; cleared otherwise cleared otherwise
Description:	Word: Sets the condit: to the contents of the contents of dst remain Byte: Same.	ion codes N and Z according e destination address; the n unmodified.
Example:	TST Rl	
	Before	After
	(R1) = 012340	(R1) = 012340
	NZVC 0011	NZVC 0000

٩

WRTLCK

	READ/LOCK WRITE/UNLC	DESTINA OCK ROIN	TION TO DESTI	NATION		:		06	05			0073DD 00
	0 0		0 1	1	1	1	1 1	0				
Operat	ion:		(ds	t) <-	-	(R0)						MR 11498
Condit	ion Co	des:	N: Z: V: C:	set set clear uncha	if if red ang	R0 < R0 = ed	: 0 : 0					
Descri	ption:		Wri bus	tes d loci	con k.	tent If	s o: mod	E R e i:	0 in s 0,	to de trap	stina s to	ation using 10.

TSTSET

	TEST 15	DESTIN	ATION	AND S	ET LOV	V ВІТ				06	05					0072DD 00	
	0	0	0	0	1 1	1 1	1 1	0	1	0)	1		
Operat	ion	:	-	(R0)	<	- (đ	lst)	, (dst) <-	- ((dst)	V	000	MR 11499 0001	(octal)
Condit	ion	Cod	les:	N Z V C	: s : s : c : g	et et lea: ets	if F if F red con	10 < 10 =	0 0 ts	of	dest	ina	ation	ъ	it (D.	
Descri	pti	on:		R i d	ead n R est	s/lo 0. ina	ocks Wri tior	de tes	sti /un If	nat loc mod	ion ks (e is	w01 (R0)	d an V 1 , tra	id i ips	sto: nto to	res : 10.	it

6.3.4.2 Shifts And Rotates - Scaling data by factors of two is accomplished by the shift instructions:

ASR -- Arithmetic shift right

ASL -- Arithmetic shift left

The sign bit (bit 15) of the operand is reproduced in shifts to the right. The low-order bit is filled with 0s in shifts to the left. Bits shifted out of the C bit, as shown in the following instructions, are lost.

The rotate instructions operate on the destination word and the C bit as though they formed a 17-bit "circular buffer." These instructions facilitate sequential bit testing and detailed bit manipulation.

ASR ASRB



Operation:	(dst) < (dst) shifted one place to the right
Condition Codes:	N: set if high-order bit of result is set (result < 0); cleared otherwise Z: set if result = 0; cleared otherwise V: loaded from exclusive OR of N bit and C bit (as set by the completion of the shift operation) C: loaded from low-order bit of destination
Description:	Word: Shifts all bits of the destination right one place. Bit 15 is reproduced. The C bit is loaded from bit 0 of the destination. ASR performs signed division of the destination by 2. Byte: Same.

Example:



ASLB

,	ARITHMETIC	SHIFT L	EFT									■063DD	
-	15								06	05		00	
	0/1 0	0	0	1	1	0	0	1	1		DD		
•												MR 11510	
Operati	.on:		(đ	st)	<-	- ((dst) s	hift	ed on	e place	to the	left
Conditi	on Coà	les:	N:	se (r	t i esul	f h lt	nigh < 0	-or	der clea	bit o red o	f resul therwis	t <mark>is se</mark> t e	
			Z: V:	se lo (a	t i ade s s era	f r d w et	esu vith by	lt ex the	= 0; clus com	clea sive O pleti	red oth R of N on of t	erwise bit and (he shift	C bit
			C:	10	ade	d v	vith	hi	gh-c	order	bit of	destinat	ion
Decarir	tion		พุ	r.d.	Sh	; , ,	- C - 2	רו	hite	e of t	he dest	ination	loft

Description: Word: Shifts all bits of the destination left one place. Bit 0 is loaded with a 0. The C bit of the status word is loaded from the most significant bit of the destination. ASL performs a signed multiplication of the destination by 2 with overflow indication. Byte: Same.



Operation:

(dst) <-- (dst) rotate right one place

Condition Codes:

N: set if high-order bit of result is set

- (result < 0); cleared otherwise
 2: set if all bits of result = 0; cleared
 otherwise</pre>
- V: loaded with exclusive OR of N bit and C bit (as set by the completion of the rotate operation)
- C: loaded with low-order bit of destination

Description: Word: Rotates all bits of the destination right one place. Bit 0 is loaded into the C bit and the previous contents of the C bit are loaded

Example:





6.3.4.3 Multiple-Precision - It is sometimes necessary to do arithmetic operations on operands considered as multiple words or bytes. The DCJ11 makes special provision for such operations with the instructions ADC (add carry) and SBC (subtract carry) and their byte equivalents.

For example, two 16-bit words may be combined into a 32-bit double-precision word and added or subtracted as shown below.



Description:	Word: desti addit the h Byte:	Adds the nation. S ion of the igh-order Same.	contents of th This permits th e low-order wor result.	ne C bit to the ne carry from the ds to be carried to
Example:	Doubl follo	e-precision wing inst	on addition may ruction sequenc	y be done with the ce.
	ADD ADC ADD	A0,B0 B1 A1,B1	;add low-ord ;add carry i ;add high-or	ler parts into high-order der parts
SBC SBCB				
SUBTRACT CARRY				■ 0 56DD
15			06 05	00

Operation:	(dst) < (dst) - (C)
Condition Codes:	N: set if result < 0; cleared otherwise Z: set if result = 0; cleared otherwise V: set if (dst) was 100000; cleared otherwise C: set if (dst) was 0 and C was 1; cleared otherwise
Description:	Word: Subtracts the contents of the C bit from the destination. This permits the carry from the subtraction of two low-order words to be subtracted from the high-order part of the result. Byte: Same.
Example:	Double-precision subtraction is done by:

SUB	A0,B0
SBC	Bl
SUB	Al,Bl

SXT

	SIGN EX	TEND												0067DD	
	15								06	05				00	
	0	0	0	0	1	1 0	1	1	1		· ·		, ,		
Operat	tion:			(đ (đ	st) st)	< (<]) if L if	N N	bit bit	is is	cle set	ar		MR-11574	
Condi	tion	Cođ	es:	N: Z: V: C:	not set cle not	affe if N ared affe	ecte N bi ecte	d ti	ls cl	Lea	r			-	
Descri	iptic	n:		If	the	cond	liti	on	code	e b:	it N	is	set,	a -1	is

placed in the destination operand; if the N bit is clear, a 0 is placed in the destination operand. This instruction is particularly useful in multiple-precision arithmetic because it permits the sign to be extended through multiple words.

Example:

SXT A

Before	After
(A) = 012345	(A) = 177777
NZVC 1000	NZVC 1000

6.3.4.4 PS Word Operators -

MFPS

				<u> </u>	<u>vo</u>				_			15
1 0 1 1 1 DD	DD	г г	1	1	' 1	່ບ	់រ	1	0	່ວ	0	۱. '

Operation:	(dst) <	PS
-	dst lower	8 bits

Condition Codes: N: set if PS <7> = 1; cleared otherwise Z: set if PS <7:0> = 0; cleared otherwise V: cleared C: not affected

Description: The 8-bit contents of the PS are moved to the effective destination. If the destination is mode 0, PS bit 7 is sign-extended through the upper byte of the register. The destination operand address is treated as a byte address.

Example: MFPS R0

Before	After
R0 [0]	R0 [000014]
PS [000014]	PS [000000]

MTPS

15							08	07					0
1	0	0	0	1	1	0	' 1	0	0	•	•	SS	T
		ب			· · ·	, ,		l °				SS	

Operation: PS <-- (src)

Condition Codes: Set according to effective SRC operand bits <3:0>

Description: The eight bits of the effective operand replace the current contents of the lower byte of the PS. The source operand address is treated as a byte address. Note: The T bit (PS bit 4) cannot be set with this instruction. The SRC operand remains unchanged. This instruction can be used to change the priority bits (PS bits <7:5>) in the PS only in kernel mode. If not in kernel mode, PS bits <7:5> cannot be changed.

Example: MTPS R1

Before	After
(R1) = 000777	(R1) = 000777
(PS) = XXX000	(PS) = XXX357
NZVC	NZVC
0000	1111

6.3.5 Double-Operand Instructions - Double-operand instructions save instructions (and time) since they eliminate the need for "load" and "save" sequences such as those used in accumulator-oriented machines.

6.3.5.1 General -

MOV MOVB



extension). Otherwise, MOVB operates on bytes exactly as MOV operates on words.

Example:

MOV XXX,Rl ;loads register l
with the contents of
memory location; XXX
represents a
programmer-defined
mnemonic used to
represent a memory
location
MOV #20,R0 ;loads the number 20
into register 0; #

- into register 0; # indicates that the value 20 is the operand
- MOV @#20,-(R6) ;pushes the operand contained in location 20 onto the stack

MOV (R6)+,@#177566 ;pops a stac

MOV R1,R3

MOVB @#177562,@#177566

;pops the operand off a stack and moves it into memory location 177566 (terminal print buffer)

;performs an inter-register transfer

;moves a character from the terminal keyboard buffer to the terminal printer buffer

CMP CMPB



Operation:

(src) - (dst)

Condition Codes:

N: set if result < 0; cleared otherwise Z: set if result = 0; cleared otherwise V: set if there was arithmetic overflow; that is, operands were of opposite signs and the sign of the destination was the same as the sign of the result; cleared otherwise C: cleared if there was a carry from the result's most significant bit; set otherwise Description:

Compares the source and destination operands and sets the condition codes, which may then be used for arithmetic and logical conditional branches. Both operands are not affected. The only action is to set the condition codes. The compare is customarily followed by a conditional branch instruction. Note: Unlike the subtract instruction, the order of operation is (src) - (dst), not (dst) - (src).

ADD

ADD SRC TO DST		,	065SDD
	2 11 0 SS	06 05 DD	00
Operation:	(dst) < (src) -	⊢ (dst)	MR-11563
Condition Codes:	N: set if result Z: set if result V: set if there result of the operands were was of the opp C: set if there most signification	< 0; cleared of = 0; cleared of was arithmetic of operation; that of the same sid posite sign; clo was a carry from ant bit; cleared	therwise therwise overflow as a t is, both gn and the result eared otherwise n the result's d otherwise
Description:	Adds the source of operand and store address. The or destination are source are not ad addition is perfor equivalent byte r	operand to the operand to the operand to the result at iginal contents lost. The contents fected. Two's ormed. Note: The mode.	destination t the destination of the ents of the complement ere is no
Example:	Add to register:		ADD 20,R0
	Add to memory:		ADD R1,XXX
	Add register to a	register:	ADD R1,R2
	Add memory to mer	nory:	ADD @#17750,XXX
	XXX is a programm memory location.	ner-defined mner	nonic for a

SUB



Condition Codes: N: set if result < 0; cleared otherwise Z: set if result = 0; cleared otherwise V: set if there was arithmetic overflow as a result of the operation; that is, if operands were of opposite signs and the sign of the source was the same as the sign of the result; cleared otherwise

C: cleared if there was a carry from the result's most significant bit; set otherwise

Description: Subtracts the source operand from the destination operand and leaves the result at the destination address. The original contents of the destination are lost. The contents of the source are not affected. In double-precision arithmetic the C bit, when set, indicates a "borrow." Note: There is no equivalent byte mode.

Example: SUB R1,R2

Befo	re		Afte	r	
(R1)	=	011111	(R1)	=	011111
(R2)	=	012345	(R2)	=	001234
NZVO	2		NZV	С	
111	1		000	0	

ASH



Z: set if result = 0 V: set if sign of register changed during shift C: loaded from last bit shifted out of register

Description: The contents of the register are shifted right or left the number of times specified by the source operand. The shift count is taken as the low-order six bits of the source operand. This number ranges from -32 to +31. Negative is a right shift and positive is a left shift.

ASHC

	ARITH	METIC	SHIFT CO	MBINED							■073R	SS
	15			 		09	08	06	05		00	- -
	0	1	1	1 0	1	1	R			SS		
											MA-118	61
Operat	tion	:		R, The ric	R ∨ e dou jht o	l < uble or l	R, word eft wl	R∨ is s here	l shifte NN =	ed NN pl (src)	laces t	o the
Condit	tion	Co	des:	N: Z: V: C:	set set load (loa the	if if if led led aded 32-	result result sign 1 with 1 with 1 with 1 bit op	c < 0 c = 0 pit c nigh- Low-c the perar	hange order last d)	es durin bit wh bit whe bit shi	ng shif hen lef en righ ifted o	t t shift; t shift ut of
Descri	iptio	on:	- 1	The ORe R V shi spe is sou	e cor ed wi fted cifi take rce +31.	nten ith (bit led en a ope . N	ts of l are s<15:(ght or by the s the rand. egativ	the trea)>) a c lef e shi low- Thi ye is	regis nd R t the ft cc order s num a ri	ster and as one ((bits<) e number ount. 1 six bi ber ran .ght shi	d the r 32-bit 31:16>) f of tin The shi its of nges fro ift and	egister word. are mes ft count the om -32 positive

When the register chosen is an odd number, the register and the register ORed with 1 are the same. In this case, the right shift becomes a rotate. The 16-bit word is rotated right the number of times specified by the shift count.

MUL

15						09	08		06	05		
~				1	1	1				1	 1	 1
0	1	1	T	0	0	0	1	R			SS	

Operation: R, $R \vee 1 < --R \times (src)$

is a left shift.

Condition Codes:

N: set if product < 0 Z: set if product = 0

- V: cleared C: set if the result is less than -2 ** 15
 - or greater than or equal to 2 **15 1.

Description: The contents of the destination register and source taken as 2's complement integers are multiplied and stored in the destination register and the succeeding register, if R is even. If R is odd, only the low-order product is stored. Assembler syntax is: MUL S,R. (Note that the actual destination is R, R v 1, which reduces to just R when R is odd. DIV

15						09	08		06	05			00
	1	T	1	-		T .	Ι	-	1	- T		T	1
0 1		1	1	0	0	1		R			SS		

Operation: R, $R \vee 1 < --R$, $R \vee 1/(src)$

Condition Codes: N: set if quotient < 0
Z: set if quotient = 0
V: set if source = 0 or if the absolute value
of the register is larger than the absolute
value of the instruction in the source. (In
this case the instruction is aborted because
the quotient would exceed 15 bits.)
C: set if divide by zero is attempted.
Description: The 32-bit 2's complement integer in R and
R v 1 is divided by the source operand. The
quotient is left in R; the remainder is of the</pre>

6.3.5.2 Logical - These instructions have the same format a those in the double-operand arithmetic group. They permi operations on data at the bit level.

same sign as the dividend. R must be even.

BIT BITB



Operation: $(src) \land (dst)$

Condition Codes: N: set if high-order bit of result set; cleared otherwise Z: set if result = 0; cleared otherwise V: cleared C: not affected

Description: Performs logical AND comparison of the source and destination operands and modifies condition codes accordingly. Neither the source nor the destination is affected. The BIT instruction may be used to test whether any of the corresponding bits set in the destination are also set in the source, or whether all corresponding bits set in the destination are clear in the source.

MR 11557

R3 = 0	000	000	000	011 000	0
Before				Aft	er
N2VC 1111		•		NZV(000)	

BIC BICB



Operation:

(dst) <-- ~(src) 🔨 (dst)

Condition Codes: N: set if high-order bit of result set; cleared otherwise 2: set if result = 0; cleared otherwise V: cleared C: not affected

Description: Clears each bit in the destination that corresponds to a set bit in the source. The original contents of the destination are lost. The contents of the source are not affected.

Example:

BIC R3,R4

Before					Aft	er		
(R3) = 0 (R4) = 0	01234 01111				(R3) (R4)) = (0012: 00010	34 Dl
NZVC 1111					NZV0			
Before:	(R3) (R4)	=	0 0	000	001 001	010 001	011 001	100 001
After:	(R4)	#	0	000	000	001	000	001

BIS BISB



Condition Codes: N: set if high-order bit of result set; cleared otherwise

- Z: set if result = 0; cleared otherwise
- V: cleared
- C: not affected

Description:

Performs an inclusive OR operation between the source and destination operands and leaves the result at the destination address; that is, corresponding bits set in the source are set in the destination. The contents of the destination are lost.

Example:

BIS RO,RI	<u> </u>				,			
Before					Afte	er		
(R0) = 00 (R1) = 00)1234)1111				(R0) (R1)) = (00123 00133	34 35
NZVC 0000					NZV(
Before:	(R0) (R1)	-	0 0	000 000	001 001	010 001	011 001	100 001
After:	(R1)	=	0	000	001	011	011	101

XOR

E	EXCLUSIVE OR													074RDD			
r	15		r				09	08	r	06	05					00	
l	0 1 1			1			0		R .				DD			·	
Operat	ion	:		(ðst) <·		(reg	n) \	÷ (4	lst))				MR.11559	
Condition Codes:				N Z V C	: s : s : c : n	et et lea: ot a	if : if : red affe	resu resu ecte	lt lt d	< 0 = 0	; c] ; c]	lear Lear	eđ eđ	oth oth	erw: erw:	ise ise	
Description:				T O T T	The exclusive OR of the register and destination operand is stored in the destination address. The contents of the register are not affected. The assembler format is XOR R,D.												
Example:			Х	XOR R0,R2													
				В	efo	re					Ai	Eter					
				(R0) R2)	= (0012 0012	234 111			(I (I	RO) R2)	= 0 = 0	012 003	34 25		
				N 1	ZVC 111						N2 00	ZVC 001					

Before: (R0) = 0 000 001 010 011 100 (R2) = 0 000 001 001 001 001 After: (R2) = 0 000 000 011 010 101

6.3.6 Program Control Instructions - The following paragraphs describe the DCJ11 instructions that affect program control.

6.3.6.1 Branches - These instructions cause a branch to a location defined by the sum of the offset (multiplied by 2) and the current contents of the program counter if:

- 1. The branch instruction is unconditional.
- 2. It is conditional and the conditions are met after testing the condition codes (NZVC).

The offset is the number of words from the current contents of the PC, forward or backward. Note that the current contents of the PC point to the word following the branch instruction.

Although the offset expresses a byte address, the PC is expressed in words. The offset is automatically multiplied by 2 and sign-extended to express words before it is added to the PC. Bit 7 is the sign of the offset. If it is set, the offset is negative and the branch is done in the backward direction. If it is not set, the offset is positive and the branch is done in the forward direction.

The 8-bit offset allows branching in the backward direction by 200 (octal) words (400 octal bytes) from the current PC, and in the forward direction by 177 (octal) words (376 octal bytes) from the current PC.

The DCJll assembler typically handles address arithmetic for the user and computes and assembles the proper offset field for branch instructions in the form:

Bxx loc

Bxx is the branch instruction and loc is the address to which the branch is to be made. The assembler gives an error indication in the instruction if the permissible branch range is exceeded. Branch instructions have no effect on condition codes. Conditional branch instructions where the branch condition is not met are treated as NOPs.

BR
PC <-- PC + (2 \times offset) Operation:

Not affected Condition Codes:

Provides a way of transferring program control Description: within a range of -128 to +127 words with a one word instruction.

New PC address = updated PC + $(2 \times offset)$

Updated PC = address of branch instruction +2

With the branch instruction at location 500, the Example: following offsets apply.

> New PC Address Offset Code Offset (decimal)

474	375	-3
476	376	-2
500	377	-1
502	000	0
504	001	+1
506	002	+2

BNE

BRANCH IF NOT EQUAL (TO ZERO)

001000 PLUS OFFSET

15						08	07 00
	,	 0	0	0	1	0	OFFECT
Ľ		 			,	<u> </u>	

MR 5232

Operation:

PC <-- PC + (2 \times offset) if Z = 0

Not affected Condition Codes:

Description: Tests the state of the Z bit and causes a branch if the Z bit is clear. BNE is the complementary operation of BEQ. It is used to test: (1) inequality following a CMP, (2) that some bits set in the destination were also in the source following a BIT operation, and (3) generally, that the result of the previous operation was not 0.

Branch to C if $A \neq B$ Example:

CMP A,B ;compare A and B BNE C ; branch if they are not equal

Branch to C if $A + B \neq 0$

ADD	A,B	;add A to B			
BNE	С	;branch if the	result	is	not
		egual to 0			

BEQ																00	1400			-	
. 1	BRANC	HIFEQ	UAL (I	IU ZEI	101					00	07					U.	14001	PLUS	JFFSE	ſ	
		······			,			1		1	07	γ	7	+					1 00	ר	
								, 		'	ا سرچيندان		<u></u>	1	- <u>5</u> E 1	. L					
Opera	atio	n:]	PC	<-	- F	·C·	+	2	×	off	set)	if	Z	=	ı		MR 52:	13	
Condi	tio	n Co	des:	: 1	lot	: a	ffe	ct	eđ							,					
Descr	ipt:	ion:		: 1 1 1 1 1	res If In In In In	sts Z llo th llo at	th is wir e c wir the	se ig ies ig ig i	sta t. a C tin a B esu	It MP at IT	of is op ion op of	the use erat we erat	e Z ed t tion te a tion e pr	bit o to , (lso , an evio	an est 2) se nd ous	d : th (3 0	cau (1) at in) g per	ses eq no the ene ati	a ual bit so ral on	branc ity s set urce ly, was 0	:h
Examp	le:			E	Bra	anc	h t	:0 (C i	f	= A	В	(A -	В	= 0)					
				(E	CMF Beç	2 A	,В						;co ;br	mpa: anci	re 2 h i	A f	and the	B ya	re	equal	
				E	Bra	anc	h t	:0 (C i	f	A +	в :	= 0								
) E	ADI BEÇ	2 2	A,E C						;ad ;br	d A anci	to h i	B f	the	re	sul	t = 0	1
BPL	BRAN 15		LUS				.			08	07						10000	0 PLU:		.ет)	
BPL	BRAN 15 1	ICH IF P	LUS		1	0	0		0	08	07)FFSE	T	10000	0 PLU:		.ет)	-
BPL Opera	BRAN 15 1 ation	0 0	PLUS 0	0	PC	0	- P		• • ► (08 0 2	07 ×	off:	, set)	if	DFFSE N	T	10000	0 PLU:	S OF FS	лет) 	
BPL Opera Condi	BRAN 15 1 1 1 1 1 1 1 1 1 1 1 1 1		o des:	0 	PC Not	_0 < = a	- P ffe	C -	o ⊢ (ed	08 0 2	07 ×	offs	set)	, c if	DFFSE N	T	10000 	0 PLU:	S OFFS	ет) ,	
BPL Opera Condi Descr	BRAN 15 1 ation tion tipt:	n: n Con ion:	des:	o I I I I I I C	PC Not If con	o < a sts N npl	- F ffe th is eme		0 + (ed star ary	08 0 2	of of pos per	off: the itiv	e N ve r on o	if bit esu f B	N an lt) MI.	т = d	10000 0 cau BP	ses L i	s offs	branc	h
BPL Opera Condi Descr BMI	BRAN 15 1 ation tion tion	n: n Co ion:	des:	0 I I : : : : : :	PC Not If	o < a sts N mpl	- P ffe th is eme		o + (ed star ary	08 0 2	of of pos per	off: the itiv	e N ve r on o	if bit esu f B	N an lt) MI.	т = •	10000 0 cau BP	ses L i	a st	branc	ħ
BPL Opera Condi Descr BMI	BRAN 15 1 1 1 1 1 1 1 1 1 1 1 1 1	ion:	des:	I I I I I I I I I I I I I I I I I I I	PC Not If com	o < a sts N npl	- F ffe th is eme		o + (ed sta eary	08 0 2	of pos per	off: the itiv	set) e N ye r on o	if bit esu f B	N an lt) MI.	T =	10000 0 Cau BP	Ses L i		branc	h
BPL Opera Condi Descr BMI	BRAN 15 1 1 1 1 1 1 1 1 1 1 1 1 1	ion:	des:	o I I I I I I I I I I I I I I I I I I I	PC Not If con	o < a sts N npl	- P ffe tr is eme		0 + (ed sta eary	08 0 2 1 te (0 08	of pos per	off: the itiv	set) e N ye r on o	if bit esu f B	offse N an lt) MI.	т = •	10000 0 Cau BP	Ses L i		branc	h
BPL Opera Condi Descr BMI	BRAN 15 1 1 1 1 1 BRAN 15 1		des:		PC Not If con	o < a sts N npl	- F ffe th is eme		0 + (ed sta eary 0	08 0 2 1 0 0 8	of pos per	off: the itiv atio	set) ze n ze r on o	if bit esu f B	offse N an lt) MI.	т = d	10000 0 Cau BP	Ses L i		branc	h
BPL Opera Condi Descr BMI	BRAN 15 1 ation tion BRAN 15 1 1 1 1 1 1 1 1 1 1 1 1 1	о	des:		PC Not If con	o < a sts N npl	- P ffe th is eme o - F		0 + (ed stary 0 + (08 0 2 1 te (0 08 1 2	of pos per	off: the itiv atio	set)	if bit esu f BN	N an lt) MI. FFSE	T = d •	10000 0 Cau BP	Ses L i	S OFFS	eT	h

•

Description: Tests the state of the N bit and causes a branch if N is set. It is used to test the sign (most

significant bit) of the result of the previous operation), branching if negative. BMI is the complementary function of BPL.

102000 PLUS OFFSET BRANCH IF OVERFLOW IS CLEAR 08 07 00 15 OFFSET 0 0 0 1 0 0 0 MR-5236 **Operation:** PC <-- PC + $(2 \times \text{ offset})$ if V = 0 Condition Codes: Not affected Description: Tests the state of the V bit and causes a branch if the V bit is clear. BVC is complementary operation to BVS. BVS 102400 PLUS OFFSET BRANCH IF OVERFLOW IS SET 08 07 00 15 OFFSET 1 0 0 0 0 1 0 MR 5237 Operation: PC <-- PC + (2 \times offset) if V = 1 Condition Codes: Not affected Description: Tests the state of the V bit (overflow) and

causes a branch if V is set. BVS is used to detect arithmetic overflow in the previous operation.

BCC

103000 PLUS OFFSET BRANCH IF CARRY IS CLEAR 08 07 00 15 0 0 0 0 0 OFFSET 1 1 1 MR 5238 PC <-- PC + (2 \times offset) if C = 0 Operation: Condition Codes: Not affected

Description: Tests the state of the C bit and causes a branch if C is clear. BCC is the complementary operation of BCS.



BVC

Operation: $PC \leftarrow PC + (2 \times offset)$ if C = 1

Condition Codes: Not affected

Description: Tests the state of the C bit and causes a branch if C is set. It is used to test for a carry in the result of a previous operation.

6.3.6.2 Signed Conditional Branches - Particular combinations of the condition code bits are tested with the signed conditional branches. These instructions are used to test the results of instructions in which the operands were considered as signed (2's complement) values.

Note that the sense of signed comparisons differs from that of unsigned comparisons in that in signed, 16-bit, 2's complement arithmetic the sequence of values is as follows.

Whereas, in considered to	unsigned, be:	16-bit	arithmetic,	the
smallest negative	000001 000000 177777 177776			
-	•			
largest positive	077777			

sequence is

highest	177777
	•
	•
	•
	•
	•
	•
	000002
•	000001
lowest	000000

BGE																		
	BRAN	ICH I	FGF	REATE	RTHA	N OR E	QUAL							(002000	PLUS	DFFSET	
	(TO Z	ERO)						08	07			00					
	0	- T	0	0	0	0	۱	0	0		•		OFF	, SET				
	.			<u>.</u>				L									MR-5240	
Opera	atio	on:			Р	C <-	P	C +	(2	× 0:	ffse	t) :	if N	\checkmark	V =	0		· .
Condi	itic	on (Cod	des:	N	ot a	affe	cte	3				•					
Descr	Condition Codes: Not affected Description: Causes a branch if N and or both set. BGE is the of BLT. Thus, BGE will when it follows an open addition of two positive cause a branch on a 0 m										nd N he d l al rati ve r resu	/ ar comp Lway ion numb ilt.	e e: leme s ca that ers.	ithe enta ause ca B	r b ry a use GE	oth o opera branc d will	clear ation ch also	
BLT																		

-					
	BRANCH	١F	LESS	THAN	(ZERO)



PC <-- PC + (2 \times offset) if N \checkmark V = 1 **Operation:**

Condition Codes: Not affected

Description:

Causes a branch if the exclusive OR of the N and V bits is one. Thus, BLT will always branch following an operation that added two negative numbers, even if overflow occurred. In particular, BLT will always cause a branch if it follows a CMP instruction operating on a negative source and a positive destination (even if overflow occurred). Further, BLT will never cause a branch when it follows a CMP instruction operating on a positive source and negative destination. BLT will not cause a branch if the result of the previous operation was 0 (without overflow).



BLE

BRANCH IF LESS THAN OR EQUAL (TO ZERO)

003400 PLUS OFFSET

15						08	07			00
	т — т О ·	0	· · · · ·	 	,			0650		
		0	بــــــــــــــــــــــــــــــــــــ	 ' '	•			1	5C I	

Operation:

PC <-- PC + (2 × offset) if Z \bigvee (N \checkmark V)^{MR 5243} = 1

Condition Codes: Not affected

Description: Operation is similar to BLT, but in addition will cause a branch if the result of the previous operation was 0.

6.3.6.3 Unsigned Conditional Branches - The unsigned conditional branches provide a means for testing the result of comparison operations in which the operands are considered as unsigned values.

BHI

BRANCH IF HIGHER

101000 PLUS OFFSET

15							80	07			00
1	0	0	0	0	0	1	1		OFESET	· · · ·	
						·					

Operation: $PC \leftarrow PC + (2 \times offset)$ if C = 0 and Z = 0

Condition Codes: Not affected

Description: Causes a branch if the previous operation caused neither a carry nor a 0 result. This will happen in comparison (CMP) operations as long as the source has a higher unsigned value than the destination.



Condition Codes: Not affected

Description: Causes a branch if the previous operation caused either a carry or a 0 result. BLOS is the complementary operation of BHI. The branch will occur in comparison operations as long as the source is equal to or has a lower unsigned value than the destination. BHIS

BRANCH IF HIGHER OR SAME

103000 PLUS OFFSET

48 5246

This

MB 5247

15							08	07 00	_
1	0	0	0	0	1	1	0	OFFSET]
						I			

PC <-- PC + $(2 \times \text{ offset})$ if C = 0 **Operation:**

Condition Codes: Not affected

BHIS is the same instruction as BCC. Description: mnemonic is included for convenience only.

BLO

BRANCH IF LOWER

103400 PLUS OFFSET

15							08	07 00
	Y					T		
1	0	0	0	0	1	1	1	OFFSET
						1	h	l l l l l

PC <-- PC + $(2 \times \text{offset})$ if C = 1 Operation:

Condition Codes: Not affected

Description: BLO is the same instruction as BCS. This mnemonic is included for convenience only.

6.3.6.4 Jump And Subroutine Instructions - The subroutine call in DCJ11 provides for automatic nesting of subroutines, the reentrancy, and multiple entry points. Subroutines may call other subroutines (or indeed themselves) to any level of nesting without making special provision for storage of return addresses at each level of subroutine call. The subroutine calling mechanism does not modify any fixed location in memory, and thus provides for This allows one copy of a subroutine to be shared reentrancy. among several interrupting processes.

JMP



transferred to any location in memory (no range limitation) and can be accomplished with the full flexibility of the addressing modes, with the exception of register mode 0. Execution of

a jump with mode 0 will cause an "illegal instruction" condition, and will cause the CPU to trap to vector address four. (Program control cannot be transferred to a register.) Register-deferred mode is legal and will cause program control to be transferred to the address held in the specified register. Note that instructions are word data and must therefore be fetched from an even-numbered address.

Deferred-index mode JMP instructions permit transfer of control to the address contained in a selectable element of a table of dispatch vectors.

Example:	First:	
	JMP FIRST	;transfers to FIRST
	• • • • •	
	JMP @LIST	;transfers to location pointed to at LIST
	• • • • •	
	List:	
	FIRST	;pointer to FIRST
	JMP @(SP)+	;transfer to location pointed to by the top of

JSR

JUMP TO SUBROUTINE 004 RDD 09 15 08 06 05 00 0 0 0 0 ٥ 1 0 DD R MR-11556

Operation:

(tmp) <-- (dst) (tmp is an internal processor register)

the stack, and remove the pointer from the stack

↓ (SP) <-- reg (Push reg contents onto processor stack)

reg <-- PC (PC holds location following JSR; this
address now put in reg)</pre>

PC <-- (dst) (PC now points to subroutine destination)

Description: In execution of the JSR, the old contents of the specified register (the "linkage pointer") are automatically pushed onto the processor stack and new linkage information is placed in the register. Thus, subroutines nested within subroutines to any depth may all be called with the same linkage register. There is no need either to plan the maximum depth at which any particular subroutine will be called or to include instructions in each routine to save and restore the linkage pointer. Further, since all linkages are saved in a reentrant manner on the processor stack, execution of a subroutine may be interrupted. The same subroutine may be reentered and executed by an interrupt service routine. Execution of the initial subroutine can then be resumed when other requests are satisfied. This process (called "nesting") can proceed to any level.

A subroutine called with a JSR reg,dst instruction can access the arguments following the call with either autoincrement addressing, (reg) +, if arguments are accessed sequentially, or by indexed addressing, X(reg), if accessed in random order. These addressing modes may also be deferred, @(reg)+ and @X(reg), if the parameters are operand addresses rather than the operands themselves.

JSR PC, dst is a special case of the DCJ11 subroutine call suitable for subroutine calls that transmit parameters through the general registers. The SP and the PC are the only registers that may be modified by this call.

Another special case of the JSR instruction is JSR PC,@(SP) +, which exchanges the top element of the processor stack with the contents of the program counter. This instruction allows two routines to swap program control and resume operation from where they left off when they are recalled. Such routines are called "coroutines."

Return from a subroutine is done by the RTS instruction. RTS reg loads the contents of reg into the PC and pops the top element of the processor stack into the specified register.

Example:

	-	R5	R6	R7
SBCALL:	JSR R5,SBR	#1	n	SBCALL
SBCALL+4:	ARG 1			
	ARG 2			
	•			
	•			
SBCALL+2+2M:	ARG M			
CONT:	Next Instruction	#1	n	CONT
	•			
CDD.		000111.4		6 00
BDR:	MOV (R5)+,dst 1 MOV (R5)+ de+ 2	SBCALL+4	n-2	SBR

6-54

EXIT:

MOV (R5)+,dst M SBC Other Instructions CON RTS R5 CON

SBCALL+2+2M CONT CONT

n-2 EXIT



RTS



Operation:

PC <-- (reg) (reg) <-- (SP) ↑

Description:

Loads the contents of the register into PC and pops the top element of the processor stack into the specified register.

Return from a nonreentrant subroutine is typically made through the same register that was used in its call. Thus, a subroutine called with a JSR PC, dst exits with a RTS PC and a subroutine called with a JSR R5, dst, may pick up parameters with addressing modes (R5) +, X(R5), or @X(R5) and finally exits, with an RTS R5.

Example:

RTS R5





SOB

SUBTRACT ONE AND BRANCH (IF # 0)



Not affected

Operation:

(R) <-- (R) - 1; if this result \neq 0, then PC <-- PC - (2 x offset); if (R) = 0 then PC <--PC

Condition Codes:

Description:

The register is decremented. If the contents does not equal 0, twice the offset is subtracted from the PC (now pointing to the following word). The offset is interpreted as a 6-bit positive number. This instruction provides a fast, efficient method of loop control. The assembler syntax is SOB R,A where A is the address to which transfer is to be made if the decremented R is not equal to 0. Note: the SOB instruction cannot be used to transfer control in the forward direction.

6.3.6.5 Traps - Trap instructions provide for calls to emulators, I/O monitors, debugging packages, and user-defined interpreters. A trap is effectively an interrupt generated by software. When a trap occurs, the contents of the current program counter (PC) and processor status word (PS) are pushed onto the processor stack and replaced by the contents of a 2-word trap vector containing a new PC and new PS. The return sequence from a trap involves executing an RTI or RTT instruction, which restores the old PC and old PS by popping them from the stack. Trap instruction vectors are located 6-56

at permanently assigned fixed addresses.

AFTER:

•

EMT									•							
ł	EMULA	TOR T	RAP											1	0400 0-10	04377
	15					r1		08	07			r	r	·		00
	1	0	0	0	1	0 i	0	0					1	1	· · ·	
Operat:	ion:		•	↓ ₽C ₽S	(SP) (SP) <	< < (3)	- PS - PC 0) 2)								м	R-5254
Condit	ion	Cođ	es:	N: Z: V: C:	loa loa loa loa	aded ided ided ided	fro fro fro	om t om t om t	rap rap rap rap	vec vec vec	tor tor tor tor					
Descri	ptic	on:		Al EM fu EM th (P	l op T in form ncti T is e wc S) i	oerat istru iation ion s at ord a s ta	tior ucti on t to t add at a aker	ons ons to t pe p lres addr n fr	des and he e erfo s 30 ess om t	from mag mula rme 30; he	m 10 y be atin d). The the word	400 us Th new ne	0 to ed t outi e tr PC w pr add	o 10 o t ap is oce ires	4377 ransm (e.g. vecto taker ssor s 32.	are hit or fo fro stat
				CA sy fo	UTIC sten r ge	DN: 1 n so: ener:	EMT ftwa al u	is are 1se.	used and	fr	eque ther	efo	y by re r	7 DI Not	GITAI recon	nmend
						PS		PS 1 .]							
			BE	FORE		PC		PC 1				TACK				
						SP		n]	-						

PS (32) PC (30) DATA 1 **PS** 1 SP n--4 PC 1 MR-6255

6--57

104400-104777



BPT																
	BREAK	POINT	TRAP													000003
	15					r		·····	r	.	·	.	,		T	
•	0	0	0	0	0	0	0	0	0	0	0	0	0	0	1	1
Operat	ion:			↓ ₽C ₽S	(SP) (SP) < <	< < (14 (16	- PS - PC 1) 5)									MR 5257
Condit	ion	Cođ	es:	N: Z: V: C:	loa loa loa loa	ded ded ded ded	fro fro fro fro	om ti m ti m ti m ti	rap rap rap	vect vect vect	tor tor tor tor					
Descri	ptic	on:	·	Pe: add The 00 aid by	rfor dres e us 0003 ds. te.)	ms a s of er i in (No	a tr 14 is c pro inf	ap s aut: gran orma	sequ Jsed ione ns r atio	ence to d ag un u n is	e wi cal gain unde s tr	th 1 d st ans	a tr ebug empl hese mitt	ap ging oyin del ed	vec g a ng o bug in	tor ids. code ging the low
IOT																
	INPUT/	OUTPU	TTRAF)												000004



	PC < (20) PS < (22)
Condition Codes:	N: loaded from trap vector Z: loaded from trap vector V: loaded from trap vector C: loaded from trap vector
Description:	Performs a trap sequence with a trap vector address of 20. (No information is transmitted in the low byte.)

service

15	РТ											000002 00
0 0 0 0	0	0	0	0	0	0	0	0	0	0	1	0
Operation:	PC PS	< <	(SP (SP	; 1								MR 5259
Condition Codes:	N: Z: V: C:	load load load load	ded ded ded ded	from from from from	pro pro pro	oces oces oces oces	sor sor sor sor	sta sta sta	ack ack ack ack			
Description:	Us ro fr T	ed to utine om th	e. ne p in t	it f The roce he P	rom PC SSO	an and r st a tr	int PS ack	erru are .] tra	ipt (res [f t]	or 1 tore he F ill	RAP d (TI	ser popp sets ur p

popped) sets the ur prior to executing the next instruction. When executed in supervisor mode, the current and previous mode bits in the restored PS cannot be kernel. When executed in user mode, the current and previous mode bits in the restored PS can only be user. RTI cannot clear PS bit 11 if it was already set.

RTT														
	RETURN FROM	TRAP											000006	
	15												00	
	0 0	0 0	0 0	0	0	0	0	0	0	0	1	1	0	
					_								MA-5260	
Opera	tion:	PC PS	C <	(SP) (SP)	$\mathbf{\hat{x}}$									
Condi	tion Code	S. N	load	leđ f	rom	Dro	oce	ssor	st	ack				
condi	ción cou	Z :	load	led f	rom	pro	oce	ssor	: st	ack				
		V	load	led f	rom	pre	oce	ssor	: st	ack				
		C	: load	led f	rom	pro	oce	ssor	st	ack				
Descr	iption:	Ol ir ti	perati hhibit cace t	on i s a rap.	is t tra I	he : ce : f tl	sam traj	e as pwł new	s RT nere PS	'I e as has	xce RTI th	pt pe	that rmits bit	it a set.

a trap will occur after execution of the first instruction after RTT. When executed in supervisor mode, the current and previous mode bits in the restored PS cannot be kernel. When executed in user mode, the current and previous mode bits in the restored PS can only be user. RTT cannot clear PS bit 11 if it was already set.

6.3.6.6 Miscellaneous Program Control -

MARK

MARK					0064NN
15		06	05		
		0 0	II	NN	<u></u>
Operation:	SP < PC + 2 PC < R5 R5 < (SP)+	x NN	tors		MR 11566
Condition Codes:	N: unaffected 2: unaffected V: unaffected C: unaffected	parame			
Description:	Used as part c convention. MA procedures inv Assembler form	of the st RK facil olved in at is: N	tandard Litates n subro MARK N.	subro the s utine	utine return tack clean-up exit.
Example:	MOV R5,-(SP) MOV P1,-(SP) MOV P2,-(SP) MOV PN,-(SP) MOV =MARKN,-(S MOV SP,R5 JSR PC,SUB At this point	;pl; ;pl; ;pl; ;th ;th ;th ;p) ;pl; ;m ;so ;in ;ju the stac	ace old ace N p e stack here by ace the ARK N o et up a hstruct imp to ck is a	R5 on aramet to be the s instr n the ddress ion subrou s foll	stack ers on used ubroutine uction stack at MARK N tine ows:
· · · · · ·	-	OLD R5 P1 PN			

And the program is at the address SUB which is the beginning of the subroutine.

MR-11569

MARK N

SUB:

;execution of the ;subroutine itself

RTS R5

;the return begins: ;this causes the contents ;of R5 to be placed in the ;PC which then results in ;the execution of the ;instruction MARK N. The ;contents of the old PC ;are placed in R5.

MR 11668

MARK N causes: (1) the stack pointer to be adjusted to point to the old R5 value; (2) the value now in R5 (the old PC) to be placed in the PC; and (3) contents of the old R5 to be popped into R5 thus completing the return from subroutine.

NOTE

If memory management is in use, the stack must be mapped through both I and D space to execute the MARK instruction.

SPL



6-61

Operation:

If MMR3 bit 3 = 1 and current
mode = kernel then
supervisor SP <-- current mode SP
temp<15:4> <-- PS<15:4>
temp<3:0> <-- 0
PS<13:12> <-- PS<15:14>
PS<15:14> <-- 01
PS 4 <-- 0
-(SP) <-- temp
-(SP) <-- PC
-(SP) <-- (dst)
PC <-- (10)
otherwise, traps to 10 in kernel mode.</pre>

Condition Codes:

N: unaffected Z: unaffected V: unaffected C: unaffected

Description:

CSM may be executed in user or supervisor mode, but is an illegal instruction in kernel mode. CSM copies the current stack pointer (SP) to the supervisor mode, switches to supervisor mode, stacks three words on the supervisor stack (the PS with the condition codes cleared, the PC, and the argument word addressed by the operand), and sets the PC to the contents of location 10 (in supervisor space). The called program in supervisor space may return to the calling program by popping the argument word from the stack and executing RTI. On return, the condition codes are determined by the PS word on the stack. Hence, the called program in supervisor space may control the condition code values following return.

6.3.6.7 Reserved Instruction Traps - These are caused by attempts to execute instruction codes reserved for future processor expansion (reserved instructions) or instructions with illegal addressing modes (illegal instructions). Order codes not corresponding to any of the instructions described are considered to be reserved instructions. JMP and JSR with register mode destinations are illegal instructions; they trap to virtual address 4 in kernel data space. Reserved instructions trap to vector address 10 in kernel data space.

6.3.6.8 Trace Trap - Trace trap is enabled by bit 4 of the PS and causes processor traps at the end of instruction execution. The instruction that is executed after the instruction that set the T bit will proceed to completion and then trap through the trap vector at address 14. Note that the trace trap is a system debugging aid and is transparent to the general programmer.

NOTE

Bit 4 of the PS can only be set indirectly by executing a RTI or RTT instruction with the desired PS on the stack.

6.3.6.8.1 Special Cases Of The T Bit - The following are special cases of the T bit.

NOTE

The traced instruction is the instruction after the one that set the T bit.

- An instruction that cleared the T bit -- Upon fetching the traced instruction, an internal flag, the trace flag, was set. The trap will still occur at the end of this instruction's execution. The status word on the stack, however, will have a clear T bit.
- An instruction that set the T bit -- Since the T bit was already set, setting it again has no effect. The trap will occur.
- 3. An instruction that caused an instruction trap -- The instruction trap is performed and the entire routine for the service trap is executed. If the service routine exits with an RTI, or in any other way restores the stacked status word, the T bit is set again, the instruction following the traced instruction is executed, and, unless it is one of the special cases noted previously, a trace trap occurs.
- 4. An instruction that caused a stack overflow -- The instruction completes execution as usual. The stack overflow does not cause a trap. The trace trap vector is loaded into the PC and PS and the old PC and PS are pushed onto the stack. Stack overflow occurs again, and this time the trap is made.
- 5. An interrupt between setting of the T-bit and fetch of the traced instruction -- The entire interrupt service routine is executed and then the T-bit is set again by the exiting RTI. The traced instruction is executed (if there have been no other interrupts) and, unless it is a special case noted above, causes a trace trap.
- 6. Interrupt trap priorities -- See Table 1-8.

6.3.7 Miscellaneous Instructions -

HALT

	HALT															000000 00
	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Opera	atio	n:		P:	(S (S C < S <	P) • P) • 1	< < cest 340	PS PC art	add	ress	5					MR 5261
Condi	itio	n Co	des:	: No	ot	affe	ecte	đ								
Descı	ript	ion:		T) OC I O I C D E S C E S I C F I C C I C C I C C I C I C I C I C	he perr urr (esps it cees aut cees it cees auccees it cees auccees it cees auccees it cees auccees it cees auccees it cees auccees it cees auc	effe atim enti rfac onse onse 7 of uter es f uter tion ster	ect mg ms Ly ng In alor tratth the tr the consort the	of F ode for ker log ap t f th DCJ of th mode and	ALT and ted nel fro P P Chro P C P C C C C C C C C C C C C C C C C	dep the red driv ead ugh rror alt he nto ALT s bi	end ha e C leta witc opt ins t 7	s up hapt ils a ha hation gion LT isole truc of	on opti er on GP on 4 :er is to the the	the on 8 - halt opti DAI code and wher 0 in ruct T. n in rouce CPU	CPU on 3 i e of l se h HA ke ior h us J er	of n UD00) ts LT is ernel ser or

WAIT FO	OR INT	ERRUP	۲												000 001
15															00
0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	' 1
<u> </u>					i							L	L		

Condition Codes: Not affected

Description:

In WAIT, as in all instructions, the PC points to the next instruction following the WAIT instruction. Thus, when an interrupt causes the PC and PS to be pushed onto the processor stack, the address of the next instruction following the WAIT is saved. The exit from the interrupt routine (i.e., execution of an RTI instruction) will cause resumption of the interrupted process at the instruction following the WAIT. If not in kernel mode, WAIT executes as a NOP.

MR-5262

RESET

RESET	EXTER	NAL BL	JS												000 005
															00
0	0	0	0	0	0	0	0	0	0	0	0	0	1	0	' ı .
				1			L'								

MR-5263

000007

MA 7198

Condition Codes: Not affected

Description: The following sequence of events occurs: (1) a GP Write cycle is performed and a GP code of 014 is generated, (2) operation is delayed for 69 microcycles, (3) a GP Write is performed and a GP code of 214 is generated, (4) operation is delayed for 600 microcycles delay. If not in kernel mode, RESET operates as a NOP.

MFPT

MOVE FROM PROCESSOR TYPE WORD

15	_														00
0	0	0	0	0	0	0	0	0	0	0	0	0	1	1	1
			است ال					·		L	.	L	<u> </u>	1	ليسبيه

Operation: R0 <-- 5

Condition Codes: Not affected

Description: The number 5 is placed in R0, indicating to the system software that the processor type is DCJ11.

MTPD/MTPI

	MOVE . MOVE	TO PRE		DATA SINSTR	SPACE	N SPAC	E						■066DD
													00
	0/1	0	0	0	1	1	0	1	1	0		DD	
Operati	lon:			(t) (đ:	emp) st)	> >	- (8 (te	SP)- emp	+)				MR 11571
Conditi	ion C	ode	es:	N: Z:	set set	if if	the	e s(ourc	;e < ;e =	00		

Z: unaffected Description: The instruction pops a word off the current stack determined by PS bits <15:14> and stores that word into an address in the previous space (PS bits <13:12>). The destination address is computed using the current registers and memory

V: cleared

map.

MFPD/MFPI

	MOVE I MOVE I	ROM	PREVIC PREVIC	DUS DA DUS INS	DATA SPACE INSTRUCTION SPACE								■ 065SS	
	0/1	0	0	0	1	1	0	1	0	1			ss	
						L								MR-11570
Operat	ion:			(1	temp (SP)) <-	< ((sr (tem	c) p)					
Condit	N : Z : V : Z :	se se cl un	et i et i .ear naff	lf t lf t ed ect	he he ed	sou	rce rce	< 0 = 0						
Descri	ptio	n:		Pu ac PS us Wh	she dre <13 sing nen	es a ess s:12 s:12 MFP	in ?>. ne c PS i	the Th urr	onto pro e so ent xect	o thevio	ne cu pus s ce ad giste l and	rre pac dre rs bo	ent st e det ss is and m th pr	ack from an ermined by computed emory map. evious mode

as though it were MFPD.

6.3.8 Condition Code Operators -

CLN SEN CLZ SEZ CLV SEV CLC SEC CCC SCC

15										05	04	03	02	01	_
0	່ກ	, 0	, 0	0	, 0	0	0	• 1	0	1	0/1	N	z		

Description:

Set and clear condition code bits. Selectable combinations of these bits may be cleared or set together. Condition code bits corresponding to bits in the condition code operator (bits <3:0> are modified according to the sense of bit 4, the set/clear bit of the operator; i.e., set th bit specified by bit 0, 1, 2, or 3, if bit 4 = 1. Clear corresponding bits if bit 4 = 0.

current mode are user, the instruction functions

Mnemonic	Operation	OP Code
CLC	Clear C	000241
CLV	Clear V	000242
CLZ	Clear Z	000244
CLN	Clear N	000250
SEC	Set C	000261
SEV	Set V	000262
SEZ	Set Z	000264
SEN	Set N	000270
SCC	Set all CCs	000277

CCC	Clear all CCs	000257
	Clear V and C	000243
NOP	No operation	000240

Combinations of the above set or clear operations may be ORed together to form combined instructions.

•

• ·

.

7.1 INTRODUCTION

The DCJ11 executes forty-six floating-point instructions. The floating-point instruction set is compatible with the FP11 instruction set for PDP-11 computers. Both single- and double-precision floating-point capabilities are available with other features, including floating-to-integer and integer-to-floating conversion.

7.2 FLOATING-POINT DATA FORMATS

Mathematically, a floating-point number may be defined as having the form (2 * K) * f, where K is an integer and f is a fraction. For a nonvanishing number, K and f are uniquely determined by imposing the condition $1/2 \le f < 1$. The fractional part (f) of the number is then said to be normalized. For the number 0, f is assigned the value 0, and the value of K is indeterminate.

The floating-point data formats are derived from this mathematical representation for floating-point numbers. Two types of floating-point data are provided. In single-precision, or floating mode, the data is 32 bits long. In double-precision, or double mode, the data is 64 bits long. Sign magnitude notation is used.

7.2.1 Nonvanishing Floating-Point Numbers - The fractional part (f) is assumed normalized, so that its most significant bit must be 1. This 1 is the "hidden" bit: it is not stored explicitly in the data word, but the microcode restores it before carrying out arithmetic operations. The floating and double modes reserve 23 and 55 bits, respectively, for f. These bits, with the hidden bit, imply effective word lengths of 24 bits and 56 bits.

Eight bits are reserved for storage of the exponent K in excess 200 notation (i.e., as K + 200 (octal)), giving a biased exponent. Thus, exponents from -128 to +127 could be represented by 0 to 377 (base 8), or 0 to 255 (base 10). For reasons given below, a biased exponent of 0 (the true exponent of -200 (octal)), is reserved for floating-point 0. Therefore, exponents are restricted to the range -127 to +127 inclusive (-177 to +177 octal) or, in excess 200 notation, 1 to 377.

The remaining bit of the floating-point word is the sign bit. The number is negative if the sign bit is a 1.

7.2.2 Floating-Point Zero - Because of the hidden bit, the fractional part is not available to distinguish between 0 and nonvanishing numbers whose fractional part is exactly 1/2. Therefore, the DCJ11 reserves a biased exponent of 0 for this purpose, and any floating-point number with a biased exponent of 0 either traps or is treated as if it were an exact 0 in arithmetic operations. An exact or "clean" 0 is represented by a word whose bits are all 0s. A "dirty" 0 is a floating-point number with a biased exponent of 0 and a nonzero fractional part. An arithmetic operation for which the resulting true exponent exceeds 277 (octal) is regarded as producing a floating overflow; if the true exponent is less than -177 (octal), the operation is regarded as producing a floating underflow. A biased exponent of 0 can thus arise from arithmetic operations as a special case of overflow (true exponent = -200 octal). (Recall that only eight bits are reserved for the biased exponent.) The fractional part of results obtained from such overflow and underflow is correct.

7.2.3 Undefined Variables - An undefined variable is any bit pattern with a sign bit of 1 and a biased exponent of 0. The term "undefined variable" is used, for historical reasons, to indicate that these bit patterns are not assigned a corresponding floating-point arithmetic value. Note that the undefined variable is frequently referred to as -0 elsewhere in this chapter.

A design objective was to assure that the undefined variable would not be stored as the result of any floating-point operation in a program run with the overflow and underflow interrupts disabled. This is achieved by storing an exact 0 on overflow and underflow, if the corresponding interrupt is disabled. This feature, together with an ability to detect reference to the undefined variable (implemented by the FIUV bit discussed later), is intended to provide the user with a debugging aid: if -0 occurs, it did not result from a previous floating-point arithmetic instruction.

7.2.4 Floating-Point Data - Floating-point data is stored in words of memory as illustrated in Figures 7-1 and 7-2.





The DCJ11 provides for conversion of floating-point to integer format and vice-versa. The processor recognizes single-precision integer (I) and double-precision integer long (L) numbers, which are stored in standard 2's complement form. (See Figure 7-3.)



Figure 7-3 2's Complement Format

7.3 FLOATING-POINT STATUS REGISTER (FPS)

This register provides mode and interrupt control for the currently executing floating-point instruction and also reflects conditions resulting from the execution of the previous instruction. (See Figure 7-4.) In this discussion a set bit = 1 and a reset bit = 0. Three bits of the FPS register control the modes of operation.

- 1. Single/Double -- Floating-point numbers can be either singleor double-precision.
- 2. Long/Short -- Integer numbers can be 16 bits or 32 bits.
- 3. Chop/Round -- The result of a floating-point operation can be either "chopped" or "rounded." The term "chop" is used instead of "truncate" in order to avoid confusion with truncation of series used in approximations for function subroutines.



Figure 7-4 Floating-Point Status Register

The FPS register contains an error flag and four condition codes (5 bits): carry, overflow, zero, and negative, which are analogous to the CPU condition codes.

The DCJ11 recognizes six floating-point exceptions:

- o Detection of the presence of the undefined variable in memory
- o Floating overflow
- o Floating underflow
- o Failure of floating-to-integer conversion
- o Attempt to divide by 0
- o Illegal floating op code

For the first four of these exceptions, bits in the FPS register are available to individually enable and disable interrupts. An interrupt on the occurrence of either of the last two exceptions can be disabled only by setting a bit that disables interrupts on all six of the exceptions, as a group.

Of the 13 FPS bits, 5 are set as part of the output of a floating-point instruction: the error flag and condition codes. Any of the mode and interrupt control bits may be set by the user; the LDFPS instruction is available for this purpose. These thirteen bits are stored in the FPS register as shown in Figure 7-4. The FPS register bits are described in Table 7-1.

Table 7-1 FPS Register Bits

В	it	Name

Description

15 Floating Error (FER) The FER bit is set by the DCJ11 if:

- 1. division by zero occurs.
- 2. an illegal op code occurs.
- 3. any one of the remaining floating point exceptions occurs and the corresponding interrupt is enabled.

Note that the above action is independent of whether the FID bit is set or clear.

Note also that the DCJll never resets the FER bit. Once the FER bit is set by the DCJ11, it can be cleared only by an LDFPS instruction (note the RESET instruction does not clear the FER bit). This means that the FER bit is up-to-date only if the most recent floating-point instruction produced a floating-point exception.

Interrupt Disable If the FID bit is set, all floatingpoint interrupts are disabled. (FID)

NOTE

- 1. The FID bit is primarily a maintenance feature. It should normally be clear. In particular, it must be clear is one wishes to assure that storage of -0 by the DCJ11 is always accompanied by an interrupt.
- 2. Throughout the rest of the chapter assume that the FID bit is clear in all discussions involving overflow, underflow, occurrence of -0, and integer conversion errors.

Reserved for future DIGITAL use.

Reserved for future DIGITAL use.

An interrupt occurs if FIUV is set and a -0 is obtained from memory as an operand of ADD, SUB, MUL, DIV, CMP, MOD, NEG, ABS, TST, or any LOAD instruction. The interrupt occurs before execution on all instructions. When FIUV is reset, -0 can be loaded and used in any floating-point operation. Note that the interupt is not activated by the presence of -0in an AC operand of an arithmetic instruction; in particular, trap on

13

14

12

11 Interrupt on Undefined Variable (FIUV)

7-5

-0 never occurs in mode 0.

A result of -0 will not be stored without the simultaneous occurrence of an interrupt.

When the FIU bit is set, floating underflow will cause an interrupt. The fractional part of the result of the operation causing the interrupt will be correct. The biased exponent will be too large by 400, except for the special case of 0, which is correct. An exception is discussed later in the detailed description of the LDEXP instruction.

When the FIV bit is set, floating overflow will cause an interrupt. The fractional part of the result of the operation causing the overflow will be correct. The biased exponent will be too small by 400.

If the FIV bit is reset and overflow occurs, there is no interrupt. The DCJll returns exact 0.

Special cases of overflow are discussed in the detailed descriptions of the MOD and LDEXP instructions.

When the FIC bit is set and a conversion to integer instruction fails, an interrupt will occur. If the interrupt occurs, the destination is set to 0, and all other registers are left untouched.

If the FIC bit is reset, the result of the operation will be the same as detailed above, but no interrupt will occur.

The conversion instruction fails if it generates an integer with more bits than can fit in the short or long integer word specified by the FL bit.

The FD bit determines the precision that is used for floating-point calculations. When set, doubleprecision is assumed; when reset, single-precision is used.

10 Interrupt on Underflow (FIU)

> Interrupt on Overflow (FIV)

Interrupt on Integer Conversion Error (FIC)

Floating Double-Precision Mode (FD)

7

	•	
6	Floating Long- Integer Mode (FL)	The FL bit is active in conversion between integer and floating-point formats. When set, the integer format assumed is double-precision 2's complement (i.e., 32 bits). When reset, the integer format is assumed to be single-precision 2's complement (i.e., 16 bits).
5	Floating Chop Mode (FT)	When the FT bit is set, the result of any arithmetic operation is chopped (truncated). When reset, the result is rounded.
4		Reserved for future DIGITAL use.
3	Floating Negative (FN)	FN is set if the result of the last floating-point operation was negative; otherwise it is reset.
2	Floating Zero (FZ)	FZ is set if the result of the last floating-point operation was 0; otherwise it is reset.
1	Floating Overflow (FV)	FV is set if the last floating- point operation resulted in an exponent overflow; otherwise it is reset.
0	Floating Carry (FC)	FC is set if the last floating- point operation resulted in a carry of the most significant bit. This can only occur in floating double-to-integer

7.4 FLOATING EXCEPTION CODE AND ADDRESS REGISTERS

One interrupt vector is assigned to take care of all floating-point exceptions (location 244). The six possible errors are coded in the 4-bit floating exception code (FEC) register as follows.

conversions.

- 2 Floating op-code error
- 4 Floating divide by zero
- 6 Floating-to-integer or double-to-integer conversion error
- 8 Floating overflow
- 10 Floating underflow
- 12 Floating undefined variable

The address of the instruction producing the exception is stored in the floating exception address (FEA) register.

The FEC and FEA registers are updated only when one of the

following occurs.

- 1. Division by zero.
- 2. Illegal op code.
- 3. Any of the other four exceptions with the corresponding interrupt enabled.

This implies that only when the FER bit is set are the FEC and FEA registers updated.

NOTE

- 1. If one of the last four exceptions occurs with the corresponding interrupt disabled, the FEC and FEA are not updated.
- 2. If an exception occurs, inhibition of interrupts by the FID bit does not inhibit updating of the FEC and FEA.
- 3. The FEC and FEA are not updated if no exception occurs. This means that the STST (store status) instruction will return current information only if the most recent floating-point instruction produced an exception.
- 4. Unlike the FPS, no instructions are provided for storage into the FEC and FEA registers.

7.5 FLOATING-POINT INSTRUCTION ADDRESSING

Floating-point instructions use the same type of addressing as the central processor instructions. A source or destination operand is specified by designating one of eight addressing modes and one of eight central processor general registers to be used in the specified mode. The modes of addressing are the same as those of the central processor, except in mode 0. In mode 0 the operand is located in the designated floating-point processor accumulator rather than in a central processor general register. The modes of addressing are as follows.

- 0 = Floating-point accumulator
- 1 = Deferred
- 2 = Autoincrement
- 3 = Autoincrement-deferred
- 4 = Autodecrement
- 5 = Autodecrement-deferred
- 6 = Indexed
- 7 = Indexed-deferred

Autoincrement and autodecrement operate on increments and decrements of 4 for F format and 10 (octal) for D format.

In mode 0 users can make use of all six floating-point accumulators (ACO - AC5) as their source or destination. Specifying floating-point accumulators AC6 or AC7 will result in an illegal op code trap. In all other modes, which involve transfer of data to or from memory or the general registers, users are restricted to the first four floating-point accumulators (ACO - AC3). When reading or writing a floating-point number from or to memory, the low memory word contains the most significant word of the floating-point number, and the high memory word the least significant word.

7.6 ACCURACY

General comments on the accuracy of the DCJ11 floating-point are presented here. The descriptions of the instructions individual instructions include the accuracy at which thev operate. An instruction or operation is regarded as "exact" if the result is identical to an infinite precision calculation The a priori accuracy of the involving the same operands. operands is thus ignored. All arithmetic instructions treat an operand whose biased exponent is 0 as an exact 0 (unless FIUV is enabled and the operand is -0, in which case an interrupt occurs). For all arithmetic operations, except DIV, a 0 operand implies that the instruction is exact. The same statement holds for DIV if the 0 operand is the dividend. But if it is the divisor, division is undefined and an interrupt occurs.

For nonvanishing floating-point operands, the fractional part is binary normalized. It contains 24 bits or 56 bits for floating mode and double mode, respectively. For ADD, SUB, MUL, and DIV, two guard bits are necessary and sufficient for the general case to guarantee return of a chopped or rounded result identical to the corresponding infinite precision operation chopped or rounded to the specified word length. Thus, with two guard bits, a chopped result has an error bound of one least significant bit (LSB); a rounded result has an error bound of 1/2 LSB. These error bounds are realized by the DCJ11 of all instructions.

In the rest of this chapter, an arithmetic result is called exact if no nonvanishing bits would be lost by chopping. The first bit lost in chopping is referred to as the "rounding" bit. The value of a rounded result is related to the chopped result as follows.

- 1. If the rounding bit is 1, the rounded result is the chopped result incremented by an LSB.
- 2. If the rounding bit is 0, the rounded and chopped results are identical.
- It follows that:
- If the result is exact: rounded value = chopped value = exact value.
- 2. If the result is not exact, its magnitude is:
 - o always decreased by chopping.
 - o decreased by rounding if the rounding bit is 0.
 - o increased by rounding if the rounding bit is 1.

Occurrence of floating-point overflow and underflow is an error condition: the result of the calculation cannot be correctly stored because the exponent is too large to fit into the eight bits reserved for it. However, the internal hardware has produced the correct answer. For the case of underflow, replacement of the correct answer by 0 is a reasonable resolution of the problem for many applications. This is done by the DCJ11 if the underflow interrupt is disabled. The error incurred by this action is an absolute rather than a relative error; it is bounded (in absolute value) by 2 ** -128. There is no such simple resolution for the case of overflow. The action taken, if the overflow interrupt is disabled, is described under FIV (bit 9) in Table 7-1.

The FIV and FIU bits (of the floating-point status word) provide users with an opportunity to implement their own correction of an overflow or underflow condition. If such a condition occurs and the corresponding interrupt is enabled, the microcode stores the fractional part and the low eight bits of the biased exponent. The interrupt will take place and users can identify the cause by examination of the FV (floating overflow) bit or the FEC (floating exception) register. The reader can readily verify that (for the standard arithmetic operations ADD, SUB, MUL, and DIV) the biased exponent returned by the instruction bears the following relation to the correct exponent.

- 1. On overflow, it is too small by 400 (octal)
- 2. On underflow, if the biased exponent is 0, it is correct. If the biased exponent is not 0, it is too large by 400 (octal).

Thus, with the interrupt enable, enough information is available to determine the correct answer. Users may, for example, rescale their variables (via STEXP and LDEXP) to continue a calculation. Note that the accuracy of the fractional part is unaffected by the occurrence of underflow or overflow.

7.7 FLOATING-POINT INSTRUCTIONS

Each instruction that references a floating-point number can operate on either single- or double-precision numbers, depending on the state of the FD mode bit. Similarly, there is a mode bit FL that determines whether a 32-bit integer (FL = 1) or a 16-bit integer (FL = 0) is used in conversion between integer and floating-point representations. FSRC and FDST operands use floating-point addressing modes (see Figure 7-5); SRC and DST operands use CPU addressing modes.

DOUBL	E·OPI	ERAND ADDRES	SING		· .			-	
r15		12	<u>11</u>		08	07 06	05		00
		oc		FOC		AC		FSRC, FDST, SRC, DST	
SINGLE	OPE	RAND ADDRES	SING						
15		12	11			06	05		00
		ос		F	oc	i		FSRC, FDST, SRC, DST	
OC = OP FOC = F AC = FL FSRC AN SRC AN	COD LOA OAT ND F D DS	E = 17 TING OPCODE ING POINT ACC DST USE FPP AU T USE CPU ADD	UMULAT DDRESSING RESSING	OR (ACO AG NG MODES MODES	^{c3)}	Point	Add	ressing Mod	MR 3608
Terms Use	eđ	in Ins	truc	tion	Def	initic	ons		
oc	=	opcode	= 1	7					
FOC	=	floati	ng o	pcode	5				
AC	=	conten field o	ts o f in	of acc struc	cumu tio	lator, n	as	specified	by AC
fsrc	=	addres	s of	floa	tin	g-poir	it s	ource opera	nđ
fdst	H	addres	s of	floa	tin	g-poin	it đ	estination	operand
f	H	fracti	on						
XL	Ξ	larges	t fr	actič	on t	hat ca	n b	e represent	ed:
		1 - 2 1 - 2	** (** (-24), -56),	FD FD	= 0; = 1;	sin dou	gle-precisi ble-precisi	on
XLL	=	smalle	st n	umber	th	at is	not	identicall	y zero =
		2 ** (-128)					
XUL	=	larges	t nu	mber	tha	t can	be	represented	=
		2 ** (127)	* XL	1				
JL	=	larges	t in	teger	th	at can	be	represente	d:

7-11

2 ** (15) - 1; FL = 0; short integer 2 ** (31) - 1; FL = 1; long integer ABS (address) = absolute value of (address) EXP (address) = biased exponent of (address) .LT. = "less than" .LE. = "less than or equal to" .GT. = "greater than" .GE. = "greater than or equal to" LSB = least significant bit Boolean Symbols

 \wedge = AND \checkmark = inclusive OR \overleftrightarrow = exclusive OR \sim = NOT



Format:	ABSF FDST
Operation:	If (FDST) < 0, (FDST) <(FDST).
	If $EXP(FDST) = 0$, (FDST) < exact 0.
	For all other cases, (FDST) < (FDST).
Condition Codes:	FC < 0 FV < 0 F2 < 1 if (FDST) = 0, else FZ < 0 FN < 0
Description:	Set the contents of FDST to its absolute value.
Interrupts:	If FIUV is enabled, trap on -0 occurs before execution. Overflow and underflow cannot occur.
Accuracy:	These instructions are exact.

ADDF/ADDD

ADD	FLO	ATIN	IG/DO	UBLE									172(AC)FSRC
15)			12	11			08	07	06	05		00
1		1	1	1	0	1	0	0	A			FSRC	
													MR 11468

ADDF FSRC, AC Format: **Operation:** Let SUM = (AC) + (FSRC)If underflow occurs and FIU is not enabled, AC <-- exact 0. If overflow occurs and FIV is not enabled, AC <-- exact 0. For all others cases, AC <-- SUM. Condition Codes: FC < -- 0FV <-- 1 if overflow occurs, else FV <-- 0 FZ <-- 1 if (AC) = 0, else FZ <-- 0 $FN \iff 1$ if (AC) < 0, else $FN \iff 0$ Description: Add the contents of FSRC to the contents of AC. The addition is carried out in single- or double-precision and is rounded or chopped in accordance with the values of the FD and FT bits in the FPS register. The result is stored in AC except for: 1. Overflow with interrupt disabled. 2. Underflow with interrupt disabled. For these exceptional cases, an exact 0 is stored in AC. If FIUV is enabled, trap on -0 in FSRC occurs Interrupts: before execution. If overflow or underflow occurs, and if the corresponding interrupt is enabled, the trap occurs with the faulty result in AC. The fractional parts are correctly stored. The exponent part is too small by 400 for overflow. It is too large by 400 for underflow, except for the special case of 0, which is correct. Errors due to overflow and underflow are Accuracy: described above. If neither occurs, then: for oppositely signed operands with exponent difference of 0 or 1, the answer returned is exact if a loss of significance of one or more bits can occur. Note that these are the only cases for which loss of significance of more

than one bit can occur. For all other cases the
result is inexact with error bounds of:

- 1. LSB in chopping mode with either single- or double-precision.
- 2. 1/2 LSB in rounding mode with either singleor double-precision.

Special Comment: The undefined variable -0 can occur only in conjunction with overflow or underflow. It will be stored in AC only if the corresponding interrupt is enabled.

CFCC

15			12	11											00
	T	1	Т		Т	1	1	T			r - 1				1
١	1	1	1	0	0	0	0	0	0	0	0	0	0	0	0

Forma	+.	CRCC
rorma	τ:	Uruu

- Operation: C <-- FC V <-- FV Z <-- FZ N <-- FN
- Description: Copy the floating-point condition codes into the CPU's condition codes.

CLRF/CLRD

15			12	11			-		06	05		
1	1	1	1	0	0	0	1	0	0		 EDST	- 1

Format:	CLRF	FDST
---------	------	------

Operation: (FDST) <-- exact 0

Condition Codes: FC <-- 0 FV <-- 0 FZ <-- 1 FN <-- 0

Description: Set FDST to 0. Set FZ condition code and clear other condition code bits.

- Interrupts: No interrupts will occur. Overflow and underflow cannot occur.
- Accuracy: These instructions are exact.

CMPF/CMPD

COM	PARE FLO	ATING/	DOUB	_E			•••							
	1 1	1	1	0	1	1	1	AC	06 0	5	FSI	3C	00	
													MR-11471	
Format:			CMP	PF F	SRC	C, AC	•							
Operation	:		(FS	RC)	-	(AC	:)							
Condition	s:	FC FV FZ FN	< < <	0 0 1 1	if if	(FS (FS	RC) = RC) -	= 0, < 0,	else else	FZ FN	< <	0 0		
Descriptio	on:		Com acc con lef	npar sumu ndit t u	e t lat ion nch	che cor. co nang	con S des ed	tent et t . F exce	s of he ag SRC a pt a	FSRC pprop and t s not	wi ria he ed	th th te fi accur below	ne loatin nulato W.	g-point r are
Interrupts	5:		If exe	FIU	V i ion	s e	nab	led,	tra	p on	-0 0	occui	rs bef	ore
Accuracy:			The	ese	ins	stru	cti	ons	are	exact	•			
Special Co	ommen	t:	An tre whe sto	ope ate re ore	ran da bot an	d t s i h o exa	hat f i per ct	has t we: ands 0 in	a bi re an are AC.	iased n exa 0, t	ex] ct (he I	poner). 1 DCJ11	nt of In thi I will	0 is s case,

DIVF/DIVD

	DIVIDE	FLOA	TING/	DOUBLE	E								174(AC+4)FSRC
	15		_	12	11			08	07	06	05		00
	1	1 	1	1	1	0	0	1	A			FSRC	
													MR 11472
Format:				DI	VF	FSR	с,ас	2					
Operati	on:			If in	EX: str	P(F: uct:	SRC) ion	= is	0, abox	(AC) rte) < d.	(AC) ar	nd the
				If	EX	P (A	2) =	= 0,	(A(C) ·	< ex	act 0.	
				Fo	r a	11 0	othe	er c	ase	s, 1	let QU	OT = (A	AC)/(FSRC).

If underflow occurs and FIU is not enabled, AC <-- exact 0.

If overflow occurs and FIV is not enabled, AC <-- exact 0.

For all others cases, AC <-- QUOT.

Condition Codes:

FC <-- 0 FV <-- 1 if overflow occurs, else FV <-- 0 FZ <-- 1 if (AC) = 0, else FZ <-- 0 FN <-- 1 if (AC) < 0, else FN <-- 0

Description:

If either operand has a biased exponent of 0, it is treated as an exact 0. For FSRC this would imply division by 0; in this case the instruction is aborted, the FEC register is set to 4, and an interrupt occurs. Otherwise, the quotient is developed to single- or double-precision with two guard bits for correct rounding. The quotient is rounded or chopped in accordance with the values of the FD and FT bits in the FPS register. The result is stored in the AC except for:

Overflow with interrupt disabled.
 Underflow with interrupt disabled.

For these exceptional cases, an exact 0 is stored in AC.

Interrupts: If FIUV is enabled, trap on -0 in FSRC occurs before execution. If (FSRC) = 0, interrupt traps on an attempt to divide by 0. If overflow or underflow occurs, and if the corresponding interrupt is enabled, the trap occurs with the faulty result in AC. The fractional parts are correctly stored. The exponent part is too small by 400 for overflow. It is too large by 400 for underflow, except for the special case of 0, which is correct.

Accuracy: Errors due to overflow and underflow are described above. If none of these occurs, the error in the quotient will be bounded by 1 LSB in chopping mode and by 1/2 LSB in rounding mode.

Special Comment: The undefined variable -0 can occur only in conjunction with overflow or underflow. It will be stored in AC only if the corresponding interrupt is enabled.

LDCDF/LDCFD



LDCDF FSRC,AC Format: If EXP(FSRC) = 0, AC <-- exact 0. **Operation:** If FD = 1, FT = 0, FIV = 0 and rounding causes overflow, AC <-- exact 0. In all other cases, AC <-- Cxy(FSRC), where Cxy specifies conversion from floating mode x to floating mode y. x = D, y = F if FD = 0 (single) LDCDF y = F, y = D if FD = 1 (double) LDCFD Condition Codes: FC <-- 0 FV <-- 1 if conversion produces overflow, else FV < -- 0FZ < --1 if (AC) = 0, else FZ < --0 $FN \leftarrow 1$ if $(AC) \leftarrow 0$, else $FN \leftarrow 0$ If the current mode is floating mode (FD = 0), **Description:** the source is assumed to be a double-precision number and is converted to single-precision. If the floating chop bit (FT) is set, the number is chopped; otherwise, the number is rounded. If the current mode is double mode (FD = 1), the source is assumed to be a single-precision number and is loaded left-justified in AC. The lower half of AC is cleared. If FIUV is enabled, trap on -0 occurs before Interrupts: execution. Overflow cannot occur for LDCFD. A trap occurs if FIV is enabled, and if rounding with LDCDF causes overflow. AC <-- overflowed This result must be +0 or -0. Underflow result. cannot occur.

Accuracy: LDCFD is an exact instruction. Except for overflow, described above, LDCDF incurs an error bounded by 1 LSB in chopping mode and by 1/2 LSB in rounding mode.

LDCIF/LDCID/LDCLF/LDCLD

LOAD AND CONVERT INTEGER OR LONG INTEGER TO FLOATING OR DOUBLE-PRECISION 177(AC)SRC 12 80 07 06 05 15 00 1 1 1 1 1 ٥ AC SRC MB-11474

Format:

LDCIF SRC,AC

Operation:

AC <-- Cjx(SRC), where Cjx specifies conversion from integer mode j to floating mode y.

	j = I if $FL = 0$, $j = L$ if $FL = 1x = F if FD = 0, x = D if FD = 1$
Condition Codes:	FC < 0 FV < 0 FZ < 1 if (AC) = 0, else FZ < 0 FN < 1 if (AC) < 0, else FN < 0
Description:	Conversion is performed on the contents of SRC from a 2's complement integer with precision j to a floating-point number of precision x. Note that j and x are determined by the state of the mode bits FL and FD.
	If a 32-bit integer is specified (L mode) and (SRC) has an addressing mode of 0 or immediate addressing mode is specified, the 16 bits of the source register are left-justified and the remaining 16 bits loaded with 0s before conversion.
	In the case of LDCLF, the fractional part of the floating-point representation is chopped or rounded to 24 bits for FT = 1 or 0, respectively.
Interrupts:	None; SRC is not floating-point, so trap on -0 cannot occur.
Accuracy:	LDCIF, LDCID, and LDCLD are exact instructions. The error incurred by LDCLF is bounded by 1 LSB

LDCIF, LDCID, and LDCLD are exact instructions. The error incurred by LDCLF is bounded by 1 LSB in chopping mode and by 1/2 LSB in rounding mode.

LDEXP



Format: LDEXP SRC,AR

Operation:

NOTE: 177 and 200, appearing below, are octal numbers.

If -200 < SRC < 200, EXP(AC) <--SRC + 200 and the rest of AC is unchanged.

If (SRC) > 177 and FIV is enabled, EXP(AC) <-- [(SRC) + 200]<7:0>.

If (SRC) > 177 and FIV is disabled, AC <-- exact 0.

If (SRC) < -177 and FIU is enabled, EXP(AC) <-- [(SRC) + 200]<7:0>.

If (SRC) < -177 and FIU is disabled, AC <-- exact 0.

Condition Codes: FC <-- 0 FV <-- 1 if (SRC) > 177, else FV <-- 0 FZ <-- 1 if (AC) = 0, else FZ <-- 0 FN <-- 1 if (AC) < 0, else FN <-- 0

Description: Change AC so that its unbiased exponent = (SRC). That is, convert (SRC) from 2's complement to excess 200 notation and insert it into the EXP field of AC. This is a meaningful operation only if ABS(SRC) LE 177.

If SRC > 177, the result is treated as overflow. If SRC < -177, the result is treated as underflow.

Interrupts: No trap on -0 in AC occurs, even if FIUV is enabled. If SRC > 177 and FIV is enabled, trap on overflow will occur. If SRC < -177 and FIU is enabled, trap on underflow will occur.

Accuracy: Errors due to overflow and underflow are described above. If EXP(AC) = 0 and (SRC) = -200, AC changes from a floating-point number treated as 0 by all floating arithmetic operations to a non-0 number. This happens because the insertion of the "hidden" bit in the microcode implementation of arithmetic instructions is triggered by a nonvanishing value of EXP.

For all other cases, LDEXP implements exactly
the transformation of a floating-point number (2
** K) * f into (2 ** (SRC)) * f where 1/2 .LE.
ABS(f) .LT. 1.

LDF/LDD

i



- Format: LDF FSRC, AC
- Operation: AC <-- (FSRC)
- Condition Codes: FC <-- 0 FV <-- 0

7-19

	FZ <1 if (AC) = 0, else $FZ <0FN <1 if (AC) < 0, else FN <0$									
Description:	Load single- or double-precision number into AC.									
Interrupts:	If FIUV is enabled, trap on -0 occurs before AC is loaded. Overflow and underflow cannot occur.									
Accuracy:	These instructions are exact.									
Special Comment:	These instructions permit use of -0 in a subsequent floating-point instruction if FIUV is not enabled and (FSRC) = -0 .									

LDFPS



Operation:	FPS < (SRC)
Description:	Load floating-point status register from SRC.
Special Comment:	Users are cautioned not to use bits 13, 12, and 4 for their own purposes, since these bits are not recoverable by the STFPS instruction.

MODF/MODD



Format:

MODF FSRC,AC

Description and Operation:

This instruction generates the product of its two floating-point operands, separates the product into integer and fractional parts, and then stores one or both parts as floating-point numbers.

and

Let PROD = (AC) * (FSRC) so that in Floating-point: ABS(PROD) = (2 * K) * f, where 1/2 .LE. f .LT. 1, and EXP(PROD) = (200 + K)

Fixed-point binary: PROD = N + g, where

N = INT(PROD) = integer part of PROD, and

g = PROD - INT(PROD) = fractional part of PROD with 0 .LE. g .LT. l.

Both N and g have the same sign as PROD. They are returned as follows:

If AC is an even-numbered accumulator (0 or 2), N is stored in AC+1 (1 or 3), and g is stored in AC.

If AC is an odd-numbered accumulator, N is not stored and g is stored in AC.

The two statements above can be combined as follows:

N is returned to AC \bigvee 1 and g is returned to AC.

Five special cases occur, as indicated in the following formal description with L = 24 for floating mode and L = 56 for double mode.

1. If PROD overflows and FIV is enabled, AC \bigvee 1 <-- N, chopped to L bits, AC <-- exact 0.

Note that EXP(N) is too small by 400 and that -0 can be stored in AC \bigvee 1.

If FIV is not enabled, AC \bigvee 1 <-- exact 0, AC <-- exact 0, and -0 will never be stored.

2. If 2 ** L .LE. ABS(PROD) and no overflow, AC ∨ 1 <-- N, chopped to L bits, AC <-- exact 0.

The sign and EXP of N are correct, but low-order bit information is lost.

3. If 1 .LE. ABS(PROD) .LT. 2 ** L, AC ∨ 1 <--N, AC <-- g.</pre>

The integer part N is exact. The fractional part g is normalized, and chopped or rounded in accordance with FT. Rounding may cause a return of + unity for the fractional part. For L = 24, the error in g is bounded by 1 LSB in chopping mode and by 1/2 LSB in rounding mode. For L = 56, the error in g increases from the above limits as ABS(N) increases above 8 because only 59 bits of PROD are generated.

If 2 ** p .LE. ABS(N) .LT. 2 ** (p + 1), with p > 2, the low order p - 2 bits of g may be in error.

4. If ABS(PROD) .LT. 1 and no underflow, AC \bigvee 1 <-- exact 0 and AC <-- g.

There is no error in the integer part. The error in the fractional part is bounded by 1 LSB in chopping mode and 1/2 LSB in rounding mode. Rounding may cause a return of + unity for the fractional part.

5. If PROD underflows and FIU is enabled, AC \bigvee 1 <-- exact 0 and AC <-- g.

Errors are as in case 4, except that EXP(AC) will be too large by 4008 (if EXP = 0, it is correct). Interrupt will occur and -0 can be stored in AC.

If FIU is not enabled, AC \bigvee 1 <-- exact 0 and AC <-- exact 0.

For this case the error in the fractional part is less than 2 ** (-128).

Condition Codes: FC <-- 0 FV <-- 1 if PROD overflows, else FV <-- 0 FZ <-- 1 if (AC) = 0, else FZ <-- 0 FN <-- 1 if (AC) < 0, else FN <-- 0

Interrupts: If FIUV is enabled, trap on -0 in FSRC occurs before execution. Overflow and underflow are discussed above.

Accuracy: Discussed above.

Applications: 1. Binary-to-decimal conversion of a proper fraction. The following algorithm, using MOD, will generate decimal digits D(1), D(2) . . . from left to right.

This algorithm is exact. It is case 3 in the description because the number of nonvanishing bits in the fractional part of

PROD never exceeds L, and hence neither chopping nor rounding can introduce error.

2. To reduce the argument of a trigonometric function.

ARG * 2/PI = N + g. The low two bits of N identify the quadrant, and g is the argument reduced to the first quadrant. The accuracy of N + g is limited to L bits because of the factor 2/PI. The accuracy of the reduced argument thus depends on the size of N.

3. To evaluate the exponential function e ** x, obtain $x * (\log e \text{ base } 2) = N + g$, then e ** x = (2 ** N) * (e ** (g * ln 2)).

The reduced argument is g * ln2 < l and the factor 2 ** N is an exact power of 2, which may be scaled in at the end via STEXP, ADD N to EXP and LDEXP. The accuracy of N + g is limited to L bits because of the factor (log e base 2). The accuracy of the reduced argument thus depends on the size of N.

MULF/MULD

15			12	11			08	07	06	05		00
1										T		T
1	۱	1	1	0	0	1	0	A	С		FSRC	
6					1)	L		,		I I	1

Format:

MULF FSRC, AC

Operation:

Let PROD = (AC) * (FSRC)

If underflow occurs and FIU is not enabled, AC <-- exact 0.

If overflow occurs and FIV is not enabled, AC <-- exact 0.

For all others cases, AC <-- PROD.

- Condition Codes: FC <-- 0 FV <-- 1 if overflow occurs, else FV <-- 0 FZ <-- 1 if (AC) = 0, else FZ <-- 0 FN <-- 1 if (AC) < 0, else FN <-- 0
- Description: If the biased exponent of either operand is 0, (AC) <-- exact 0. For all other cases PROD is generated to 48 bits for floating mode and 59 bits for double mode. The product is rounded or chopped for FT = 0 or 1, respectively, and is stored in AC except for:

- 1. Overflow with interrupt disabled.
- 2. Underflow with interrupt disabled.

For these exceptional cases, an exact 0 is stored in AC.

- Interrupts: If FIUV is enabled, trap on -0 in FSRC occurs before execution. If overflow or underflow occurs, and if the corresponding interrupt is enabled, the trap occurs with the faulty result in AC. The fractional parts are correctly stored. The exponent part is too small by 400 for overflow. It is too large by 400 for underflow, except for the special case of 0, which is correct.
- Accuracy: Errors due to overflow and underflow are described above. If neither occurs, the error incurred is bounded by 1 LSB in chopping mode and 1/2 LSB in rounding mode.
- Special Comment: The undefined variable -0 can occur only in conjunction with overflow or underflow. It will be stored in AC only if the corresponding interrupt is enabled.

NEGF/NEGD



Format:	NEGF FDST
Operation:	(FDST) < (FDST) if (FDST) = 0, else (FDST) < exact 0
Condition Codes:	FC < 0 FV < 0 FZ < 1 if (FDST) = 0, else FZ < 0 FN < 1 if (FDST) < 0, else FN < 0
Description:	Negate the single- or double-precision number; store result in same location (FDST).
Interrupts:	If FIUV is enabled, trap on -0 occurs before execution. Overflow and underflow cannot occur.
Accuracy:	These instructions are exact.

SETD

	SET FL	ITAO.	NG DOU	BLE M	DDE											170011
	15			12	11		· · · ·			·						00
	1	1		1	0	0	0	0	0	0	0	0	' 1 '	0	0	1
		,											<u> </u>			MR-11481
Format:				SETD												
Operati	on:			FD < 1												
Descrip	tio	n:		Se	t t	he I	DCJ I	.l i	n đ	oub:	le p	prec	cisi	on	mod	е.

.

SETF

•

į

i

_											11	12			15
ł	0	0	0	0	0	0	0	0	0	0	0	' 1	1	1	1

Format:	SETF
Operation:	FD < 0
Description:	Set the DCJ11 in single-precision mode.

SETI



Operation:	FL < 0
Description:	Set the DCJll for short-integer data.

SETL



Format:

Operation: FL <-- 1

SETL

.

-

Description: Set the DCJ11 for long-integer data.

STCFD/STCDF

	STORE	AND (ROM D	CONVE	RT FRO	M FLO	ATING- G	TO DO	UBLE					176(AC)FDST	
	15			12	11			08	07 06	05				00	
	1	۱	1 1	1	1	1	0	0	AC		· · ·	FDST	· · · · ·		
													h	AR-11485	
Format:	×			STC	FD	AC , 1	FDSI	ſ							
Operatio	n:			If	(AC) =	0,	(FC)ST) <-	- ex	act	0.			
				If ove	FD rfl	= 1 ow,	, F1 (F1	r = Ost)	0, FIV < e	y = 0 exact	anđ 0.	rou	ndin	g cau	ses
				In Cxy flc	In all other cases, (FDST) < Cxy(AC), where Cxy specifies conversion from floating mode x to floating mode y.										
					x = x =	F, D,	y = y =	= D = F	if FD if FD	= 0 = 1	(sin (dou)	gle) ble)	STC	FD DF	
Conditio	s:	FC FV FV FZ FN	< < <	0 1 0 1	if (if (CONV (AC) (AC)	version = 0, < 0,	else else	duce: FZ FN	s ov < <	erflo 0 0	ow, ei	Lse		
Descript	ion	:		If acc the	the umu lo	cu: lato wer	rrer or i hal	ntπ is s Lf i	ode is tored s clea	sind left red.	gle-j jus	prec tifi	ision ed in	n, the n FDS:	e F and
				If con sin on	the ten gle the	cu: ts c -pro sta	rren of t ecis ate	nt m the sion of	node is accumu , chop FT, an	dou lato ped o d sto	ble- rar orrc ored	prec e co ound in 1	isio nver ed d FDST	n, the ted to epend: •	e D Lng
Interrup	ts:			Tra ena Und for	p o ble erf ST	n - (d be low CFD) wi ecau car	ill Ise Inot	not oc FSRC i occur	cur (s an . Ov	even accu erflo	if 1 umula ow ca	FIUV ator anno	is t occi	ır
				A t wit ove	rap h S rfl	oco TCDI oweo	curs F ca d re	s if ause esul	FIV i s over t. Th	s ena flow is mu	able . (1 ust	d, an FDST be +	nd i:) < 0 or	f rour - -0.	nding
Accuracy				STC ove bou in	FD rfl nde rou	is ow, d by ndir	an e des y l ng m	exac scri LSB node	t inst bed ab in ch	ruct ove, oppin	ion. STCI ng ma	Ex DF i ode	cept ncur and 1	for s an o by 1/3	error 2 LSB

STCFI/STCFL/STCDI/STCDL

	STORE TO INT	AND C	ONVE	RT FRO NG INTE	M FLO	ATING	ORDO	UBLE		175(AC+4)DST					
	15	.		12	11			80	07 06	05			т т	_00	
	1	1	1	1	1	0	1	- ـــــــــــــــــــــــــــــــــ	AC		l	DST			
													M	R-11486	
Format:				STC	FI	AC,I	DST								
Operatio	on:			(DS els fro	T) e (m f	< DST loa	Cx;) <- ting	j (AC C g mc) if -), wher de x t	JL - e Cjx o int	l < spe eger	Cxj(cifi mod	AC) es c e j.	< JL onve	+ 1, rsion
					j = x =	I : F :	if H if H	7L = 7D =	0, j 0, x	= L'i = D i	f FL f FD	= 1 = 1			
				JL	is	the	la	ges	t inte	ger.					
					2 ** 15 - 1 for FL = 0 2 ** 32 - 1 for FL = 1										
Conditio	on C	ođe	s:	C, C, V, Z, N,	FC FC FV FZ FN	< < < <	0 : 1 0 1 : 1 :	if - if (if (JL - 1 DST) = DST) <	< Cx 0, e 0, e	j(AC lse lse	2) < 2, F N, F	JL + Z <- N <-	1, - 0 - 0	else
Descript	tion	:		Conversion is performed from a floating-point representation of the data in the accumulator to an integer representation.										nt or to	
				If the conversion is to a 32-bit word (L mode), and an addressing mode of 0 or immediate addressing mode is specified, only the most significant 16 bits are stored in the destination register.											
				If sel of	the ect the	ope ed l DST	erat by H Fan	Tion TL, re s	fC is ou FC is set to	t of set t 0.	the o 1	inte and	ger the	rang cont	e ents
				Num (ra Thi cle	ber the s i are	s to r th s th d in	o be nan rue n ti	e co rou eve ne F	onverte Inded) En when PS reg	d are befor the ister	e alw e th chop	ays ey a mod	chop re c e bi	ped onve t FT	rted. is
Interrug	ots:			The ena not con	se ble in ver	ins d, d men sion	truc beca nory n fa	ctio ause 7. I ailu	ns do the - f FIC re wil	not i 0, if is en 1 occ	nter pre able ur.	rupt sent d, t	if , is rap	FIUV in on	is AC,
Accuracy	<u>/</u> :			The flo int The ran	se ati ege y a ge	ins ng-p r ma re e imp]	truc poir pst exac Lied	ctio nt c clo ct i by	ns sto perand sely a f the FL.	re th , whi pprox integ	e in ch m imat er p	tege ay n ing art	r pa ot b the is w	oper oper vithi	f the e and. n the

7-27

STEXP

	STOR	EEXPOR	VENT											175	(AC)DST	
	15			12	11			08	07	06	05				00	
	,	1	1 1	1	1	0	1	0	A	с		, ,	ost L			
															MR-11487	
Format	:			ST	EXP	AC	,DS	r								
Operati	ion:			(D	ST)	<	- E)	KP (A	AC)	- 2	00					
Conditi	ion	Cođ	es:	C, V, Z, N,	FC FV F2 FN	< < <	- 0 - 0 - 1 - 1	if if	(DS (DS	T) T)	= 0, < 0,	else else	Z, N,	FZ FN	< <	0 0
Descrip	ptic	n:		Co to	nve 21	rt A s co	AC ¹ s omp]	s ex Leme	kpon ent	ent and	fro sto	m exc re th	ess e r	20(esul) no ⁴ lt in	tation n DST.
Interru	upts	:		Th an	is d u	ins nde	truc flc	ctic bw c	on w cann	ill ot	not occu	trap r.	on	-0.	. Ove	erflow
Accurac	cy:			Th	is .	ins	truc	ctic	on i	s e	xact	•				

STF/STD



Format:	STF AC, FDST
Operation:	(FDST) < AC
Condition Codes:	FC < FC FV < FV FZ < FZ FN < FN
Description:	Store single- or double-precision number from AC.
Interrupts:	These instructions do not interrupt if FIUV is enabled, because the -0 , if present, is in AC, not in memory. Overflow and underflow cannot occur.
Accuracy:	These instructions are exact.
Special Comment:	These instructions permit storage of a -0 in

memory from AC. There are two conditions in which -0 can be stored in an AC of the DCJ11. One occurs when underflow or overflow is present and the corresponding interrupt is enabled. A second occurs when an LDF or LDD instruction is executed and the FIUV bit is disabled.

STFPS



Format:	STFPS DST
Operation:	(DST) < FPS
Description:	Store the floating-point status register in DST.
Special Comment:	Bits 13, 12, and 4 are loaded with 0. All other bits are the corresponding bits in the FPS.

STST



Format: STST DST

Operation: (DST) <-- FEC (DST + 2) <-- FEA

Description: Store the FEC and FEA in DST and DST+2. Note the following.

- If the destination mode specifies a general register or immediate addressing, only the FEC is saved.
- 2. The information in these registers is current only if the most recently executed floating-point instruction caused a floating-point exception.

SUBF/SUBD

SUBTRACT FLOATING/DOUBLE

173(AC)FSRC

MR 11491

15	5			12	11			08	07	06	05				00
 ,	-1		1	1		1	1	0	Á	.c			FSRC	1	
		, 						J				L			

Format: SUBF FSRC,AC Let DIFF = (AC) - (FSRC)Operation: If underflow occurs and FIU is not enabled, AC <-- exact 0. If overflow occurs and FIV is not enabled, AC <-- exact 0. For all others cases, AC <-- DIFF. Condition Codes: FC < -- 0FV <-- 1 if overflow occurs, else FV <-- 0 FZ < -- 1 if (AC) = 0, else FZ < -- 0 $FN \iff 1$ if (AC) < 0, else $FN \iff 0$ Subtract the contents of FSRC from the contents Description: of AC. The subtraction is carried out in single- or double-precision and is rounded or chopped in accordance with the values of the FD and FT bits in the FPS register. The result is stored in AC except for: 1. Overflow with interrupt disabled. 2. Underflow with interrupt disabled. For these exceptional cases, an exact 0 is stored in AC. If FIUV is enabled, trap on -0 in FSRC occurs Interrupts: before execution. If overflow or underflow occurs, and if the corresponding interrupt is enabled, the trap occurs with the faulty result in AC. The fractional parts are correctly stored. The exponent part is too small by 400 for overflow. It is too large by 400 for underflow, except for the special case of 0, which is correct. Errors due to overflow and underflow are Accuracy: described above. If neither occurs: for like-signed operands with exponent difference of 0 or 1, the answer returned is exact if a loss of significance of one or more bits can occur. Note that these are the only cases for which loss of significance of more than one bit can For all other cases the result is occur. inexact with error bounds of: 1. LSB in chopping mode with either single- or double-precision.

2. 1/2 LSB in rounding mode with either singleor double-precision.

Special Comment: The undefined variable -0 can occur only in conjunction with overflow or underflow. It will be stored in AC only if the corresponding interrupt is enabled.

.

TSTF/TSTD

														•			
	TEST F	LOATI	NG/DOU	JBLE											1705	FDST	
	15			12	11		· · · · · · · · · · · · · · · · · · ·		·	06	05		·			00	
	1	' 1	1	1	0	, 1	' 0	, 1 		1			F(DST			
															M	8 11492	×
Format	:			TS	STF	FDS	T										
Operat	ion:			(F	DST	')											
Condit:	FC FV FZ FN	<pre>< - </pre>	- 0 - 0 - 1 - 1	if if	(F) (F)	DST) DST)	= <	0, 0,	els els	e Fi e Fi	s <- 1 <-	- 0 - 0					
Descrip	ptio	n :		Se to	t t th	he e c	flo ont	ati ent	ng-p s of	oin FD	t c ST.	cond	itio	on c	odes	acc	ording
Interru	upts	•		If ex	FI	UV tio	is n.	set Ove	, tr rflo	ap w a	on Ind	-0 d unđ	occu erf]	urs Low	befo cann	ore ot o	ccur.
Accurac	cy:			Th	ese	in	str	uct	ions	ar	e e	exact	t.				

• • -

8.1 INTRODUCTION

This chapter covers topics related to the interfacing of external logic to the DCJ11.

8.2 GENERAL-PURPOSE (GP) CODES

An important means of communicating with external logic is through the use of GP Reads and Writes (see Chapter 3 - Bus Cycles). GP Reads and Writes are associated with codes that specify the function performed during the GP Read or Write cycle. External logic interprets these codes to implement system functions. Table 8-1 summarizes the GP codes.

Table 8-1 GP Codes and Functions

GP Code (octal)	GP Read or Write	Function
000	Read	Reads the power-up mode, HALT option, FPA option, POK, and boot address.
001	Read	Reads FPA data (if FPA exists)
002	Read	Reads the power-up mode, HALT option, FPA option, POK, and boot address, and (if an FPA exists) clears the FPA's FPS.
003	Read	Acknowledges FPE and reads the FEC (floating exception code) register
003	Write	Writes FPA 16-bit data (if FPA exists)
014	Write	Asserts bus reset signal
034	Write	Signals exit from console ODT
040	Write	Reserved for future use
100	Write	Acknowledges EVENT
140	Write	Acknowledges power fail
214	Write	Negates bus reset signal
220	Write	Microdiagnostic test 1 passed
224	Write	Microdiagnostic test 2 passed
230	Write	Microdiagnostic test 3 passed
234	Write	Signals entry into console ODT

Specific external logic designs may need to interpret only a subset of the GP codes. For example, a minimal system with no FPA and no need for POK or a bus reset signal would only have to identify a GP code associated with the reading of power-up configuration data during the DCJll's initialization sequence. As shown in the flowchart in Paragraph 8.3.2, this is GP code 002.

8-1

8.3 POWER-UP AND INITIALIZATION

The DCJ11 performs a specific sequence of events at power-up or when it is initialized. These initialization microroutines are described in this paragraph. Also, during power-up the DCJ11 reads the contents of a configuration register to determine its initial mode of operation. This configuration register is also described. A typical power-up circuit is also provided.

8.3.1 Initialization Timing - Initialization timing is shown in Figure 8-1. When external logic asserts INIT for a minimum of 25 clock periods, the DCJ11 is forced into a power-up initialization sequence. As shown in Figure 8-1, the DCJ11 asserts SCTL shortly after the assertion of INIT. SCTL is deasserted approximately five clock periods after INIT is deasserted.



Figure 8-1 Initialization

8.3.2 Initialization Microroutine - The microroutine that is executed when the DCJll is powered up or initialized is shown in Figure 8-2. Note that GP codes that indicate some event (such as the passing of a microdiagnostic test) can be used by external logic to light LEDs for a visual indication of the event.



Figure 8-2 Initialization Sequence



.

Figure 8-2 Initialization Sequence (Continued)



.

Figure 8-2 Initialization Sequence (Continued)



Figure 8-2 Initialization Sequence (Continued)

\$.3.3 Power-Up Configuration - The power-up configuration is specified by setting bits in an external register which is read (via the DAL) during the DCJ11's initialization sequence. It specifies various user-defined initial conditions. The register is shown in Figure 8-3.



Figure 8-3 Power-Up Configuraton Register

Bit(s) Name Description Boot Address <15:9> Contains the most significant seven bits (bits <15:9>) of a user-defined boot address used in power-up mode 3. The lower bits of the boot address (bits <8:0>) are zeroes. 8 FPA Here Indicates the presence of an optional floating-point accelerator (FPA) when set. When cleared, the FPA is not present. <7:4> Unused These bits are not interpreted by the DCJ11. 3 Halt Option Indicates how a HALT instruction will execute in kernel mode. If set, the DCJ11 traps through location 4 and sets bit 7 of the CPU error register when HALT is executed. If cleared, the DCJ11 enters console ODT when HALT is executed. <2:1> Power-Up Mode Indicates one of four powerup mode options. Bits 2 1 Mode 0 0 Trap through location 24 0 1 Enter console ODT 1 0 Power-up to 17773000 1 1 Power-up to the user-defined address specified by bits <15:9> 0 POK Indicates whether the power supply is operating within its normal range. Set when power is at an acceptable value.

MR-11449

8.3.4 Power-Up Circuit - A circuit such as that shown in Figure 8-4 can be used to power-up the DCJ11.

INIT is provided to the DCJ11 by power-up logic and the AIO code is latched by the assertion of ALE. The decoder indicates whether a GP Read of 000 or 002 is being executed.

In this simple application, only DAL<8,3:0> are affected by the power-up configuration register. The register is configured to indicate that no FPA is present, power-up mode 0 (trap through location 24) is selected, and power is always OK. The DAL is driven with configuration data when BUFCTL is asserted and a GP Read with a code of 000 or 002 occurs.



Figure 8-4 Power-Up Circuit

8.4 OTHER MICROROUTINES

Figures 8-5 and 8-6 illustrate two other microroutines whose operation can be monitored by external logic: the power-down microroutine and the console ODT response to entering the "go" command.



Figure 8-5 Power-Down Sequence



.

Figure 8-6 Console Start Sequence



Figure 8-6 Console Start Sequence (Continued)

APPENDIX A DC CHARACTERISTICS

.

~

Absolut	e Maximum Rating										
Storage Active Supply Input c	e Temperature Rang Temperature Range Voltage: or Output Voltage	e: : Applied:	-65 C to -55 C to +7.0V Vss -0.3V Vcc +0.3V	+150 C +125 C							
Electri	cal Characteristi	CS									
Specifi Specifi Test Co	ed Temperature Ra ed Voltage Range nditions	nge	0 C to +70 C +4.75V to +5.25V Temperature = +70 C Vss = 0V Vcc = +4.75V (except as noted)								
Symbol	Parameter	Min.	Max.	Units	Test Condition						
V IH	High level MOS input	70% V cc		v							
v _{IL}	Low level MOS input		30% V _{cc}	v							
V IHT	High level TTL input	2.2		v							
V ILT	Low level TTL input		0.8	v							
I	Input leakage current except TEST inputs (note l)	-10.0	10.0	uA	ov ≤ v _I ≤ v _{cc}						
I ILL	Input current TEST inputs (note l)	0.1	5.0	mA	v = 0v						
I OH	Output current at high level	-2.0		mA	$v_0 = v_{cc} - 0.4v$						
I OL	Output current at low level	2.0		MA	$V_0 = 0.4V$						

A-1

Symbol	Parameter	Min.	Max.	Units	Test Condition
I OHT	Output current at high TTL level	-2.0		Am	$V_0 = 2.4V$
I OSH	High level sustainer current (note l)	-0.2	-0.6	mA	$v_0 = v_{cc} - 1.0v$
I OSL	Low level sustainer current (note 1)	0.2	0.6	μA	v ₀ = 1.0v
I OZ	Output leakage current (notes 1,2)	-10.0	10.0	uA	$0V \leq V_0 \leq V_{cc}$
I CCSB	Static power supply current (notes 1,3)		30	mA	
C IN	Input capacitance (note 4)		7	pF	
с ₁₀	Input/output capacitance (note 4)		15	pF	
C _{OUT}	Output capacitance (note 4)		15	pF	
C MAX	DCJ11 capacitance plus external capacitance		50	pF	
	NC	DTES			

- 1. Tested at Vcc = 5.25V.
- 2. Only applies in the high impedance condition.
- 3. With TEST1 and TEST2 asserted, all outputs open circuit, and all other inputs equal to Vcc.
- 4. Sampled and guaranteed, but not tested. Does not apply to TEST1 or TEST2.

SIGNAL SUMMARY

TYPE	NAME	APPLICABLE DC TEST
TTL INP	UT IRQ<3:0>, HALT, EVENT, PARITY, MISS, CONT, DMR INIT, FPE	PWRF, V _{IHT} , V _{ILT} , I _I DV,
TTL OUT	PUT DAL<21:16>, AIC ALE, BUFCTL, SC STRB, BS<1:0>, PRDC	<3:0>, I _{OL} , I _{OHT} , I _{OZ} TL, MAP,
MOS INP	UT TEST1, TEST2	V _{IH} , V _{IL} , I _{ILL}
MOS OUT	PUT CLK, CLK2	I OH ' I OL ' I OZ
TTL I/O	ABORT *	V ILT , I OL , I OHT , I OZ , I OSH
TTL I/O	DAL<15:00>	V _{IHT} , V _{ILT} , I _{OL} , I _{OHT} , I _{OZ}
Power	Vcc	I CCSB

* $\overline{\text{ABORT}}$ must be driven with an open collector driver because the DCJ11 has a pull-up device that supplies I_{OSH}.

•



Figure A-1 Voltage Waveforms

Test Conditions:

•

Temperature = +70 C Vss = 0V Vcc = +4.75V (except as noted) CMAX = 50 pF

.

Timing Requirements

Symbol	Parameter	Min	Max	Units
t INITW	INIT pulse width	10		clock periods
t SCTLLH	Initialization interval	225		ns
t DS	DAL<15:00> setup, with respect to T3	35		ns
t _{DH}	DAL<15:00> hold, with respect to T3	20		ns
t dvds	DAL<15:00> setup, with respect to DV	35		ns
t dvdh	DAL<15:00> hold, with respect to DV	35		ns
t DVW	DV Pulse width	35		ns
t _{DVF}	DV Fall time		15	ns
t DVH	DV deassertion with respect to T6.	0		ns
t DVS	DV deassertion with respect to T4.	0	· ·	ns
t HMS	MISS setup	30		ns
t HMH	MISS hold	10		ns
t svcs	IRQ<3:0>, HALT, PWRF, FPE, EVENT setup (see note)	20		ns
Symbol	Parameter	Min Ma	x Units	
-------------------	---	------------------------	---------	
t svch	IRQ<3:0>, HALT, PWRF, FPE, EVENT hold (see note)	20	ns	
t PARS	PARITY setup (see note)	20	ns	
t _{PARH}	PARITY hold (see note)	20	, ns	
t ABS	ABORT drive	30	ns	
t ABD	ABORT delay	0	ns	
t ABW	ABORT width	40 + t _{CLKH}	ns	
t CNTS	CONT setup (see note)	30	ns	
t CNTH	CONT hold (see note)	20	ns	
t DMRS	DMR setup (see note)	30	ns	
t DMRH	DMR hold (see note)	20	ns	

Note:

Setup and hold requirements are only to guarantee recognition at next sample point.

.

Timi	na	Res	DOD	9 A A
	** 3		POIL	

Symbol	Parameter	Min	Max	Units	Figure References
t CYCLE	CLK cycle time	67		ns	B-1, B-4
t CLKH	CLK high width	28		ns	B-1, B-4
t CLKL	CLK low width	28		ns	B-1, B-4
t _R	CLK rise time		7	ns	B-1, B-4
t _F	CLK fall time		7	ns	B-1, B-4
t PCYC	CLK2 cycle time	67		ns	B-1, B-3
t PCLKH	CLK2 high width	28 B-2		ns.	B-1, B-3

Parameter	Min	Max	Units	References
CT V2 Jour width				
CLK2 IOW WIGHN	28		ns	B-1, B-3
CLK2 rise time		7	ns	B-1, B-3
CLK2 fall time	• •	7	ns	B-1, B-3
CLK2 valid delay		tbs	ns	B-1, B-3
MAP delay		45	ns	B-1, B-3
Strobe active delay	0		' ns	B-3
Strobe inactive delay	0		ns	B-3
DAL output disable		35	ns	B-2
DAL valid delay		65	ns	B-3
DAL valid hold	0		ns	B-3
PRDC valid delay		50	ns	B-3
PRDC invalid delay		50	ns	B-3
AIO<3:0> delay		75	ns	B-3
	CLK2 low width CLK2 rise time CLK2 fall time CLK2 valid delay MAP delay Strobe active delay Strobe inactive delay DAL output disable DAL valid delay DAL valid delay PRDC valid delay PRDC invalid delay AIO<3:0> delay	CLK2 low width 28 CLK2 rise time CLK2 fall time CLK2 fall time CLK2 valid delay MAP delay Strobe active 0 delay Strobe inactive 0 delay DAL output disable DAL valid delay DAL valid delay DAL valid delay PRDC valid delay PRDC invalid delay AIO<3:0> delay	CLK2 low width28CLK2 rise time7CLK2 fall time7CLK2 valid delaytbsMAP delay45Strobe active0delay0Strobe inactive0DAL output disable35DAL valid delay65DAL valid delay50PRDC valid delay50PRDC invalid50Alo<3:0> delay75	CLK2 low width28nsCLK2 rise time7nsCLK2 fall time7nsCLK2 valid delaytbsnsMAP delay45nsStrobe active0nsdelay0nsStrobe inactive0nsDAL output disable35nsDAL valid delay65nsDAL valid delay50nsPRDC valid delay50nsPRDC invalid50nsAIO<3:0> delay75ns

٠





B-3



Figure B-2

Three State Disable Test Circuit



TTL Output Test Circuit



MR-9425

Figure B-4

MOS Output Test Circuit



Figure B-5 Non-Stretched Bus Read Timing

















Figure B-9 General-Purpose Write Timing

MR 11681









.

APPENDIX C

DCJ11 HARDWARE AND SOFTWARE DIFFERENCES

C.1 HARDWARE DIFFERENCES BETWEEN THE DCJ11 AND THE PDP-11/44

The DCJll may replace the PDP-11/44 in certain applications; however, it does not contain the following PDP-11/44 hardware features:

- o Cache Data and Maintenance Registers (17777750, 1777754)
- o Memory System Error Register (17777744)
- o Switch Register (17777570).

The DCJ11 does contain additional functionality not present in the 11/44:

- o Dual general register set
- O SPL, MTPS, MFPS, TSTSET, WRTLCK instructions.

The following list summarizes the hardware differences between the 11/44 and the DCJ11:

Address	Function	Differences
17777776	PS	Added register set select bit<11>.
17777772	PIRQ	No difference.
17777766	CPU Error	Unibus monitoring bits not implemented.
17777754	Cache Data	Not implemented.
17777752	Hit/Miss	No difference.

1777750	Maintenance	Not implemented.
17777746	Cache Control	Hardware specific changes (see Paragraph 5.2.1).
17777744	Memory Error	Not implemented.
17777676 to 17777660	User Data PAR	No difference.
17777656 to 17777640	User Instruction PAR	No difference.
17777636 to 17777620	User Data PDR	No difference.
17777616 to 17777600	User Instruction PDR	No difference.
17777576	MMR2	No difference.
17777574	MMR1	No difference.
17777572	MMR0	Eliminated maintenance mode.
17777570	Switch Register	Not implemented.
17772516	MMR 3	No difference.
17772376 to 17772360	Kernel Data PAR	No difference.
17772356 to 17772340	Kernel Instruction PAR	No difference.
17772336 to 17772320	Kernel Data PDR	No difference.
17772316 to 17772300	Kernel Instruction PDR	No difference.
17772276 to 17772260	Supervisor Data PAR	No difference.

17772256 to 17772240	Supervisor Instruction PAR	No	difference.
17772236 to 17772220	Supervisor Data PDR	No	difference.
17772216 to 17772200	Supervisor Instruction PDR	No	difference.

C.2 HARDWARE DIFFERENCES BETWEEN THE DCJ11 AND THE PDP11/70

The DCJll may replace the PDP-11/70 in certain applications; however, it does not contain the following PDP-11/70 hardware features:

- o Stack Limit Register (1777774)
- o Micro Break Register (17777770)
- o System ID Register (17777764)
- o System Size Registers (17777760, 17777762)
- o Maintenance Register (17777750)
- o Memory System Error Register (1777744)
- o Physical Error Address Registers (17777740, 1777742)
- o Switch Register (17777570).

The DCJ11 does contain additional functionality not present in the 11/70:

O MTPS, MFPS, MFPT, CSM, TSTSET, WRTLCK instructions

o Bypass cache bit in PDRs.

The following list summarizes the hardware differences between the 11/70 and the DCJ11:

Address	Function	Differences
17777776	PS	Added suspended instruction bit <8>.
17777774	Stack Limit	Not implemented.
1777772	PIRQ	No difference.
17777770	Micro Break	Not implemented.
17777766	CPU Error	No difference.
17777764	System ID	Not implemented.
17777762	System Size	Not implemented.
1777760	System Size	Not implemented.
17777752	Hit/Miss	No difference.
1777750	Maintenance	Not implemented.
17777746	Cache Control	Hardware specific changes (see section 5.2.1).
17777744	Memory Error	Not implemented.
17777742	High Error Address	Not implemented.
17777740	Low Error Address	Not implemented.
17777676 to 17777660	User Data PAR	No difference.
17777656 to 17777640	User Instruction PAR	No difference.
17777636 to 17777620	User Data PDR	Added bypass cache, eliminated access flags and access modes other than 0, 2, and 6.
17777616 to 17777600	User Instruction PDR	Added bypass cache, eliminated access flags and access modes other than 0, 2, and 6.
17777576	MMR2	No difference.
17777574	MMR1	No difference.

.

17777572	MMR 0	Eliminated traps, maintenance mode, and instruction complete.
17777570	Switch Register	Not implemented.
17772516	MMR 3	Added CSM enable bit <3>.
17772376 to 17772360	Kernel Data PAR	No difference.
17772356 to 17772340	Kernel Instruction PAR	No difference.
17772336 to 17772320	Kernel Data PDR	Added bypass cache, eliminated access flag and access modes other than 0, 2, and 6.
17772316 to 17772300	Kernel Instruction PDR	Added bypass cache, eliminated access flag and access modes other than 0, 2, and 6.
17772276 to 17772260	Supervisor Data PAR	No difference.
17772256 to 17772240	Supervisor Instruction PAR	No difference.
17772236 to 17772220	Supervisor Data PDR	Added bypass cache, eliminated access flag and access modes other than 0, 2, and 6.
17772216 to 17772200	Supervisor Instruction PDR	Added bypass cache, eliminated access flag and access modes other than 0, 2, and 6.

C.3 SOFTWARE DIFFERENCES

Table C-1 summarizes the programming differences (at the assembly language level) between the DCJ11 and other processors in the PDP-11 family.

ITEM	23/24	44	04	34	LSI11	05/10	15/20	35/40	45	70	60	J-11	T-11	VAX
1. OPR %R, (R) +; OPR %R, $-$ (R) using the same register as both source and destination: contents of R are incre- mented (decremented) by 2 before being used as the source operand. OPR %R, (R) +; OPR %R, $-$ (R) using the same register as both register and des- tination: initial contents of R are used as the source operand.	X	×	X	x	x	x	X	x	x	x	X	X	X	x
2. OPR %R, $@$ (R) +; OPR %R, $@$ – (R) using the same register as both source and destination: contents of R are incre- mented (decremented) by 2 before being used as the source operand. OPR %R, $@$ (R) +; OPR %R, $@$ – (R) using the same register as both source and destination: initial contents of R are used as the source operand.	x	x	x	x	x	x	X	X	x	×	x	×	×	x
3. OPR PC, X (R); OPR PC, @ X (R); OPR PC, @ A; OPR PC, A: location A will con- tain the PC of OPR $+ 4$. OPR PC, X (R); OPR PC, @ X (R), OPR PC, A; OPR PC, @ A: location A will con- tain the PC of OPR $+ 2$.	x	x	x	x	x	X	X	X	x	x	X	X	X	x
4. JMP (R) + or JSR reg, (R) +: contents of R are incremented by 2, then used as the new PC address. JMP (R) + or JSR reg, (R) +: initial con- tents of R are used as the new PC.	x	x	x	x	×	×	×	x	x	x	×	x	x	x

PROCESSORS

Table C-1 DCJ11 Programming Diffenences

ITEM	23/24	44	04	34	LSI11	05/10	15/20	35/40	45	70	60	J-11	T-11	VAX
5. JMP %R or JSR reg, %R traps to 10 (illegal instruction). JMP %R or JSR reg, %R traps to 4 (illegal instruction).	X	x	x	x	x	X	X	X	x	x	X	x	Х	NA NA
6. SWAB does <i>not</i> change V. SWAB clears V.	x	x	x	x	x	x	х	x	x	x	x	x	x	x
7. Register addresses (177700-177717) are valid program addresses when used by CPU.						X							_1	. –1
Register addresses (177700–177717) time out when used as a program address by the CPU. Can be addressed under console operation.		X	×	X			X	×	X	X	X			NA
Register addresses (177700–177717) time out when used as an address by CPU or console.	X				X							X		
8. Basic instructions noted in PDP-11 processor handbook.	X	X	X	X	X	Х	X	X	X	X	X	X	X	Х
SOB, MARK, RTT, SXT instructions*	X	X		X	х			x	x	Х	х	x	х	_2
ASH, ASHC, DIV, MUL, XOR	X	Х		X	x			X	X	Х	Х	x		x
Floating Point instructions in base machine.											х	x		
MFPT Instruction.	X	Х										x		
The external option KE11-A provides MUL, DIV, SHIFT operation in the same data format.						X	x							

*RTT instruction is available in 11/04 but is different than other implementations.

¹ Register addresses (177700–177717) are handled as regular memory addresses in the I/O page. ² All but MARK.

.

ITEM	23/24	44	04	34	LSI11	05/10	15/20	35/40	45	70	60	J-11	T-11	VAX
The KE11-E (Expansion Instruction Set) provides the instructions MUL, DIV, ASH, and ASHC. These new instructions are 11/45 compatible.								х						
The KE11-F (Floating Instruction Set) adds unique stack ordered oriented point instructions: FADD, FSUB, FMUL, FDIV.				 				х	·					
The KEV-11 adds EIS/FIS instructions					Х									
MFP, MTP instructions	X	Х		Х				X		x	X	X		
SPL Instruction		Х							х	х		Х		
CSM Instruction		Х										х		
9. Power fail during RESET instruction is not recognized until after the instruction is finished (70 milliseconds). RESET instruction consists of 70 millisecond pause with INIT occurring during first 20 milliseconds.		-					×	x			x			
Power fail immediately ends the RESET instruction and traps if an INIT is in progress. A minimum INIT of 1 micro- second occurs if instruction aborted. PDP11-04/34/44 are similar with no minimum INIT time.		×	X .	×					x	x				
Power fail acts the same as 11/45 (22 milliseconds with about 300 nano- seconds minimum). Power fail during RESET fetch is fatal with no power down sequence.						X				•				

ITEM	23/24	44	04	34	LSI11	05/10	15/20	35/40	45	70	60	J-11	T-11	VAX
RESET instruction consists of 10 micro- seconds of INIT followed by a 90 micro- second pause. Reset instruction con- sists of a minimum 8.4 microseconds followed by a minimum 100 nanosecond pause. Power fail not recognized until the instruction completes.	×		•		x							x		
10. No RTT instruction If RTT sets the "T" bit, the "T" bit trap occurs after the instruction following RTT.	x	x	x	×	x	×	x	x	X	x	x	x	x	x
11. If RTI sets "T" bit, "T" bit trap is acknowledged after instruction following RTI.		X				×	x							x
If RTI sets "T" bit, "T" bit trap is acknowledged immediately following RTI.	×	Х	X	X	X			X	X	X	X .	X	x	
12. If an interrupt occurs during an instruction that has the "T" bit set, the "T" bit trap is acknowledged before the interrupt.	×	X	X	X	X	X	x	x			x	x	X	NA ¹
If an interrupt occurs during an instruc- tion and the "T" bit is set, the interrupt is acknowledged before "T" bit trap.								i.	x	x				NA
13. "T" bit trap will sequence out of WAIT instruction.	X	X	X	x		X	X	х			X		x	NA
"T" bit trap will not sequence out of WAIT instruction. Waits until an interrupt.					x				x	x				

¹Interrupts not visible to VAX compatibility mode.

ITEM	23/24	44	04	34	LSI11	05/10	15/20	35/40	45	70	60	J-11	T-11	VAX
14. Explicit reference (direct access) to PS can load "T" bit. Console can also load "T" bit.			x			x	X							
Only implicit references (RTI, RTT, traps and interrupts) can load "T" bit. Console cannot load "T" bit.	X	X		X	X			X	x	X	X	X	X	X
15. Odd address/non-existent references using the SP cause a HALT. This is a case of double bus error with the second error occurring in the trap servicing the first error. Odd address trap not imple- mented in LSI-11, 11/23 or 11/24.		X	X	X	×	×	×							
Odd address/non-existent references using the stack pointer cause a fatal trap. On bus error in trap service, new stack created at 0/2.	x							X	X	X	X	X	_1	_2
16. The first instruction in an interrupt routine will not be executed if another interrupt occurs at a higher priority level than assumed by the first interrupt. The first interrupt in an interrupt service is guaranteed to be executed.	×	X	X	x	x	×	x	×	X	X	X	X	X	x
17. Single general purpose register set implemented.	X	x	x	x	X	x	x	X			x		x	X
Dual general purpose register set implemented.									X	×		X		

¹Odd address/non-existent references using SP do not trap. ²Odd address aborts to native mode.

ITEM	23/24	44	04	34	LSI11	05/10	15/20	35/40	45	70	60	J-11	T-11	VAX
18. PSW address, 177776, not imple- mented; must use instructions MTPS (move to PS) and MFPS (move from PS).					X								X	_3
PSW address implemented, MTPS and MFPS not implemented.		Х	X			X	×	X	X	X	×		,	
PSW address and MTPS and MFPS implemented.	X			X								×		
19. Only one interrupt level (BR4) exists. Four interrupt levels exist.	×	x	x	x	x	×	x	×	x	x	x	x	x	NA
20. Stack overflow not implemented. Some sort of stack overflow implemented.	x	x	x	x	x	x	x	x	x	x	x	x	×	×
21. Odd address trap not implemented. Odd address trap implemented.	×	x	x	x	x	×	x	x	X	x	x	x	×	x
22. FMUL and FDIV instructions implicity use R6 (one push and pop); hence R6 must be set up correctly. FMUL and FDIV instructions do not implicitly use R6					X			x						NA
23. Due to their execution time, EIS instructions can abort because of a device interrupt.					×									X
EIS instructions do not abort because of a device interrupt.	x	x		×				×	x	Χ.	×	X		NA
24. Due to their execution time, FIS instructions can abort because of a device interrupt.					X			x						NA

· •••••

³Can reference PSW only from native mode.

,

.

x													
	X		X					Х	X	Х	X		NA
				X									
X	х		×				x	х	x	х	х		NA
X	x		X	×			x	x	x	x	×		1
		X			X	Х						_2	
x	X	x	x	x	X	- X		X	x		x		
							×					_3	×
×	Y	v	Y	x	X	X	X	X	X		X	_3	
	x x x	x x x x x x x x x x x x x x x x x x x	x x x x x x x x x x x x x x x x x x x	xxxxxxxxxxxxxxxxxx	xxxxxxxxxxxxxxxxxxxxxxxxxxxxx	xxxxxxxxxxxxxxxxxxxxxxxxxxxxxxxxxxxxxxx	xx	xx	xxx	xx	x x	xxx <t< td=""><td>x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x <t< td=""></t<></td></t<>	x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x x <t< td=""></t<>

Integral floating point assumed on 11/23 and 11/24; FP11E assumed for 11/60.
Implementation dependent.
MOV instruction does a DATI and a DATO bus sequence for last memory cycle.
³ Does not support bus errors.

ITEM	23/24	44	04	34	LSI11	05/10	15/20	35/40	45	70	60	J-11	T-11	VAX
30. If register contains an odd value in mode 2 and a bus error occurs, register will be incremented.	x				x			x	x	X		X	_3	_4
If register contains an odd value in mode 2 and a bus error occurs, register will be unchanged.		х	x	X		x	x							
31. Condition codes restored to original values after FIS interrupt abort (EIS doesn't abort on 35/40).								x						
Condition codes that are restored after EIS/FIS interrupt abort are indeterminate.					X									NA
32. Opcodes 075040 through 075377 unconditionally trap to 10 as reserved opcodes. If KEV-11 option is present, opcodes 75040 through 07533 perform a memory read using the register specified by the low order 3 bits as a pointer. If the register contents are a non-existent	×	x	X	X	x	×	X	X	X	X	X	X	x	1
register contents are an existent address, a trap to 10 occurs.														
33. Opcodes 210 thru 217 trap to 10 as reserved instructions. Opcodes 210 thru 217 are used as a maintenance instruction.	X	X	X	X	x	X	X	x	X	X	X	X	X	_1

.

³ Does not support bus errors. ⁴ Unpredictable. ¹ Traps to native mode.

ITEM	23/24	44	04	34	LSI11	05/10	15/20	35/40	45	70	60	J-11	T-11	VAX
34. Opcodes 75040 thru 75777 trap to 10 as reserved instructions.	X	х	х	X		x	Х	x	x	x	X	х	х	_1
If KEV-11 options is present, opcodes 75040 thru 75577 can be used as escapes to user microcode. If no user microcode exists, a trap to 10 occurs.					X									
35. Opcodes 170000 thru 177777 trap to 10 as reserved instructions.			х			X	. X	х					х	_1
Opcodes 170000 thru 177777 are implemented as floating point instructions.	x	X		х					x	x	x	X		
Opcodes 170000 thru 177777 can be used as escapes to user microcode. If no user microcode exists, a trap to 10 occurs.					x									
Opcode 076600 used for maintenance.											Х			
36. CLR and SXT do just a DATO sequence for the last bus cycle.	×	Y	Y	Y	v	v	Y	×	Y	Y	Y	X	_2	_1
for the last bus cycle.		^		^					Ŷ	Ŷ				
37. MEM MGT maintenance mode MMR0 bit 8 is implemented.		x		x				x	х	х	X			
MEM MGT maintenance mode MMR0 bit 8 is not implemented.	x											x		NA
38. PS<15:12>, non-kernel mode, non- kernel stack pointer and MTPx and MFPx instructions exist even when MEM MGT is not configured.	X	x							X	X	. X	X		

¹ Traps to native mode.

¹ Unpredictable. ² CLR and SXT do DATI-DATO.

,

C-14

23/24	44	04	34	LSI11	05/10	15/20	35/40	45	70	60	J-11	T-11	VAX
							X						NA
x			x				x						NA
	X							X	x		x		
			X										
X										-	X		NA
x			x								x		NA
	23/24 X X X X	23/24 44 X X X X X X	23/24 44 04	23/24 44 04 34 X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X	23/24 44 04 34 LSI11 X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X	23/24 44 04 34 LSI11 05/10 X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X	23/24 44 04 34 LSI11 05/10 15/20 X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X <	23/24 44 04 34 LSI11 05/10 15/20 35/40 X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X X	23/24 44 04 34 LSI11 05/10 15/20 35/40 45 X	23/24 44 04 34 LSI11 05/10 15/20 35/40 45 70 X <td< td=""><td>23/24 44 04 34 LSI11 05/10 15/20 35/40 45 70 60 X <t< td=""><td>23/24 44 04 34 LSI11 05/10 15/20 35/40 45 70 60 J-11 X</td><td>23/24 44 04 34 LSIII 05/10 15/20 35/40 45 70 60 J-11 T-11 X</td></t<></td></td<>	23/24 44 04 34 LSI11 05/10 15/20 35/40 45 70 60 X <t< td=""><td>23/24 44 04 34 LSI11 05/10 15/20 35/40 45 70 60 J-11 X</td><td>23/24 44 04 34 LSIII 05/10 15/20 35/40 45 70 60 J-11 T-11 X</td></t<>	23/24 44 04 34 LSI11 05/10 15/20 35/40 45 70 60 J-11 X	23/24 44 04 34 LSIII 05/10 15/20 35/40 45 70 60 J-11 T-11 X

¹ Unpredictable. ² CLR and SXT do DATI-DATO.

ITEM	23/24	44	04	34	LSI11	05/10	15/20	35/40	45	70	60	J-11	T-11	VAX
42. Programs cannot execute out of internal processor registers. Programs can execute out of internal processor registers.	x	x		x				x	X	x	x	x		
43. A HALT instruction in user or super- visor mode will trap thru location 4. A HALT instruction in user or supervisor mode will trap thru location 10.	x	x		x				x	x	x	x	x	1	_2
44. PDR bit <0> implemented. PDR bit <0> not implemented.	x	х		x				x	Х	×	x	x	x	x
45. PDR bit <7> (any access) implemented. PDR bit <7> (any access) not implemented.	x	x		x				x	X	x	x	x	x	x
46. Full PAR $<15:0>$ implemented. Only PAR $<11:0>$ implemented.	×	Х		x				x	x	x	x	x	x	x
47. MMR0<12>trap-memory management-implemented. MMR0<12> not implemented.	x	x		x				x	X	X	x	x	x	x
48. MMR3<2:0>-D space enable- implemented. MMR3<2:0> not implemented.	x	х		x				x	x	x	x	x	x	x
49. MMR3<5:4>IOMAP, 22-bit mapping enabledimplemented. MMR3<5:4> not implemented.	X	X		x				x	x	x	X	X	x	x

.

¹ HALT pushes PC & PSW to stack, loads PS with 340 and PC with < powerup address> + 40. ² Traps to native mode.

•

ITEM	23/24	44	04	34	LSI11	05/10	15/20	35/40	45	70	60	J-11	T-11	VAX
50. MMR3<3>-CSM enable- implemented. MMR3<3> not implemented.	×	X		- X				x	x	x	x	×	x	x
51. MMR2 tracks instruction fetches and interrupt vectors. MMR2 tracks only instruction fetches.	x	x		x				x	х	x	x	x	NA	NA
52. MFPx %6, MTPx when PS<13:12>= 10 gives unpredictable results. MTPx %6, MTPx %6 when PS<13:12>= 10 uses user stack pointer.	X	X		X				x	x	x	X	×	NA	NA

 1 HALT pushes PC & PSW to stack, loads PS with 340 and PC with $<\!powerup address\!>$ + 40. 2 Traps to native mode.

-.

The execution time for an instruction depends on: (1) the type of instruction executed, (2) the the mode of addressing used, and (3) the type of memory being referenced. In general, the total execution time is the sum of the base instruction fetch/execute time plus the operand(s) address calculation/fetch time.

The tables in this appendix can be used to calculate the length of an instruction in terms of microcycles (MC). In the first group of tables, the first column specifies the number of microcycles required to fetch/execute the base instruction. The R/W column specifies how many of these microcycles are read microcycles and how many are write microcycles (any remaining microcycles are NIO). If the instruction involves the calculation/fetch of one or more operands, a reference to a separate table (a source or destination table) is made in the last column(s). The source/destination tables how many microcycles the reveal source/destination calculation/fetch takes and also specifies how many of these are read or write microcycles (again, any remaining microcycles are NIO).

The numbers in the tables are based on the assumption that a memory read must last a minimum of four CLK periods, a memory write must last a minimum of eight CLK periods, and an NIO lasts four CLK periods (no DMA). Any wait states caused by slower memory or a DMA transfer must be added to the total instruction time. If wait states are required, the first wait state of a non-stretched read or NIO cycle will last four clock periods, and can continue in increments of two clock periods. Further wait states for stretched cycles occur in increments of two clock periods.

Floating-point instruction execution times are given as a range. The actual execution time will vary depending on the type of data being operated on. Here are two examples of how to use the tables:

Example 1:

How long does a MOV R0, @#2044 instruction last?

- Step 1: From the tables, the execution time for the MOV base instruction is found to be 1 microcycle (MC), or four CLK periods. This consists of one read and no write microcycles. Depending upon the type of memory in the system, the microcycle may be stretched. If so, the microcycle lasts at least eight CLK periods and may be stretched thereafter in increments of two CLK periods.
- Step 2: To find the operand calculation/fetch time for the source operand (R0), refer to Table S1. From Table S1, it is seen that a mode 0 register 0 calculate/fetch takes 0 microcycles. Note that the operand is already available to the DCJ11 (in the register file).
- Step 3: To find the operand calculation/fetch time for the destination operand (the contents of memory location 2044), refer to Table D3. From Table D3, it is seen that a mode 3 register 7 calculate/fetch takes 3 microcycles, one of which is a read microcycle and one of which is a write microcycle. Note that the remaining microcycle is an NIO microcycle. Once again, the type of memory in the system must be taken into account. If the read cycle is stretched, the stretched cycle lasts at least eight CLK periods and may be stretched thereafter in increments of two CLK periods. The write microcycle lasts at least eight CLK periods and may be stretched in increments of two CLK periods.
- Step 4: For a determination of the minimum time required, total up the microcycles. In this example, It is 1 + 0 + 3, or 4 microcycles (which is 16 CLK periods if no microcycle stretching occurs).

Example 2:

The source and destination tables for floating point instructions show a negative number in the MC column for certain mode 2 register 7 operations. This example shows a timing calculation for one of these.

How long does an CLRD #2000 instruction last?

Step 1: The base instruction time for the CLRD instruction is 14 microcycles.

- Step 2: From Table F2, the calculation/fetch time for the operand (a mode 2 register 7 reference) is shown as (-1). This means that one microcycle should be subtracted from the base instruction time. However, add one microcycle for the memory write operation. There are no memory read cycles to account for.
- Step 3: Total up the microcycles: 14 1 + 1 = 14 microcycles minimum (assumes no cycle stretching).

SINGLE OPERAND

Mnemonic	Instruction	Exec MC	ution R/W	Source Table	Dest Table
General					
CLR (B) COM (B) INC (B) DEC (B) NEG (B) TST (B)	Clear Complement (l's) Increment Decrement Negate (2's complement) Test	1 1 1 1 1	1/0 1/0 1/0 1/0 1/0 1/0		D3 D4 D4 D4 04 D4 D4
Rotate an	d Shift				
ROR (B) ROL (B) ASR (B) SWAB	Rotate right Rotate left Arithmetic shift right Swap bytes	1 1 1 1	1/0 1/0 1/0 1/0		D4 D4 D4 D4
Multiple	Precision				
ADC (B) SBC (B) SXT	Add carry Subtract carry Sign extend	1 1 1	1/0 1/0 1/0		D4 D4 D3
Multiproc	cessing				
TSTSET WRTLCK	Test and set (low bit interlocked) Write interlocked	5 4	1/1 1/1		D4 D4
DOUBLE OI	PERAND			TIMING	
Mnemonic	Instruction	Exec MC	ution R/W	Source Table	Dest Table
General					
MOV (B) CMP (B) ADD SUB	Move Compare Add Subtract	1 1 1	1/0 1/0 1/0 1/0	S1 S1 S1 S1	D3 D2 D4 D4
Logical					
BIT(B) BIC(B) BIS(B)	Bit test (AND) Bit clear Bit set (OR)	1 1 1	1/0 1/0 1/0	S1 S1 S1	D2 D4 D4

Register

MUL	Multiply	22 1/0 -	D1 (Notes 5,11)
DIV	Divide	34 1/0	D1 (Notes 6,7,12)
ASH	Shift automatically	4 1/0	Dl
ASHC	Arith shift combined	5 1/0	D1 (Note 13)
XOR	Exclusive (OR)	1 1/0	D4

.

BRANCH

TIMING

		Bran Not	nch Taken	Brai Tak	nch en
Mnemonic	Instruction	MC	R/W	MC	R/W
Branches					
BR BNE BEQ BPL BMI BVC BVS BCC BCS	Branch (unconditional) Br if not equal (to 0) Br if equal (to 0) Br if plus Br if minus Br if overflow is clear Br if overflow is set Br if carry is clear Br if carry is set	2 2 2 2 2 2 2 2 2 2 2 2 2 2	1/0 1/0 1/0 1/0 1/0 1/0 1/0 1/0	4 4 4 4 4 4 4 4 4	2/0 2/0 2/0 2/0 2/0 2/0 2/0 2/0 2/0
Signed Con	ditional Branches				
BGE BLT BGT BLE	Br if greater or equal (to 0) Br if less than (0) Br if greater than (0) Br if less or equal (to 0)	2 2 2 2	1/0 1/0 1/0 1/0	4 4 4	2/0 2/0 2/0 2/0
Mnemonic	Instruction	Bran Not MC	ch Taken R/W	Brar Take MC	nch en R/W
Unsigned Co	onditional Branches				
BHI BLOS BHIS BLO	Branch if higher Branch if lower or same Branch if higher or same Branch if lower	2 2 2 2	1/0 1/0 1/0 1/0	4 4 4 4	2/0 2/0 2/0 2/0
SOB	Subtract 1 and branch (if \neq 0)	3	1/0	5	2/0
JUMP and Si	BROUTINE		TIMIN	G	

		Execution			
Mnemonic	Instruction	MC	R/W	DST	Table
JMP	Jump			D5	
JSR	Jump to subroutine			D6	(Note 4)
RTS MARK	Return from subroutine Stack cleanup	5 10	3/0 3/0	-	(Note 14)

• D-5

TRAP and INTERRUPT		TIMING		
		Exe	cution	
Mnemonic	Instruction	MC	R/W	
EMT	Emulator trap	20	4/2	
TRAP	Trap	20	4/2	
BPT	Breakpoint trap	20	4/2	
IOT	Input/output trap	20	4/2	
RTI	Return from interrupt	9	4/0	
RTT	Return from interrupt	9	4/0	
CONTRACIN	CODE OBERAMORIC		•	
CONTIN	CODE OFERATORS	TIMI	NG	

_		Execution	
Mnemonic	Instruction	MC	R/W
CLC	Clear C	3	1/0
CLV	Clear V	3	1/0
CLZ	Clear Z	3	1/0
CLN	Clear N	3	1/0
CCC	Clear all CC bits	3	1/0
SEC	Set C	3	1/0
SEV	Set V	3	1/0
SEZ	Set Z	3	1/0
SEN	Set N	3	1/0
SCC	Set all CC bits	3	1/0

MISCELLANEOUS

TIMING

Mnemonic	Instruction	Exe MC	cutión R/W	Dest Table
HALT	Halt	-		
WAIT	Wait for interrupt	-		
RESET	Reset external bus	-		
NOP	(No operation)	3	1/0	
SPL	Set priority level to N	7	1/0	
MFPI	Move from previous instr space	5	1/1	Dl
MTPI	Move to previous instr space	3	2/0	D3
MFPD	Move from previous data space	5	$\frac{1}{1}$	DI
MTPD	Move to previous data space	3	$\frac{2}{2}$	50
MTPS	Move byte to PSW PS \leftarrow (svc)	8	$\frac{1}{1}$	DI
MFPS	Move byte from PSW (dst) < PS <7:0>	1	1/0	22
MFPT	Move from processor (R0<7:0> <proc code<="" td=""><td>2</td><td>$\frac{1}{1}$</td><td></td></proc>	2	$\frac{1}{1}$	
CSM	Call to supervisor mode	28	3/3	Dl

FLOATING POINT		TIMING			
Mnemonic	Instruction	Execu Min	ition Typ	(MC) Max	Non Mode O Table
ABSD	Make Absolute	23		24	F3
ABSF	Make Absolute	19	_	20	F3
ADDD	A dd	41	48	119	F1
ADDF	Add	31	35	102	Fl
CFCC	Copy Floating Condition Codes	5		5	
CLRD	Clear	14		14	F2
CLRF	Clear	12		12	F2
CMPD	Compare	24		25	Fl
CMPF	Compare	18		19	F1
DIVD	Divide	160		167	F1
DIVE	Divide	59		63	Fl
LDCDF	Ld & C from D to F	24		26	F1
LDCFD	Ld & C from F to D	20		21	Fl
LDCID	Ld & C Integer to D	31		4 2 ·	F4
LDCIF	Ld & C Integer to F	26		36	F4
LDCLD	Ld & C Long Integer to D	31		52	F4
LDCLF	Ld & C Long Integer to F	26		44	F4
LDD	Load	16		17	Fl
LDEXP	Load Exponent	17		18	F4
LDF	Load	12		13	Fl
LDFPS	Load FPP Program Status	6		6	F4
MODD	Multiply and Separate	202	217	268	Fl
MODF	Integer and Fraction	82	94	115	Fl
MULD	Multiply	165		173	Fl
MULF	Multiply	56		61	Fl
NEGD	Negate	22		23	F3
NEGE	Negate	18		19	F3
SETD	Set Floating Double Mode	6		6	
SETF	Set Floating Mode	6		6	
SETI	Set Integer Mode	6		6	
SETL	Set Long Integer Mode	6		6	-
STCDF	St & C from D to F	17		20	F 2
STCDI	St & C from D to Integer	26		38	F5
STCDL	St & C from D to Long Integer	26		54	F5
STCFD	St & C from F to D	19		20	F2
STCFI	St & C from F to Integer	23		35	F 5
STCFL	St & C from F to Long Integer	23		51	F 5
STD	Store	12		12	F 2
STECP	Store Exponent	16		16	F 5
STF	Store	8		8	F2
STFPD	Store FPP Program Status	9		9	F 5
STST	Store FPP Status	7		7	F 5
SUBD	Subtract	47	55	122	Fl
SUBF	Subtract	37	41	104	Fl
TSTD	Test	11		12	Fl
TSTF	Test	9		10	Fl

ł

D-7

SOURCE AND DESTINATION TABLES:

Table Sl	Source Address	Time: All Dou	ble Operand
Source Mode	Source Register	Microcode Cycles	Read Memory Cycles
0	0-7	0	0
1	0-7	2	1
2	0-6	2	1
2	7	1	1
3	0-6	4	2
3	7	3	2
4	0-6	3	1
4	7	6	2 (Note 1)
5	0-6	5	2
5	7	8	3 (Note 1)
6	0-7	4	2
7	0-7	6	3

Table Dl Destination Address Time: Read Only Single Operand

	Destination Mode	Destination Register	Microcode Cycles	Read Memory Cycles
	0	0-7	0	0
	1	0-7	2	1
	2	0-6	2	1
	2	7	1	1
	3	0-6	4	2
	3	7	3	2
	4	0-6	3	1
	4	7	7	2 (Note 2)
	5	0-6	5	2
	5	7	9	3 (Note 3)
	6	0-7	4	2
	7	0-7	6	3
Table :	D2 Destinatio	on Address Tim	e: Read Only	y Double Operand
				Read
	Destination	Destination	Microcode	Memory
	Mode	Register	Cycles	Cycles
	0	0-7	0	0
	1	0-7	3	1
	2	0-6	3	1
	2	7	2	1
	3	0-6	5	2
	3	7	4	2
	4	0-6	4	1
	4	7	8	2 (Note 2)
	5	0-6	6	2
	5	7	10	3 (Note 3)
	5	0-7	5	2
	1	0-/	1	3

Table D3 Destination Address Time: Write Only

Destination Mode	Destination Register	Microcode Cycles	Memory Read	Cycles Write
0	0-6	0	0	0
0	7	5	1	0
1	0-6	2	0	1
1	7	6	1	1
2	0-6	2	0	1
2	7	6	1	1
3	0-6	4	1	1
3	7	3	1.	1
4	0-6	3	0	1
4	7	7	1	1
5	0-6	5	1	1
5	7	9	2	1
6	0-7	4	1	1
7	0-7	6	2	1

Table D4 Destination Address Time: Read Modify Write

Destination Mode	Destination Register	Microcode Cycles	Memory Read	Cycles Write
0	0-6	0	0	0
0	7	5	1	0
1	0-6	3	1	1
1	7	7	2	1
2	0-6	3	1	1
2	7	7	2	1
3	0-6	5	2	1
3	7	4	2	1
4	0-6	4	1	1
4	7	8	2	1 (Note 2)
5	0-6	6	2	1
5	7	10	3	1 (Note 3)
6	0-7	5	2	1
7	0-7	7	3	1

Table D5 Destination Address Time: JMP

Destination Mode	Destination Register	Mic rocode Cycles	Memory Read	Cycles Write
1	0-7	4	2	0
2	0-7	6	2	0
3	0-7	5	3	0
4	0-7	5	2	0
5	0-7	6	3	0
6	0-6	6	3	0
6	7	5	3	0
7	0-7	7	4	0

Table D6 Destination Address Time: JSR

Destination Mode	Destination Register	Microcode Cycles	Memory Read	Cycles Write
1	0-7	9	2	1
2	0-7	10	2	1
3	0-6	10	3	1
3	7	9	3	1
4	0-7	10	2	1
5	0-7	11	3	1
6	0-6	10	3	1
6	7	9	3	1
7	0-7	12	4	i.

Table Fl Floating Source Modes 1-7

Single Precision

Mode	Register	Microcode Cycles	Memory Read	Memory Write
1	0-7	3	2	0
2	0-6	3	2	0
2	7	1	1	0
3	0-6	4	3	0
3	7	3	3	0
4	0-7	4	2	0
5	0-7	5	3	0
6	0-7	4	3	0
7	0-7	6	4	0

Double Precision

Mode	Register	Microcode Cycles	Memory Read	Memory Write
1	0-7	5	4	0
2	0-6	5	4	0
2	7	0 (Note 15)	1	0
3	0-6	6	5	0
3	7	5	5	0
4	0-7	6	4	0
5	0-7	7	5	0
6	0-7	6	5	0
7	0-7	8	6	0

Table F2 Floating Destination Modes 1-7

Single Precision

Mode	Register	Microcode Cycles	Memory Read	Memory Write
1	0-7	3	0	2
2	0-6	3	0	2
2	7	1	0	1
3	0-6	4	1	2
3	7	3	1	2
4	0-7	4	0	2
5	0-7	5	1	[′] 2
6	0-7	4	1	2
7	0-7	6	2	2

Double Precision

Mode	Register	Microcode Cycles	Memory Read	Memory Write
1	0-7	5	0	4
2	0-6	5	0	4
2	7	(-1) (Note 15)	0	1
3	0-6	6	1	4
3	7	5	1	4
4	0-7	6	0	4
5	0-7	7	1	4
6	0-7	6	1	4
7	0-7	8	2	4

Table F3 Floating Read Modify Write Modes 1-7

Single Precision

Mode	Register	Microcode Cycles	Memory Read	Memory Write
1	0-7	5	2	2
2	0-6	5	2	2
2	7	1 (Note 15)	1	1
3	0-6	6	3	2
3	7	5	3	2
4	0-7	6	2	2
5	0-7	7	3	2
6	0-7	6	3	2
7	0-7	8	4	2
Double Precision

Microo Mode Register Cycles		Microcode Cycles	Memory Memory Read Write	
1	0-7	9	4	4
2	0-6	9	4	4
2	7	(-2) (Note 15)	1	1
3	0-6	10	5	4
3	7	9	5	4
4	0-7	10	4	4
5	0-7	11	5	4
6	0-7	10	5	4 .
7	0-7	12	6	4

Table F4

Integer Source Modes 1-7

Integer

Mode	Register	Microcode Cycles	Memory Read	Memory Write
1	0-7	2	1	0
2	0-6	2	1	0
2	7	0 (Note 15)	1	0
3	0-6	3	2	0
3	7	2	2	0
4	0-7	3	1	0
5	0-7	4	2	0.
6	0-7	3	2	0
7	0-7	5	3	0

Long Integer

Mode	Register	Microcode Cycl es	Memory Read	Memory Write
1	0-7	4	2	0
2	0-6	4	2	0
2	7	Ö (Note 15)	1	0
3	0-6	5	3	0
3	7	4	3	0
4	0-7	5	2	0
5	0-7	6	3	0
6	0-7	5	3	0
7	0-7	7	4	0

Integer

Mode	Register	Microcode Cycles	Memory Read	Memory Write
1	0-7	2	0	1
2	0-6	2	0	1
2	7	2	0	1
3	0-6	3	1	1
3	7	2	1	1
4	0-7	3	0	1
5	0-7	4	1	´ 1
6	0-7	3	1	1
7	0-7	5	2	1

Long Integer

Mode	Register	Microcode Cycles	Memory Read	Memory Write
1	0-7	4	0	2
2	0-6	4	0	2
2	7	2	0	1
3	0-6	5	1	2
3	7	4	1	2
4	0-7	5	0	2
5	0-7	6	1	2
6	0-7	5	1	2
7	0-7	7	2	2

NOTES

- 1. Subtract 2 microcycles (MC) and one read if both source and destination modes autodecrement PC, or if WRITE-ONLY or READ-MODIFY-WRITE mode 07 or 17 is used.
- 2. READ-ONLY and READ-MODIFY-WRITE destination mode 47 references actually perform 3 READ operations. For bookkeeping purposes, one of the READs is accounted for in the EXECUTE, FETCH TIMING.
- 3. READ-ONLY and READ-MODIFY-WRITE destination mode 57 references actually perform 4 READ operations. For book-keeping purposes, one of the READs is accounted for in the EXECUTE, FETCHING TIMING.
- 4. Subtract 1 MC if the link register is PC.
- 5. Add 1 MC if the source operand is negative.
- 6. Subtract 1 MC if the source mode is not zero.
- 7. Add 1 MC if the quotient is even. Add 2 MC if overflow occurs. Add 5 MC and 1 read if the PC is used as a destination register, but only if source mode 47 or 57 is not used.
- 8. Add 1 MC per shift.
- 9. Add 1 MC if source operand <15:6> is not zero.
- 10. Subtract 1 MC if one shift only.
- 11. Add 4 MC and 1 read if the PC is used as a destination register, but only if source mode 47 or 57 is not used.
- 12. Divide by zero executes in 5 MC (see note 6).
- 13. Timing for no shift. Add 1 MC if a left shift. (Notes 8, 9, 11 apply.) Add 2 MC for a right shift. (Notes 8, 10, 11 apply.)
- 14. Add one MC if a register other than R7 is used.
- 15. Mode 27 references only access single word operands. The execution time listed has been compensated in order to accurately compute the total execution time.

Bus lock -An indication to memory to prevent or "lock" out other accesses to that location until it is unlocked. This occurs during an RMW read bus microcycle with the bus lock control bit asserted. Memory is automatically unlocked by the following Bus Write microcycle by that processor. Cache bypass -Unconditionally bypass cache and access main memory directly. If the cache entry is valid, typically invalidate it. Cache force miss -Unconditionally bypass cache and access main memory directly. If the cache entry is valid, typically do not invalidate it but ignore it. Data stream bus cycle -Any microcycle which is a Read, Read/Modify/Write or Write microcycle. Demand abort -An abort during a demand bus microcycle. Instruction stream bus cycle -Any microcycle which is a prefetch microcycle. Internal registers -These explicitly addressable registers are the PS, PIRQ, MMR0, MMR1, MMR2, MMR3, Hit/Miss, CPU Error, PARs, and PDRs. Predecode -An indication to decode the next PDP-11 instruction. This occurs during a microcycle in which the DCJll asserts PRDC and decodes the prefetch buffer contents as the next PDP11 instruction. Read/Modify/Write (RMW) operation -Two consecutive microcycles in which the first is a Bus Read microcycle and the second is a Bus Write microcycle. Both microcycles access the same location. Request abort -An abort during a request bus microcycle. If it is a memory management or address abort, it will not stretch the microcycle.

Abort (ABORT) line, 2-6, 2-11 Aborts, 1-12, 1-13 AC characteristics, B-1 through B-7 Address input/output (AIO) line, 2-3, 2-10 Address latch enable (ALE) line, 2-5, 2-12Addressing modes direct register, 6-6 direct autoincrement, 6-7 direct autodecrement, 6-8 direct index, 6-9 through 6-11 deferred, 6-11 through 6-14 double-operand, 6-3 through 6 - 4general, 6-1 through 6-3 PC relative, 6-14 through 6 - 18single-operand, 6-3 Bank select (BS) lines, 2-2, 2 - 11Buffer control (BUFCTL) lines, 2-4, 2-13Bus cycles AIO codes for, 3-2bus read, 3-4 through 3-6 bus write, 3-6 through 3-7 duration of, 3-2general-purpose read, 3-8 general-purpose write, 3-9 interrupt acknowledge, 3-10 non-I/O (NIO), 3-3 parts of, 3-3Bus read cycle, 3-4 through 3-6 non-stretched, 3-5 stretched, 3-5 Bus write cycle, 3-6 through 3-7 Cache control register force cache miss bit, 5-2 unconditional cache bypass bit, 5-2 uninterpreted bits, 5-2 Cache memory cache control register (CCR), 5-1 through 5-2general operation, 5-3 in multiprocesing environment, 5-4

Cache memory (continued) sample implementation, 5-4 through 5-8 Cache miss (MISS) line, 2-6, 2 - 12Clock 1 (CLK) line, 2-5, 2-12 Clock 2 (CLK2) line, 2-5, 2-12 Console start microroutine, 8-10 through 8-11 Console ODT, 5-9 through 5-19 address specification, 5-17 carriage return command, 5-14 command set, 5-12 through 5 - 17control-shift-S command, 5-17 floating-point accumulators and, 5-18 general register references, 5-17 go command, 5-16 initialization, 5-11 invalid characters, 5-19 internal register designator, 5 - 15line feed command, 5-14 octal notation for, 5-18 output sequence, 5-12 proceed command, 5-16 processor status word designator, 5-15 receiver control/status register (RCSR), 5-9 receiver buffer register (RBUF), 5-10 slash command, 5-13 stack pointer references, 5 - 18terminal interface, 5-9 timeout, 5-19 transmitter control/status register (XSCR), 5-10 transmitter buffer register (XBUF), 5-11 Continue (CONT) line, 2-4, 2-12 Control chip, 1-1 CPU error register, 1-15 through 1-16 Data/address (DAL) lines, 2-2, 2-11, 2-13 lower, 2-2upper, 2-2Data chip, 1-1 Data valid (DV) line, 2-4, 2-13

Index-1

DC characteristics, A-1 through A-4DCJ11 block diagram, 1-1 DCJ11 pin assignments, 2-1 Direct memory access (DMA) mechanism, 1-17 Direct memory access (DMA) requests and grants, 3-11 Direct memory access request (DMR) line, 2-8, 2-11 Event (EVENT) line, 2-9, 2-10 Floating-point arithmetic data formats, 7-2 through 7-3 nonvanishing numbers, 7-1 zero, 7-2undefined variables, 7-2 Floating-point exception code (FEC) register, 7-7 Floating-point exception (FPE) line, 2-8, 2-10 Floating point instructions ABSF, 7-12ABSD, 7-12 ADDF, 7-13 ADDD, 7-13 CFCC, 7-14CLRF, 7-14 CLRD, 7-14 CMPF, 7-15 CMPD, 7-15 DIVF, 7-15 DIVD, 7-15 LDCDF, 7-16 LDCFD, 7-16LDCIF, 7-17LDCID, 7-17 LDCLF, 7-17LDCLD, 7-17LDEXP, 7-18 LDF, 7-19 LDD, 7-19 LDFPS, 7-20MODF, 7-20 MODD, 7-20 MULF, 7-23MULD, 7-23 NEGF, 7-24 NEGD, 7-24 SETF, 7-25 SETI, 7-25 SETL, 7-25 STCFD, 7-26STCDF, 7-26

Floating point instructions (continued) STCFI, 7-26 STCFL, 7-26 STCDI, 7-26 STCDL, 7-26 STEXP, 7-28 STF, 7-28 STD, 7-28 STFPS, 7-29STST, 7-29 SUBF, 7-29 SUBD, 7-29 TSTF, 7-31 TSTD, 7-31 accuracy, 7-9 through 7-10 addressing, 7-8 through 7-9 Floating-point status (FPS) register, 7-3 through 7-7 Floating-point processing, 1-17 General-purpose (GP) codes, 8-1 General-purpose read cycle, 3-8 General-purpose registers, 1-2 General-purpose write cycle, 3-9 Ground (GND) pins, 2-10, 2-11, 2 - 13Halt line, 2-6, 2-11 Halting DCJ11 operation, 2-24 I space and D space, 4-2 through 4-3Initialization microroutine, 8-2 through 8-6 Initialize (INIT) line, 2-5, 2 - 12Instruction set ADC, 6-34 ADCB, 6-34 ADD, 6-39 ASH, 6-40ASHC, 6-41 ASL, 6-31 ASLB, 6-31 ASR, 6-30ASRB, 6-30 BCC, 6-48BCS, 6-48 BEQ, 6-47 BGE, 6-50 BGT, 6-50BHI, 6-51 BHIS, 6-52 BIC, 6-43 BICB, 6-43

BIS, 6-43

BISB, 6-43		SOB, 6-56
BIT, 6-42		SBC, 6-35
BTTB. 6-42		SBCB. 6-35
BLE 6-51		SEC 6-66
BLO, 6-52		SEN, 6-66
BLOS, 6-51		SEV, 6-66
BLT. 6-50		SE2, 6-66
BMT = 6 - 47		SCC, 6-66
		SPL. 6-61
DNL, 0-40		SUB 6-39
BPL, 6-47		SUB, 0-39
BPT, 6-58		SWAB 6-33
BR, 6-45		SXT, 6-35
BVC. 6-48		TRAP, 6-58
BVS 6-48		TST, 6-28
		TSTB, 6-28
		TSTSET 6-29
CLC. 6-66		$\frac{101001}{10} = 6.4$
CLN, 6-66		WAIT, 0-04
CLV, 6-66		WRTLCK, 6-29
CLZ, 6-66		XOR, 6-44
CLR, 6-26		byte instructions, 6-22
CLPB $6-26$		formats, 6-19 through 6-22
COM = 6.26		list. 6-23 through 6-26
		$\frac{1}{2} \frac{1}{2} \frac{1}$
COMB, 6-26		The symbols, 6-16 through 6-19
CMP, 6-38		interrupt acknowledge cycle, 3-10
CMPB, 6-38		Interrupt and DMA control lines,
CSM, 6-61		. 2-7
DEC 6-27		interrupt request (IRQ) lines,
$\frac{DEC}{CP} = \frac{C}{C}$		2-7, 2-11
DECD, 0-2/		direct memory access request
DIV, 6-42		
EMT, 6-57		(DMR), 2-0, 2-11
HALT, 6-64		power lair (PWRF), 2-8, 2-10
IOT, 6-58		floating-point exception
INC. 6-27		(FPE), 2-8, 2-10
INCB 6-27		event (EVENT), 2-9, 2-10
TMD $6-52$		Interrupt request (IRO) lines.
JOP (52		2-7, 2-11
JSR, 6-53		Interrupts and trans 1-11
MARK, 6-60		through 1 14
MFPS, 6-36		through 1-14
MFPT, 6-65		
MOV. 6-37		Map enable (MAP) line, 2-7, 2-11
MOVB 6-37		Memory management
		addressing, 4-1
MEPD, 0-05		fault recovery, 4-8
MFP1, 6-65		I space and D space $A=2$
MTPD, 6-65		1 Space and D Space, 4-2
MTPI, 6-65		Unrough 4-3
MTPS, 6-36		implementation, 4-14 through
MUL, 6-41		4-18
NEC 6-28		instruction back-up/restart
NECP $6-20$		with, 4-14
NEGD, $0-20$		interrupt conditions. 4-8
NOP, 0-0/		multiple faults A-14
RESET, 6-65		marcipie rauros, 4-14
ROL, 6-32		Paye autress registers
ROLB, 6-32		(PAKS), 4-6
ROR, 6-31		page descriptor registers
RORB, 6-31		(PDRs), 4-6
RTI. 6-59		physical address construction
PTS 6-55		4-3 through 4-5
DTT 6-60		register #0 (MMRO), 4-9
NTT1 0-33		register #1 (MMR1). 4-10
	Index-?	register $#2$ (MMR2), $4-11$
	THUCK-C	,

Memory management (continued) register #3 (MMR3), 4-11register map, 4-19 through 4-20 registers, 4-5 Memory management register #0 (MMR0), 4-9 enable relocation bits, 4-10 error flags, 4-9 page address space bits, 4-10 page number bits, 4-10 processor mode bits, 4-10 reserved bits, 4-10 Memory management register #1 (MMR1), 4-10 Memory management register #2 (MMR2), 4-11 Memory management register #3 (MMR3), 4-11 enable 22-bit mapping bit, 4-11 enable CSM instruction bit, 4 - 13enable I/O map bits, 4-11 kernel, supervisor, and user mode D space bits, 4-13 reserved bits, 4-11 Memory system registers, 1-17 Non-I/O (NIO) bus cycle, 3-3 Oscillator pins, 2-9 XTALI, 2-9, 2-12 XTALO, 2-9, 2-12 Page address registers, 4-6 Page descriptor registers access control field, 4-8 bypass cache bit, 4-7 expansion direction bit, 4-7 page length field, 4-7 page written bit, 4-7 reserved bits, 4-8 Parity error (PARITY) line, 2-6, 2-11 Pin description summary, 2-10 through 2-13 Pipeline processing, 5-20 through 5-22 Power-down microroutine, 8-9 Power fail (PWRF) line, 2-8, 2 - 10Power pins, 2-9 ground (GND), 2-10, 2-11, 2 - 13power (Vcc), 2-10, 2-11, 2-13 Power-up circuit, 8-8

Power-up configuration, 8-6 through 8-8 Predecode (PRDC) line, 2-7, 2-1; Processor status word (PS), 1-3 through 1-11 condition code bits, 1-6 initialization, 1-11 processor mode bits, 1-5 protection, 1-7 through 1-10 priority level bits, 1-5 trace/trap bit, 1-6 Program interrupt request register, (PIRQ), 1-15 Receiver buffer register (RBUF) 5 - 10Receiver control/status register (RCSR), 5-9 Stack protection, 1-16 Start/stop control lines, 2-5 halt (HALT), 2-6, 2-11 initialize (INIT), 2-5, 2-12 Status signals, 2-6 abort (ABORT), 2-6, 2-11 cache miss (MISS), 2-6, 2-12 map enable (MAP), 2-7, 2-11 parity error (PARITY), 2-6, 2-11 predecode (PRDC), 2-7, 2-12 Stretch control (SCTL) line, 2-5, 2-12 Strobe (STRB) line, 2-5, 2-12 System control lines, 2-2 address input/output (AIO), 2-3, 2-10 bank select (BS), 2-2, 2-11 buffer control (BUFCTL), 2-4 2 - 13continue (CONT), 2-4, 2-12 data valid (DV), 2-4, 2-13 Test 1 (TEST1) line, 2-9, 2-10 Test 2 (TEST2) line, 2-9, 2-12 Test pins, 2-9 test 1 (TEST1), 2-9, 2-10 test 2 (TEST2), 2-9, 2-12 Timing signals, 2-4 address latch enable (ALE), 2-5, 2-12 clock 1 (CLK), 2-5, 2-12 clock 2 (CLK2), 2-5, 2-12stretch control (SCTL), 2-5, 2 - 12strobe (STRB), 2-5, 2-12 Transmitter buffer register (XBUF), 5-11 Transmitter control/status register (XSCR), 5-10

Digital Equipment Corporation • Bedford, MA 01730